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Abstraet

Identification of metabolic regulation is a key point in metabolic engineering .
Metabolic regulation phenomena depend on intracellular compounds such as en-
zymes, metabolites, nucleotides and cofactors . Quantitative knowledge about these

compounds in combination with the known network of metabolic pathways allows the
construction of mathematical models that describe the dynamic changes in metabo-
lite concentrations over time. The models are high-dimensional systems of ordinary,
non-linear differential equations . To solve the problems of this approach (setup of
the equations that describe the metabolic pathways in form of kinetic rate equations
and the parameter identification of the System parameters), a variety of pathway
modeling software has been developed over the last years.

This work describes a Metabolic Modeling Tool (MMT) software for pathway anal-

ysis that is based on a relational database . It has been developed and applied to

experimental data during this Ph.D . thesis . The software can be used to define and

analyze dynamic pathway models based on non-linear kinetic rate equations. The

design of the MMT software differs from other available pathway modeling tools in

a variety of things. With respect to model definition, the tool has no limitations in

dimension and it is based on a relational database that stores all model information

and results . In order to allow definition of a large number of models the tool has a

structured approach that allows to build groups of models and groups of parameter
fittings . Groups share e.g . the metabolic pathway or kinetic equations .

With respect to simulation and parameter fitting, MMT provides high flexibility .

Parameters can e.g . be optimized or set to experimental data values or initial condi-

tions can be optimized . Stoichiometric coefficients can also be subject to parameter

fitting . Additionally simulation time is variable and artificial noise can be added

to the calculated values to simulate experimental data . The software creates a C-

Code with the System of differential equations and also with the explicit derivatives

with respect to initial conditions and parameters, thus providing a complete set of

sensitivity functions .

The tool allows not only visualization of the solution of the differential equations

but also of each individual flux between compounds . These values correspond to the

kinetic equations over time and are helpful in analyzing simulation and parameter

fitting results .

MMT was applied to the central metabolic pathways of Escherichia coli. Data from

fast sampling experiments (Glucose pulse experiments, 4 samples per second over

40 seconds) was used for parameter fitting. Some metabolic regulation phenomena

in relation to these experiments are discussed in this thesis .
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1. Introduction

Imagine a civilization that uses cars as a general mean of transport but never figured
out why cars drive. At some point, an increasing amount of people starts to rip
old cars apart, take a look at single parts, e.g . rolling tires down a hill trying to
approximate how fast a car can possibly run . Other scientists look at running cars,
estimating how much gas and other resources are entering the car, and into what
substances the car transforms these resources . Through the years, the civilization
has found all relevant parts needed to build cars and even possesses the tools to put
the parts together . Also, conditions could be found that are good for certain cars and
not so good for others . Now some people are very enthusiastic and claim that they
are able to improve existing cars to make them perfect or even build new superior
cars from scratch - all that with their huge amount of knowledge they gained by
looking at all the single parts of cars, their resource consumption and seeing them
drive . Would you believe these people? Would you give them a chance to make a
specific improvement on a car?

Looking at the history of microbiology, a very similar situation to the above story
exists . For a long time cell behavior has been examined, growth conditions have
been optimized . Over the years, cells have been taken apart and substances as e.g .
enzymes have been analyzed in detail . Then techniques for DNA recombination
became available . Almost 20 years ago, in the year of 1981, some people started to
talk about directed pathway modifications in cells, having in mind better cells e.g .
for the production of amino acids [53] . This process nowadays is often referred to
as metabolic engineering, a term coined by Bailey in 1991 [3] . A widely used defini-
tion of metabolic engineering is the one of Stephanopoulus et al in [96] : Metabolic
engineering means the directed improvement of product formation or cellular prop-
erties through the modification of specific biochemical reaction(s) or introduction of
new one(s) with the use of recombinant DNA technology . Nowadays, a variety of
additional tools is available for creation of new strains for metabolic engineering .

The chances that our civilization mentioned above will improve any car are rather
low . Fortunately some people in this civilization started a new approach, trying to
analyze a combination of parts of the car like e.g . a complete engine. Even better,
they developed methods to examine engines and other parts while the car is driving!
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Therefore interactions between single parts and the importance of single parts for
the functioning of the car could be analyzed . Single parts could be analyzed in a
working environment which resulted in completely différent results than considering
thé parts in an isolated environment in thé garage . Using these techniques it was
e.g . possible to give hints what limits thé speed or thé resource consumption of
thé available cars . Now thé part specialists could use their knowledge to replace or
improve parts that limited overall car performance .

In thé yeld of metabolic engineering, new measurement techniques focus on in-
tracellular, in vivo tell behavior - trying to figure out what controls and limits
intracellular behavior . Now thé intracellular evolution of living tells is analyzed un-
der various conditions ranging from steady drive at high or low speed (steady state
analysis) to dynamic drive under rapidly changing conditions (e.g . pulse experi-
ments, severely limited tells, temperature shift) . Because an increasing number of
intracellular and extracellular substances tan be quantified with increasing accuracy
during such processes, thé dream metabolic engineering has become closer .

1 .1 .

	

Targets and methods of metabolic engineering

Metabolic engineering is of interest for industrial purposes Because it increases ef-
ficiency of production processes [96] . The main purpose here is to decrease the
amount of used substrates, cofactors, and energy, to decrease the amount of waste
(by-producs), and to increase the amount and quality of desired product, thus
increasing the product yield (see Figure 1 .1) .

On the other hand, the way to achieve the goals of metabolic engineering are of
great interest from a scientific point of view Because they give insight in the reg-
ulation and control of cellular evolution . Successful metabolic engineering requires
a thorough understanding of microbiology of cells starting from the genetic infor-
mation of the DNA all the way up to large-scale processes of cell populations [4] .
Figure 1 .2 shows the main functional units of microorganisms . Lately developed and
improved methods provide the basis of a global understanding of cellular life . Thefollowing methods are extensively used now in order to fill the still remaining gapsin the picture of a functioning cell :

Genomics: The genetic information of many organisms is already available . Anal-ysis is automated (robots) . The data, however, only describes thé geneticBackground of a tell, not what genes may be expressed and activated . Thegenetic Background is fixed during thé life time of tells and is strain specific .
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Ressources:
Substrate,
Cofactors,
Energy

Desired
Product »«JWP>

	

Quality

Product Yield =
Desired Product
per Ressources

Waste

Figure 1 .1 . : Industrial process and interests .

It is, however, altered through spontaneous mutations [40] .

Transcriptomics: Measurement of mRNA with

	

DNA-arrays gives information

about expressed genes in a specific physiological situation . Therefore tran-

scriptomics is an important tool for measuring global responses of a tell to

changes in environmental conditions. The time scale of such responses is in

the range of several minutes to hours.

Proteomics:

	

Gel-electrophoresis allows estimation of enzyme concentrations in

tells . Next to transcriptomics, it provides the second important source of data

to distinguish between tells growing under différent conditions. With pro

teomics data, thé amount of enzymes in a living tell tan be analyzed . How-

ever, thé accuracy is low making thé method especially useful for a qualitative

comparison of différent tells . The exact amount of enzymes plays an impor-

tant role for thé dynamic system behavior . Production of a single enzyme is

a process that takes at least 1 to 2 minutes, sometimes significantly longer,

therefore measurements during a short time frame are not useful .

Metabolomics : Analysis of intracellular and extracellular metabolite concentra-

tions is termed metabolomics . It provides data for estimation of intracellular

dynamics in a time frame of seconds or even milliseconds. Because metabolite

concentrations are thought to be thé main regulators of intracellular dynamics

besides enzyme concentrations, but tan change much faster, they are necessary
input for simulations of e.g . a response to a substrate pulse [79] .

Biotransformation
Process
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Influence ori
'L-anscription

Metabolome

Metabolite A

Catalyzes
Reaction

	

M

	

Metabolite S

Enzyme

Proteome

Figure 1.2 . : Genome, transcriptome, proteome, fluxome and metabolome : five cen-
tral parts determining fate of tells .

Fluxeomics: Intracellular fluxes and fluxes through the tell membrane tan be cal-
culated from measurements with labeled substrates as e.g . 13C [110], [111} .
Such measurements give important information about material fluxes inside
tells as well as uptake and excretion fluxes . They are important also for a
dynamic model because measurement of metabolite concentrations alone does
not give information about fluxes .

Fluxonie
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Table 1.1. : Methods for detailed in vivo analysis of microorganisms .

Method

	

II Measurement tools I Established, ease of use

Genomics DNA-Sequencing Well established, commercialized,
automated, complete sequencing of an
organism is still time intensive .

Transcriptomics mRNA
concentrations
using DNA array
chip technology

Development phase, chip production is
time intensive, analysis difficult due to
thé lack of experience . Quantifications
difficult but comparison between cells
or strains is possible .

Proteomics Protein
concentrations
with 2-D, 3-D gel
electrophoresis,
MALDI-TOF

Still in development, quantification and
interpretation difficult, interpretation
time intensive. Mainly useful for
comparison between cells or strains .

Fluxeomics Relative fluxes
with isotopomer
analysis, NMR

Established, commercialized, depending
on the used substrate the
measurements are of high accuracy,
limited to certain pathways .

Metabolomics Metabolite
concentrations
with enzymatic
tests, HPLC,
LC-MS, GC-MS

Enzymatic tests, HPLC established,
LC-MS, GC-MS in development . All
measurements are time intensive .
Accuracy can be influenced by other
substrates that behave similar then the
ones to be analyzed .

Table 1.1 gives an overview over thé measurement methods and thé current state
of thé art . Data gained from classical methods considering in vitro processes is also
useful, but usually their applicability is limited to a qualitative interpretation . Rate
equations for enzymatic reactions can be derived from in vitro experiments because
these equations depend on thé structure of thé enzyme. Additionally, thermody-
namic constants of chemical reactions can be measured in vitro in order to gain
information about thé preferred direction of thé réactions . However, kinetic con-
stants of thé rate equations (e.g . a maximum reaction rate vmax) that are measured
in vitro should be handled with care . This results from thé fact that thé différences
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between in vitro and in vivo conditions have been proven too large in general in order

to use quantitative information as e.g . kinetic constants from in vitro experiments

for in vivo models [76], [16] .

The amount of data gained with the used methods and the complexity of the cel-

lular system make it impossible for a human mind to gain a complete understanding

of the functioning of a tell . Figure 1.3 pictures a common problem in computational

biology : whereas the amount of available data exploded due to the availability of

machinery for generation of accurate data, tools for efficient data analysis have

not been developed with the Same effort . Increasingly detailed mathematical mod-

els describing cellular systems are thought to give hints about intracellular control

mechanisms . Due to the complexity of the System, however, a mathematical model

for a cellular system probably will in the near future not be better than a five day

weather forecast . But if a certain behavior tan be predicted with a sufficiently high

probability, metabolic engineering will probably achieve its goals faster .

1 .2 .

	

Scope of thfs work

The conceptual approach of the pathway modeling project at the Institute of Biotech-

nology is based on an interaction between modeling and experimental work. The

approach of the four basic tasks experiment, analytics, modeling, and experimental
design is shown in Figure 1.4. The process was initially started with experiments .
This work provides a modeling framework for analysis of such data. Results of mod-
els should then be used to design new experiments . This will increase the amount
of knowledge that can be gained with a new experiment.

The goal of this dissertation was the identification of pathway models for the main
metabolic pathwys of Escherichia coli. The dissertation focuses on mathematical
models based on data from metabolomics . Modeling was guided by experimental
data gained in two pulse experiments with Escherichia coli K 12 . For the time span
of the experiments of 45 seconds, models that focus on metabolic regulation have
been developed .

The main reactions o£ the considered Escherichia coli pathway tan be found at
the end of this work in Figure B.1 and on the last page that tan be folded out in
order to have the pathway in mind while reading this work . Abbreviations used in
the pathway maps and throughout this work are summarized in Appendix A.
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7i-aditional way
of data generation :

few experiments,
manual measurements

Modern way of
data generation :

automated experiments and
measurements allow large
number of experiments

Available
knowledge

2000

Requires:

sophisticated
software tools
for automated
analysis and

data
interpretation

Figure 1 .3 . : The methods for data generation and collection have changed in recent
years, whereas the methods for data analysis could not keep up and
are just starting to catch up.
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substrate pulse,
fast sampling

`enzymatie
assays
HPLC,
LC-MS

Figure 1.4 . : Conceptual approach with experiments, analytics, modeling and ex-

perimental design .

In short, the modeling work that is part of this thesis is based on the following
main features :

" Dynamic models are used in order to model the dynamic concentration changes
of intracellular substances after a substrate pulse .

" Enzyme kinetics are used as a basis for each chemical reaction of the pathway
in order to build a mechanistic model of the metabolic network .

" Intracellular metabolite concentration measurements are used for fitting mod-
els to data .

" A Metabolic Modeling Tool Software (MMT) is developed that allows model
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buildup, simulations and parameter fitting of a large number of alternative
models in a short period of time.

" The MMT Software is used to identify a model that is able to describe some dy-
namic phenomena after a substrate pulse that could not be explained without
the help of the model.

Details about underlying experimental data are to be found in Chapter 2 . Chap-
ter 3 explains the concept of mathematical models and metabolic engineering and
compares this work with other approaches. The developed software is described in
Chapter 4, the mathematical solution and parameter fitting methods are part of
Chapter 5. First results with so far available experimental data are part of Chap-
ter 6, and a discussion and outlook are presented in Chapter 7.
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THE MODEL ORGANISM ESCHERICHIA COLI

. have a genome that has been sequenced completely [9] .

. are considered to be rather simple organisms due to their prokaryotic structure .

" are easy to handle in the laboratory and do not require high safety standards

because they are a widespread organism .

Table 1 .1 in Chapter 1 summarized the available methods for in vivo analysis of
microorganisms. Table 2.1 shows the methods and experiments performed at the

Institutes of Biotechnology at the Research Center Jülich using strains of Escherichia
coli . Because many established methods are applied in one research facility for a
specific organism, one can hope for quick improvements in metabolic engineering at
the Research Center Jülich .

Table 2.1 . : Methods and research focus at the Institute of Biotechnology at the
Research Center Jülich.

Method

Genomics

Transcriptomics

Proteomics

Fluxeomics

Metabolomics

Fermentation

11
Research at the Institute of Biotechnology

Construction of new strains with focus on aromatic acid
synthesis pathways

DNA Chip Technology to analyze mRNA concentrations
under varying growth conditions

2-D gel electrophoresis, using MALDI-TOF for analysis of
enzyme concentrations

Labeling experiments with NMR analysis to quantify
intracellular fluxes, mainly with 13C

Quantification of intracellular metabolite concentrations
enzymatic tests, HPLC, LC-MS . Steady state and pulse
experiments in laboratory scale .

Development of fermentation tools and technologies for
defined growth conditions suitable for comparison of strains .
Scale-ups .

with



2.1 . METABOLOMICS AT THE INSTITUTE OF BIOTECHNOLOGY

Based on the experimental data from the applied methods, mathematical analysis
is performed with increasing effort . This dissertation, as one of the starting points
of extensive mathematical modeling at the Research Center deals with the modeling
of data gained from metabolomics only. Therefore, in the following work, this part
of the current research will be discussed extensively whereas details from other fields
will be omitted . However, modeling of the metabolomics data was used as a sample
process and will be extended to more complex models in the future .

2.1 .

	

Metabolomics at the Institute of Biotechnology

In order to measure dynamics of intracellular metabolite concentrations a fast sam-
pling device machinery has been developed at the Institute of Biotechnology. The
goal of the approach is to develop models describing intracellular dynamics as a
response to fast changes in the substrate concentration . For this approach, a fast
sampling device is required to provide sufficient data in the first seconds after such
a change in the substrate concentration.

The fast sampling technique has sonie apparative constraints as mixing time of the
pulsed substrate, accuracy of the pulse, and accuracy of the sampling volume . Ex-
tensive work was necessary for development of a machinery with sufficient sampling
frequency and good accuracy in sampling volume [90] . The sampling time finally
could be reduced to 220 ms. See Figure 2.2 and 2.3 for a picture of the device . The
experimental schedule for data generation is

1 . Setup a substrate limited chemostat for 7.5 residence times to achieve a stable
steady state .

2 . Start the fast sampling device about five seconds before a pulse of substrate is
given . With a sampling time of approximately 220 ms, about 20 samples from
the steady state are taken .

3 . Perform a substrate pulse by injecting substrate with pressure in the fermenter .
Pressure is needed in order to minimize mixing time .

4 . Continue sampling for about 35 seconds which results in additional 140 to 150
samples of cells after the substrate pulse .

5 . Perform analysis of extra- and intracellular compounds with enzymatic as-
say, HPLC, and LC-MS. Due to the large number of samples and the time
consuming methods, this analysis takes about three months of time.
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THE MODEL ORGANISM ESCHERICHIA COLI

Table 2.3 summarizes intracellular compounds that can be analyzed by various
methods and their corresponding limit of detection . The table shows that a mix of
available methods is important to maximize the number of metabolites that can be
analyzed with good accuracy. See Figure B.1 for location of the measured compounds
in the metabolic pathway. For a complete description of the experimental procedure
and analysis of compounds with enzymatic assay, HPLC, and LC-MS, see [90], [13],
[14] .

Experimental data gained with the methods mentioned in this section provides
the main input for the mathematical models developed in this thesis . The models
presented in Chapter 6 where fitted to data from the glucose-glucose experiment .
In these experiments, a sudden change in the substrate concentration from 0.1 to
15 mM GLC was used during the experiments . It is believed that in order to
optimize growth conditions, estimation of intracellular reactions under changing
conditions is necessary, and this can be simulated with a substrate pulse. Large
changes from limited growth conditions to saturated growth conditions are likely
to show limitations in the metabolic pathways as well as maximum reaction rates .
Additionally, large changes in the substrate concentrations cause large changes in
many intracellular metabolites . This makes it easy to distinguish between error of
measurements and real concentration change during an experiment .



2.1 . METABOLOMICS AT THE INSTITUTE OF BIOTECHNOLOGY

Table 2.3 . : All metabolites quantified in cell extracts of Escherichia coli K12 under
glucose (GLC) limited conditions at cell densities of 10 g 1-l . Intra-
cellular limits of detection in [mM] . PP-Pathway: Pentose-Phosphate
Pathway. TCA-Cycle: Citric-Acid-Cycle . See tables A.2 and A.3 for
abbreviations of compounds.

continued on next page

Compound
Name

Enzymatic
Assay [mM]

HPLC (UV)
[MM]

LC-MS
[MM]

glycolysis :
G6P 0.23 0.13
F6P 0.30 0.16
FBP 0.30 0.09
GAP 0.53
DHAP 0.53
3PG 0.30
2PG 0 .38
PEP 0 .38
PYR 0 .38

TCA-cycle :
CIT 0.15
ISC 0.17
FUM
OXO 0.18
SUC 0.19
MAL 0.06
OAA 0.68
ACoA 0.30 0.13

glycerol metabolism :
GLY 0 .27
G3P 0 .40

nucleotides :
cAMP 0 .63
AMP 0.38
ADP 0 .63 0.24
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Table 2 .3 . : (continued)

Compound
Name

Enzymatic

	

HPLC (UV)

	

LC-MS
Assay [mM]

	

I

	

[MM]

	

I

	

[MM]

0 .08
0 .15
0.08
0.14
0.10
0.02

0.38
0.22

2.2 .

	

Biological data interpretation

This section aims at an qualitative understanding of the data used for modeling . Thegeneral problem of interpreting phenomena seen in the data is discussed . Ideas pre-sented here are based on common biochemical knowledge and theories . In Chapter 6mathematical models are presented in order to gain more insight in the phenomenaof the data that are discussed here .

Figure 2.4 shows the substances analyzed in one of the experiments . In the exper-iment a GLC limiting chemostat was combined with a substrate pulse according tothe experimental setup described in Section 2.1 . Figures 2.5 shows the intracellu-lar concentrations over time of the metabolites involved in the phosphotransferasesystem (PT-System) of Escherichia coli for uptake of GLC. The figures include mea-sured values and error bars . Error estimates include the whole process from samplingto enzymatic assay respectively HPLC . Concentrations of other metabolites are notdiscussed here but can be found in Figure 6.15 .
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Figure 2 .4. : Pathway of Escherichia coli. Shown are only compounds measured in
the glucose-glucose pulse experiment discussed in Section 2.2 .
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Figure 2.5 . : Intracellular G6P, PEP and PYR concentrations in [mM] before and
after the GLC pulse . The pulse was given at time t = Os . See text for
comments on error bars .
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2.2 .1.

	

Dynamics of G6P, PEP, and PYR

During growth on GLC, the main substrate uptake of Escherichia coli is performed
with the PT-System . This uptake system consists of three phosphotransfer reactions
with a complicated mechanism [60], [73] . Figure 2.6 describes the main mechanism
with the conversion of one molecule of PEP to PYR for each GLC molecule taken
up.

Because only GLC was available as a substrate for the cells in the discussed ex-
periment, it can be assumed that all substrate is taken up by the cell through the
PT-System and converted to G6P. Under balanced growth on GLC, about 17% of
the flux from the total flux exiting the G6P pool enters the PP-Pathway, whereas
80% enter the glycolytic pathway [60] . These rates are estimated from cell com-
position, substrate consumption, and growth rate of cells . Under such conditions,
another 13% of the initial uptake flux exits the glycolytic pathway to form certain
biomass components . Still about 70% of the flux end in PEP, the metabolite re-
quired for GLC uptake [60] . Because each G6P is split into two PEP molecules
(see pathway map, reaction from FBP to DHAP and GAP), under normal growth
conditions, PEP is not a limiting metabolite . As will be seen now, the situation
changes in the experimental data, where PEP seems to be limiting for GLC uptake
and therefore control dynamic system behavior . This question will also be discussed
further in Chapter 6 where models for the PT-System and glycolysis are described .

The hypothesis of sufficiently high intracellular PEP concentration is in agreement
with the experimentally determined PEP concentration that can be seen in Fig-
ure 2 .5 for the time -5 to 0 seconds . The cell seems to have sufficient PEP available
under GLC limited conditions . Things change dramatically after a sudden pulse
of GLC . Because the PT-System is considered to have a large influence on the dy-
namic response to a substrate pulse, PEP and G6P concentrations as well as PYR,
which is produced from PEP, will be further discussed in this section . The intra-
cellular metabolite concentrations from these data sets are also used in Chapter 6
to model the observed short term dynamics . The following section gives some data
interpretation from a biochemical viewpoint without use of mathematical analysis
tools .
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1.5 to 10 seconds : PEP limitation? During this time, the PEP concentration
remains below detection limit . At the same time, G6P concentration decreases
from its high value present right after the pulse . It is important to keep in
mind that a PEP concentration below detection limit does not imply a zero flux
to PEP or zero GLC uptake. Instead, it is likely that all new PEP molecules
produced are immediately used for the uptake of GLC. Besides, the sum of the
fluxes from G6P to F6P and into the PP-Pathway seem to be larger than the
GLC uptake, following a negative net-flux for G6P which results in a decrease
of G6P.

10 to 15 seconds : During this time a sudden increase in the concentration of PEP
tan be seen to values higher than before the pulse. With a short delay, the
G6P concentration also increases . The reason for the sudden PEP increase
is not clear, however, the following increase in G6P seems to have its reason
in the availability of PEP for uptake of GLC. It is also worthwhile noticing
that the PYR concentration follows the PEP concentration. Because PYR is
produced from PEP, an increase in PEP Gould mean a decreased flux to PYR
and should therefore result in a decrease of PYR. Obviously, some other flux
distribution seems to be present .

15-25 seconds : A low PEP concentration coincides with a high G6P concentration .
Approximately 15 seconds after the GLC pulse a situation occurs that Gould
not be expected a priori . Apparently all PEP is used for additional uptake of
GLC, triggering the increase in G6P until 15 seconds after the pulse. Now,
however, G6P does not decrease after PEP is used up in contrast to the situa-
tion at time 1.5 to 10 seconds . It seems likely that other processes control the
fluxes in this time span . A Gloser look at intermediate metabolite concentra-
tions and their direct influence does not give immediate insight . The eûects
will be discussed in more detail together with the mathematical models for
glycolysis presented in Chapter 6 .

25-35 seconds : Repeating situation from 1 .5 to 15 seconds . In this time frame
the tell behaves like in the time frame 1.5 to 15 seconds . G6P decreases, and
shortly thereafter there seems to start another peak in the PEP concentration.
Because measurements ended at 35 seconds, it Gould not be observed how the
dynamics continue . It is possible that the system evolves to a steady state
or continues to oscillate . Further investigation with extended experiments is
helpful to answer these questions .

Other Metabolites :

It is not surprising that considering just two metabolites out of many can not
explain the above behavior for the complete time of the experiment . Looking at the
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other compounds, however, provides not enough additional insights and does not give
any hints what could happen in the time span 15-25 seconds . The example shows
the difficulties occurring to interpret a dynamic data set of a highly connected, non-
linear dynamic system . The following chapters will show that various mathematical
methods used nowadays in order to gain more insight in the system behavior are
also difficult to apply and require a large amount of biochemical knowledge .

2.3.

	

Aims of the models for Escherichia coli

The background of this work has been presented in this chapter .

	

Additionally,
experimental data is shown and phenomena are discussed . The aim of this work is
to provide a framework for modeling of microorganisms based on a software tool .
The models build with this tool should help to analyze

. intracellular dynamic changes after a substrate pulse as described above

. control mechanisms and limiting steps in complex reaction systems

. importance of futile cycling and anaplerotic reactions

. fluxes in a dynamical system calculated with enzymatic rate equations

Answers to these questions require additional input from flux analysis and steady
state experiments . Also, these questions cannot be answered completely because
estimation of fluxes in a dynamical system are difficult . However, the chosen concept
is thought to be useful for answering some of the above questions .



3. The Modeling Concept

3.1 .

	

Modeling approaches

Developing mathematical models in order to describe a system is a common scientific

technique . In many areas, e.g . electrical engineering, detailed models based on well

established physical laws can be developed . Such models then can be used to predict

system behavior a priori, before the systems are actually setup . Situations also occur

where a detailed model describes an existing system but not all the parameters are

known. This means that all parts of the model are known and can be put in a

mathematical form. Then, even if some parameters are not known a priori, the

model can be fitted to a set of experimental data - assuming that the quality of the

experimental data is good enough .

These situations are almost ideal modeling cases . Models can be used for exact

representation of systems, without neglecting significant parts of the system or intro-

ducing strong assumptions . The requirements to build such models are a complete

qualitative and quantitative knowledge of the system to be simulated . In the bi-

ological sciences, this ideal modeling case is far from reality at this time. Many

simplifications and assumptions are introduced for all models in this field . This is a

result of the complexity of biological systems, the difficulties to gain sufficient and

exact data from experiments, and the lack of qualitative and quantitative under-

standing of the system.

In particular, when dealing with intracellular mechanisms of microorganisms, a

situation occurs where many simplifications and assumptions have to be made in

order to get models that can still be handled and where enough data is available .

This requires a focus on a small part of a "simple" microorganism, and a certain time

frame of interest . It requires the formulation of guiding questions that should be

answered with help of the models. Then, it is possible to develop models specifically

for these questions, neglecting all other parts of the microorganisms . This assumes

that these parts are not important for behavior of that part of the system which lies

in the focus of interest .
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Therefore, models of microorganisms, or models for parts of microorganisms are

still limited to specific questions and cannot describe a complete tell as a system .

Nevertheless, there are already attempts to build models for a complete microor-

ganism, as e.g . the E-tell project [1001 . This complete model tries to simulate the

dynamics of a single tell by including genetic and metabolic control mechanisms.

Because the simulations are not based on experimental data, results have to be

interpreted with cane.

Keeping in mied the available qualitative and quantitative knowledge for this work,

a concept was chosen (Figure 3.1)

. that deals with a reasonably well understood biological system

. that allows measurements with sufficient accuracy for modeling of the system

" that is guided by a number of questions that can be answered with the limited

model

. that results in a model that tan be handled and results tan be interpreted

from a biological point of view

This chapter gives an overview of common modeling approaches and describes the

details of the presented approach . In general, the following three steps are part of

modeling approaches in thin field :

1 . Biological assumptions: At the beginning, biological assumptions about a

system (e.g. a part of a microorganism) are made. This determines the type
of model useful for description of the system .

2 . Mathematical setup o£ the modeln: The modeln resulting from the biologi-
cal assumptions have a certain structure that is important during the process
of modeling . This structure determines the application of the model for de-
scribing the system .

3 . Approaches for simpler mathematical modelst Because models for describ-
ing biological systems are often complex, certain simplifications and model
structures are used. Such approaches are e.g . based certain mathematical
model formulations of enzymatic rate equations or a reduction of model di-
mension.

Microorganisms are complex in their structure . A complete, exact model of a
colony of tells would have to include many phenomena as e.g . population, structure
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or spatial distributions . Biological assumptions are used in order to focus on a
part of the system that can be modeled . The main aspects are discussed in the
following sections, as well as the kind of models that result from these assumptions .
Following are examples of approaches that lead to simpler mathematical structures
of the model.

Modeling:
- available mathematics

- solutions and their
robustness

- time requirements

Biochemistry :
- knowledge of system
- theories, assumptions

- open questions

Experiments:
- possible measurements

- time requirements
- accuracy of measurements

- reproducibility of
measurements

Figure 3.1. : A project plan for modeling parts of a microorganism has to con-

sider availability of technologies, qualitative and quantitative knowl-

edge about the system . Biochemistry provides the knowledge base

for models and performing experiments . Experiments are useful for

increasing knowledge in Biochemistry and provide data for modeling.

Modeling helps to understand complex systems in Biochemistry and

is needed for analyzing data from experiments .
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3 .2 .

	

Biological assumptions

This section describes and justifies thé assumptions thé dynamic models presented

in this work are based on .

3 .2 .1 .

	

Population and single tell models

A population of microorganisms usually consists of tells that are différent under

many respects . Some main différences are

. size and weight

. age o£ tells, or thé stage they are in, i .e . which part of thé tell cycle

. genetic regulation states, i.e. which genes are expressed for thé current condi-

tions

a chemical composition

Some average values of these tell properties tan be found in standard text books

as [60] or [65] . Différences between tells in a population are important because thé

tells behave differently. E.g. larger tells have a larger surface area for nutrient
uptake, or tells produce more structural material in certain parts of thé tell cycle .
Genetically, mutations occur and a population tan consist ofcertain sub-populations .

Some models try to incorporate population variety. An introduction is given in [19] .
Cell-cycles o£ a population of tells are considered e.g . in [15], a mass structured
population model tan be found in [61], thé penicillin production in combination with
a morphological structured model is presented in [114] . For three reasons, différences
between tells of a population are often neglected in models of microorganisms :

1 . From thé above mentioned différences in tells of a population, size and weight
are probably thé only ones that tan be measured . However, différences in size
and weight, as all other différences, tan hardly be measured with sufficient
accuracy for modeling.

2 . Including différences between tells of a population would increase thé com-
plexity of thé models dramatically, because a system of Integro-Differential
Equations would result .
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3. Cells grown in a small laboratory fermenter as described in Chapter 2 and [90]
are considered to differ in rather small ranges .

The underlying experimental data of this work consists of measurements of intracel-
lular compounds of an average cell . No information is available about thé various
différences in thé cells of thé population . Also, in thé time span of 45 seconds of thé
experiment, no significant changes in thé above mentioned properties as e.g . size
occur . Therefore, and in order to reduce complexity of thé model, an average cell of
thé population has been considered and modeled.

3 .2 .2 .

	

Environmental conditions

In a fermenter, gradients in nutrient concentration, temperature, pressure, or pH

exist . These différences result in varying non-stationary intracellular dynamics be-

cause not all cells see exactly thé same environmental conditions . The différences

can be especially important in changing environments, where non-linear effects oc-

cur . An exact model representation of such a system therefore has to include these

différences . Such a model then should also include thé influence of such différences

on thé cells . This, however, requires to model thé movement of single cells in e.g.

a fermenter . The movement can be modeled by stochastic processes or by dividing

thé fermenter in a number of compartments and describe movement between these

compartments . As a result, either a system of Stochastic Differential Equations

or an additional system of Ordinary Dif érential Equations on thé population level

results .

Obviously, it is difficult to measure gradients in an environment of cells . It is

basically impossible to measure and model thé movement of cells inside a fermenter

with good accuracy. Recently, a model incorporating différent micro environments

of a large-scale fermenter and movement between thé micro environments has been

presented in [44] . Also, models have been developed that simulate distribution of

gas and liquid in reactors [93] . For thé laboratory fermenter used to generate thé

data of this work, measurements show a well-mixed environment [90] . Therefore, in

this work, effects of gradients in a fermenter are omitted .

3 .2.3 .

	

Structured and unstructured models

Often models of microorganisms are divided into two classes, structured and un-

structured models. In a population, e.g . an age structure or size structure could be
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introduced . For single cell models, "structured" means the consideration of intra-

cellular compartments . It is important to distinguish between spatial compartments

and conceptual compartments, see Figure 3 .2 . Spatial compartments are parts of

the cell that are surrounded by a membrane, as e.g . mitochondria in eukaryotes.

Such compartments are discussed in the following section . A conceptual compart-

ment is the totality of a compound or a group of compounds inside the cell . An

example is the biomass in general, a group of compounds of a reaction group (e.g .

of glycolysis), or amino acids as a group of compounds . With this definition, a cell

can e.g . be structured into reaction groups .

Strictly speaking, a completely unstructured model of a single cell is a model that

considers only fluxes into a cell and out of a cell . All intracellular processes are

neglected, the cell is viewed as a one-compartment black box. An example can be

found in [96] .

Defining useful compartments is difficult . If too many compounds are considered

as one conceptual compartment, the model might not represent the cell behavior
in a correct way. Fluxes between compartments are usually considered as chemical
reactions . Therefore, adding more compartments to the models does not change the
structure of the model but increases the dimension of the model.

Most single cell models include some structure of cells . The level of detail varies
in a wide range. Some models include only a few conceptual compartments that
represent a storage of compounds in the cell and therefore e.g . allow estimation of
growth after nutrients are depleted [38] . Other models include uptake systems of
nutrients and many intracellular compounds and enzymatic reactions [79] .

Because most single cell models are structured and just dif£er in the level of detail,
it is not very helpful to divide models just in structured or unstructured . Also, an
unstructured model often results from a structured model after simplifications (e.g .
with a quasi-stationary assumptions) have removed all the structure . For this work,
a structured representation of the single cell is used. The more important questions
about the level of structure and the kind of structure are discussed in the following
sections .
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Spatial Compartments

Intracellular
Membrane

Intracellular
Membrane

Conceptual Compartments

Figure 3 .2 . : Spatial (top) and conceptual (bottom) compartments in a tell . In the

graph on the top, two parts of the tell surrounded by a membrane are

shown. Together, three spatial compartments result . In the graph on

the bottom, two compounds or groups of compounds are shown that

are modeled as two compartments . A combination of conceptual and

spatial compartments is also possible.
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3.2.4.

	

Spatial compartment models

Spatial compartmentation is important for eukaryotes or higher tells . These tells are

complicated in structure and possess various intracellular compartments surrounded
by membranes . For some metabolites and nucleotides (e.g . OAA, ATP), two or
more separate pools exist in tells . E.g . in yeast, these two separate pools exist in
the cytoplasm and in the mitochondria .

Models including spatial compartments have to account for transport systems or
incorporate equations to describe the exchange of such metabolites from one pool
to the other via the membrane. The approach increases the dimension of the math-
ematical model but tan be formulated in a straightforward way.

The problem, however, is that it is almost impossible to measure intracellular com-
pound concentrations specific to a compartment . Therefore quantitative knowledge
about compartment specific concentrations in e.g . eukaryotes is rare . Still, one
of the famous single tell models for yeast described in [79] includes such a com-
partmentation . Escherichia colt is a prokaryote and does not contain structural
compartments . Therefore, the models of this work Gould be formulated without
structural compartmentation .

3.2.5 .

	

Genetic and metabolic efFects

Cells adapt to their environment by changing their intracellular levels of enzyme
concentrations . This genetic regulation of the tell results from the dependence of
enzyme production on the induction of certain genes. Genetic regulation is a rela-
tively slow process . lt usually takes at least 30 minutes until enzyme concentrations
change significantly.

Genetic regulation can in principle be included in the model as a chemical reaction
network . Then the model increases in dimension whereas the mathematical structure
remains the same .

Some models are used to model long term effect by inclusion of genetic régu-lation [58] . These models focus on genetic control and neglect metabolic controlthrough metabolic regulation. Complete models including genetic and metabolicregulation are rare because they are complex and operate on very différent timescales . One example that provides a framework for modeling a complote tell includ-ing both genetic and metabolic regulation tan be found in [55] .
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This work considers short term effects, in particular data from pulse experiments
from a time frame less than one minute . In this time frame, enzyme concentrations
can be considered to remain constant . Therefore, genetic regulation is neglected in
the models .

3 .2 .6 .

	

Intracellular spatial gradients

As discussed above, populations of tells experience gradients in e.g . nutrient supply
or temperature . Inside a tell, concentration gradients also exist . These gradients are
likely to exist mainly for certain compounds. After a substrate pulse e.g ., substrate
will enter the tell . However, it will take some time until the Substrate moves from
the tell membrane further to the middle of the tell . Therefore, a gradient exists .
Gradients also can result from membrane bound enzymes . Reaction products of the
reactions catalyzed by these enzymes can be present in significantly larger concen-
trations at the places produced than elsewhere in the tell . Another point is the high
number of large molecules in a tell which makes it difficult for large molecules to
move around quickly.

For a correct description of reaction rates the described phenomena need to be
included in a model. This requires formulations of movements o£ molecules inside
a tell . Due to the high number of large molecules inside a tell and the complex
structure of a tell, such models seem to be impossible to define . Also, measurements
of intracellular concentration gradients are almost impossible in microorganisms .
Some o£ the effects of intracellular movement may be examined theoretically. How-
ever, so far no modeling approaches describing these ef£ects in small microorganisms
like Escherichia coli are known. For higher tells, measurements have been taken
and a model has been developed. The results obtained indicate that intracellular
concentration changes depend strongly on the size of tells [7] .

Another effect that has been modeled is the influence of an overall high concen-
tration of large molecules in a cell . This is called macromolecular crowding and is
known to increase rates of enzymatic reactions in the cell . The phenomenon can be
explained with laws from the theory of entropy [105] .

In the presented models, uniform distribution of the present molecules in the cyto-
plasm of the tell is assumed . Because modeled tell sizes are in the range of < 10 /cm,
an uniform distribution seems realistic . In [91] the diffusion coefficient of water
molecules in bacteria was estimated . A mean residence time of water molecules
of less than 10 ms was calculated . This residence time means that water molecules
move around the Gell in a matter of milliseconds . It is therefore reasonable to assume
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a uniform distribution of other small molecules (e.g . metabolites) in the cytoplasm

of microorganisms .

3 .2.7 .

	

Stochastic processes

For correct modeling of reaction rates, the total number of compound molecules is of

interest . Especially for compounds that are only present in low numbers, a correct

model must use the total number of molecules instead of a simple concentration . This

results from the fact that binding of compounds to enzymes is a stochastic process .

These effects are especially important for genetic regulation, where sometimes only a

few molecules inside the cell are sufficient for regulation. Models that describe such
effects include stochastic components. If these models also include other chemical
reactions, a mixed stochastic and deterministic model results that is difficult to
analyze .

Measurements of the intracellular number of molecules is impossible in microor-
ganisms . However, the effects of stochastic if compared to deterministic models can
sometimes be examined theoretically. Some models taking into account that only a
few molecules of a certain compound are present exist [46], [56] . These models are
based on stochastic processes and describe genetic regulation .

This work focuses on metabolic pathways. Metabolites are only a small part of
the cell, approximately 3 .5% of the dry weight ([65] chapter 3) . Nevertheless, be-
cause metabolites are small molecules, this small amount of the cell still consists of
a large number of metabolite molecules . For an intracellular metabolite concentra-
tion of 0.1 mM and an estimated number of 5.0 . 1010 cells per liter, the number
of molecules in each cell is larger than 108 (estimated from values found in [65],
chapter 3) . Therefore, stochastic processes are neglected and a deterministic model
is formulated .

3.3 .

	

Structure of the models

The models of this work

" are single cell models describing an average cell

" assume environmental conditions to be equal for all cells

" include conceptual but no spatial compartments
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" describe metabolic control phenomena

o neglect genetic regulation

" assume uniform distribution of molecules in the cells

" neglect stochastic effects inside the cells

This section describes the general model structure that results from these assump-
tions . Then, some aspects of model application for various purposes are discussed .
An example is introduced that will be used for illustration of concepts throughout
this work .

3 .3 .1 .

	

The general metabolie model

All intracellular physiological processes can be described by generalized chemical
reaction kinetics . In particular, equations for transport phenomena can be formu-
lated as chemical reactions (e.g. transport of GLC through the membrane with the
PT-System is a reaction GLC + PEP -> G6P -I- PYR). The complete cell is then
described with a biochemical reaction network in common chemical notation [98] in-
cluding transport phenomena. The chemical reactions describe all four basic fluxes :

Compound fluxes : Fluxes in anabolism and catabolism are processes based on
chemical reaction kinetics .

Energy fluxes: Energy fluxes are described by chemical reactions, e.g . usage or
production of nucleotides such as ATP or NADH.

Signaling fluxes : Fluxes for signaling (processing of information) are described
with reaction kinetics, e.g . reactions for synthesis of macromolecules as DNA
or RNA.

Transport fluxes : Fluxes for transport through membranes (intra- or extracellular
membranes) are described with chemical reaction kinetics .

Transport phenomena are difflcult and many are not yet fully understood . GLY e.g .

is taken up by Escherichia coli by diffusion, a physical process [65] . GLC, on the

other hand, is taken up by the PT-System with a membrane bound enzyme and the

consumption of PEP [65] . This is a mix of a physical and a chemical processes . In

the models described in Chapter 6 only the PT-system is included and described as

a reaction GLC + PEP --~ G6P + PYR. In the following, it is shown how systems

of chemical reaction kinetics are formulated in mathematical terms .
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3.3.2 .

	

Notation used for this work and an illustrative example

As an illustrative example of a chemical reaction network, consider the reactions

from F6P to FBP and from FBP to PEP (the intermediate steps of glycolysis are not

included here) . Figure 3 .3 shows the example system . This illustrative example will

be used throughout this work . it is always referred to as the "illustrative example" .

PEP

Figure 3.3 . : Example of a chemical reaction system with five compounds F6P, FBP,
ADP, ATP and PEP and five reaction rates (fluxes) vl to v5. For

clarity, fluxes are numbered here in contrast to the text where e.g .

V3 = VFBP {i F6P ls used .

The following notation is used:

" By abuse of notation, intracellular compound concentrations will always be
written with their abbreviations only, i.e . F6P is the intracellular compound
concentration of F6P in [mM]. This is a short but more readable form of the
usually used CFGP.

" Kinetic rates are written as VA->B for an irreversible kinetic rate from com-
pound A to compound B and VAtrB for a reversible kinetic rate . Notice that v
in the reversible case can be a net reaction rate (net-flux) that represents the
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sum of a forward and a backward reaction that might be catalyzed by différent
enzymes. Sometimes (for clarity) the fluxes are summarized in a vector and
numbered vl to vN,, where Nv is the number of reaction rates included in the
model.

Reaction rates v have a preferred direction . This direction is usually the thermo-

dynamically favorable direction . This direction is defined for each reaction rate . A

positive rate means that the flux is in the thermodynamically favorable direction,

whereas a negative value means that the flux is in the opposite direction . In this

case, the reaction would take place against the thermodynamic equilibrium .

A -+ in the index of v indicates that only the forward reaction is possible (irre-

versible reaction), i.e . v is always positive . A ++ means that the reaction is reversible

and v can also obtain negative values . In such cases a positive value of v means a

flux from left to right . E.g. if VFBP H PEP is positive, a flux from FBP to PEP is

active .

Using this notation, the example network can be written as

To distinguish between the forward and backward kinetic rates as for vFBP +_> PEP

is important e.g . if a different mechanism is active for the reaction rates . Mostly,

however, these kinetic rates are summarized in one rate equation for the net-flux

VFBP t-> PEP

3.3 .3 .

	

Bipartite graphs

A discrete description of a chemical reaction network leads to a bipartite graph

or digraph [2], [109] . Figure 3.4 shows such a graph of the illustrative example .

Metabolites are represented by the squared nodes of the graph, reaction rates vz are
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represented by circle nodes. The nodes are connected to indicate chemical reactions
between metabolites . Because reaction V4 converts one FBP molecule into two
molecules of PEP, a double arrow is used in the graph. In this way stoichiometric
properties of reactions are incorporated in the graphs . Additionally the bipartite
graph or digraph is enhanced by effectors on the connectors (reaction rates) that are
indicated by an arrow and a + or - sign . This enhancement violates the definition
of a bipartite graph because two nodes of the same kind (two circled nodes) are
connected .

Representation of chemical reaction networks using enhanced bipartite graphs or
digraphs is useful for model development because it allows structured overviews .
Identification of end points (i.e . compounds that appear only once in the system as
a substrate or product) becomes easy. From bipartite graphs, mathematical models
can be derived easier and automatically. Representation with this approach allows
usage of theoretical methods and software from graph theoretical approaches like
petri-nets [75] .

3.3 .4 .

	

The stoichiometric matrix

The chemical reaction network can be described in matrix form. This matrix can be
extracted from the enhanced bipartite graph of digraph described in the last section .
For the graph of the illustrative example in Figure 3.4, the matrix representation is
(- represent zero entries)

F6P FBP ADP ATP PEP
VF6Pin --+ F6P

	

+1
VF6P -> FBP

	

- 1

	

+1

	

+1

	

-1

	

,
VFBP -~ F6P

	

+1

	

-1
VFBP ta PEP

	

-1

	

+2
VPEP t-> PEPout

	

-1

N

Numbers indicate the stoichiometric constants . VFBP +-> pEp e.g. describes the reac-tion 1 FBP <-+ 2 PEP, i.e . the overall PEP production via glycolysis . Notice thatthe reversible reaction from F6P to FBP is split into a forward and a backward re-action . This is necessary because only in the forward reaction ATP is consumed andADP produced whereas in the backward reaction these compounds are not part ofthe chemical reaction . In contrast, the reversible reaction from FBP to PEP can beincluded as one kinetic rate equation. Only compounds for which balance equationsexist in the model are included in the matrix . This is the reason for the fact thatexternal metabolites as e.g. GLC are not part of the matrix .
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F6P ATP

PEP

Figure 3.4. : Representation of the example pathway using an enhanced bipartite

graph. See text for details .

The matrix that represents the chemical reaction network is called the stoichio-

metric matrix and labeled N. Each row of the matrix represents one node vi of

the bipartite graph that itself represents a chemical reaction rate. Thus the rows

can directly be derived from the bipartite graph and vice versa . Notice that the

rows contain only the stoichiometric coefficients for the chemical reactions without

effectors . The matrix therefore contains not all information from the bipartite graph

or digraph .
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3 .3 .5 .

	

Chemical reaction kinetics

Quantitative models are based on the stoichiometric matrix N and a flux vector v .

In the non-stationary case, the flux vector v is not constant with time . Instead,

the reaction rates that form the flux vector are functions of compound and enzyme

concentrations . The rate equations for v are therefore functions

v=v(C,S,E,P) (3.2)

where C and S are the vectors of all intracellular respectively extracellular compound
concentrations, E is the vector of enzyme concentrations, and P is the vector of all
kinetic parameters . Because E is usually not known in vivo, Most kinetic rates
assume v to be a function of C, S and P only. The enzyme concentration E is then
hidden in the kinetic constants in the parameter vector P . In the following v will
therefore be written as a function v(C, S, P) .

The rate equations include effectors such as the negative influence of PEP on v2
in the illustrative example . Thus the effectors included in the bipartite graph or
digraph in Figure 3.4 can be found in the reaction kinetics vi, whereas the rows of
the stoichiometric matrix contain only the stoichiometry of the chemical reactions
as explained in the last section .

The description of the reaction rate equations is based on biochemical knowledge
and tries to explain cell behavior based on such knowledge, e.g. [80], [27], [22] . The
most famous equation is the Michaelis-Menten equations which reads as

Cv = vmax ' K,�. + C (3.3)

for a reaction rate v in dependence on two parameters vmax and K,n and a substrate
concentration C . The equation is derived from the mechanism of one substrate
binding to one enzyme . vmax is the maximal reaction rate, Km is the concentration
where v = 0.5 vmax, the half saturation constant . Not for all reactions such easy
mechanistic formulations are appropriate, e.g . if effectors are included in the equa-
tion the equations get more complicated . In particular, many reaction rates have
several parameters that have to be determined for a correct model of the system .
These problems will be discussed later in this chapter .

For the example chemical reaction system above, assume a constant influx vl to
F6P, and Michaelis-Menten equations for all other fluxes . The reaction rate from
FBP to PEP is modeled with a reversible Michaelis-Menten equation . This equation
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contains two v,,,ax values indicated by Vf for the forward and V,. for the backward
reaction. Also K�,, values exist for both directions of the reaction. The equation
reduces to the irreversible Michealis-Menten equation if the Substrate or product
concentration is zero . Now the non-stationary model has the following form:

F6P

	

=

	

VF6PIn ~F6P - vmax,v2 *

	

F6P

	

+

	

,v3
,

	

FBP
Km,v2 + F6P

	

vm,ax

	

Km,v3 + FBP

FBP = F6P

	

FBP
vmax,v2 '

KM,V2 + F6P - vmax,v3 ' Km v3 + FBP

FBP
Vf,v4 . Km.S,v4

PEP1.w4 " KMP,v4
1 + FBP + PEP

KMS,v4 KrriP,v4

F6P
ADP

	

=

	

vmax,v2 ' K

	

-I- F6Pm,v2

F6P
ATP

	

=

	

-vmax,v2 ' Km,v2 + F6P

FBP _

	

PEP

PEP

	

=

	

2 Vf,v 1'
+mSFBP

	

+r'vPEPf
mP,v4 _ vmax,v5

	

m' PEP

	

(3 .4)
K

	

+ PEP
FiMS,v4 KMP,v4

For a realistic description of ADP and ATP, additional terms would have to be
included in the model . Otherwise all ATP will be converted to ADP in a short
period of time .

3 .3.6 .

	

Representation of dynamic pathway models

Using the stoichiometric matrix N and the flux vector v, the following equation

describes the dynamic behavior of the chemical reaction system:

C=N-v(C,S,P) (3.5)

C is the derivative with respect to time of the vector of intracellular compound

concentrations. Therefore, a system of Ordinary Differential Equations (ODES)

results . The System is non-linear due to the non-linearity of the rate equations

v(C, S, P) .
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For the illustrative example network of chemical reactions above the following

system of ODES results (v with numbers in short notation used for clarity) :

C

/ F6P

	

vl - v2 -I- vs
FBP

	

V2 - V3 - v4

ADP = v3

ATP -vs
PÉP 2 v4 - v5

(3.6)

During this work, estimated reaction rates do not depend explicitly on time, i.e . v is
only a function of C, S and P. This means that the right hand side of Equation 3 .5 is
time independent and the system of ODEs is autonomous. Changes in extracellular
compound concentrations (e.g . to simulate a substrate pulse) can be modeled by
changing the corresponding value in the vector S at a specific time during solution
of the system of ODEs.

Non-stationary models allow the investigation o£ dynamic system behavior . Simple
systems were used more than 20 years ago to describe e.g . the reaction F6P --> FBP
that is catalyzed by the enzyme phosphofructokinase. It has been shown that oscil-
lations can occur with a simple system of 2 ODEs due to the product inhibition of
phosphofructokinase by FBP ([95], [36]) .

If a steady-state for the microorganism is assumed all components of the flux
vector v are constant, i.e . C = 0. Additionally, fluxes into and out of each compound
pool add up to 0, i.e . the right hand side of Equation 3.5 is 0 :

0 = N - v(C, S, P)

	

(3.7)

The now constant vector of reaction rates v must fulfill the system of linear stoi-
chiometric equations

which is the classical stoichiometric model .

The system o£ equations of the stoichiometric model usually does not have a unique
solution for pathways of microorganisms . This can already be seen in the illustrative
example model: Suppose a solution of Equation 3 .8 has been found for the model,
then multiples of this solution will be a solution again . E.g . vi = v2 = v4 = 1,
v5 = 2 and v3 = 0 is a solution, and v1 = v2 = v4 = 2, v5 -- 4 and v3 = 0 is also a
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solution. Additionally, the exchange flux between F6P and FBP is not unique. E.g.
v1 = v4 = 1, v5 = 2, v2 = 3 and v3 = 2 is also a solution .

Stoichiometric models are used e.g . for metabolic flux analysis [102], and calcula-

tion of possible intracellular fluxes or optimal fluxes under a certain constraint [106],

[20], [10] . Data from flux analysis can be used to estimate model parameters in the

steady state situation . Stationary models have not been analyzed during this work.

3.3 .7 .

	

The standard system of non-linear ODEs

For this work, non-stationary models are developed that are based on the straightfor-

ward modeling approach described above that leads to systems of non-linear ODEs.

These models are based on a mechanistic description of chemical reaction kinetics.

Many models can be found in the literature that are based on this approach . Some

include genetic regulation or properties as e.g . size in addition to metabolic con-

trol . A classical model that predicts cell composition, cell size and cell shape can be

found in [27] . It describes transcriptional effects and structures the cell in différent

compartments .

In [81] and [79] models describing the central metabolic pathways of Saccharomyces

cerevisiae can be found. These models are focused only on metabolic control for

short term experiments . The model presented in [79] is the more complicated of the

two with 22 dynamic mass balances and 23 reaction rates . The system of differential

equations depends on more than 80 parameters. These two models are mentioned

here because their focus is the same as for the models in this work: describe metabolic

regulation of the central metabolic pathways with a mechanistic model.

The examples mentioned are difficult to handle in a mathematical sense because

they depend on many parameters . The illustrative example as presented in Equa-

tion 3.4 contains a total of 11 kinetic parameters. These parameters are not known

a priori but have to be estimated from experimental data. Such data is still rare and

is often noisy. The advantage of such straightforward modeling is the biochemical

description of the system . This allows interpretation of results from a biological

point of view. E.g. if the parameter v�,,a,,, is identified, the capability of an enzyme

catalyzing the reaction can be estimated .

The next section presents methods for simplifying such models and different struc-

tural approaches that lead to mathematical formulations that are easier to handle .
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3.4.

	

Linearization, simplification and structuring

techniques

A model that was build to represent a system can be used e.g . for simulation of an

experiment, parameter estimation, and prediction of system behavior . As already

mentioned above, many modeling approaches for microorganisms result in large

mathematical models with many parameters . Therefore, many attempts can be

found in the literature for simplifications of such models in order to simplify usage

of the model . Some attempts use a different structure of the models in order to

simplify usage of models . A few famous examples are given here.

3 .4.1 .

	

Linearization of enzyme kinetic reactions

Metabolic Control Analysis

Linear representations for enzyme kinetics of metabolic networks have been intro-
duced in order to estimate kinetic parameters important in the control of fluxes .
Metabolic Control Analysis (MCA) is based on such linear representations [50], [42] .
The approach is based on Flux Control Coefficients (FCCs) kj" defined by

yi _= _Ej dvi - dln(vi)
vi dEj dln(Ej)

(3.9)

where vi is a flux for reaction i at a defined steady state and Ej is an enzyme activity
of enzyme j . Flux control coefficients therefore indicate the relative change in the
steady state flux vi in dependence on an infinitesimal change in the enzyme activity
of Ej . In a similar way, Concentration Control Coefficients (CCCs) can be defined

Ct = Ej dCi

	

dln(Ci)
Ci dEj

	

dln(Ej)
(3 .10)

for the change in a compound concentration Ci in dependence on an infinitesimal
change in the enzyme activity Ei. Sometimes the enzyme concentrations Ei are
replaced e.g . by compound concentrations that can be analyzed easier . The principle
approach is the calculation o£ flux coefficients in dependence on other concentrations
or fluxes of any kind .

If the theory is extended to reaction networks it allows the estimation of overall
changes in fluxes and compound concentrations . Because the linearization assumes
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a linear relationship between enzyme activity and flux it is only valid locally. The
approach is therefore limited to flux distributions and compound concentrations
close to the steady state used to calculate the control coefficients .

In the illustrative example five FCCs and five CCCs can be calculated if fluxes and
enzyme or compound concentrations from steady states are available . It is obvious
that the enzyme with the largest FCC exerts the largest control of the fluxes . If this
specific enzyme is changed, it results in a large change of the overall flux .

Biochemical System Theory

Biochemical System Theory (BST, [87], [88]) assumes that reaction rates can be
described by general power law representations :

vi = aiHXgi7

	

(3 .11)
.9

where Xj are system variables (metabolite concentrations, enzyme activities, ef-
fectors, etc .), ai are apparent rate constants, and gib are apparent kinetic orders .
Notice that vi represent usually the sum of the influx and outflux of a metabolite .
If logarithmic variables are introduced in the rate equations of the form in Equa-
tion 3.11, a linear system of equation results . Therefore the BST approach can be
considered as a linearization of non-linear kinetics in logarithmic coordinates .

If compared to direct linearizations, linearizations on a logarithmic scale are much

more accurate for non-linear reaction kinetics . Notice that the approach lacks a bio-

chemical background of parameters for complex reactions . Therefore interpretation

of results is difficult . An advantage of the approach is the easy way to define a model

of a large chemical reaction system .

In order to setup models for reaction networks, reaction rates vi in power law form

have to be formulated for each reaction. As a result, a system of ODEs evolves .

The system with power-law representations for reaction rates is called a synergistic

system, or S-system [89], [104] . In [40] a model based on the S-system approach

is discussed and applied to optimization of regulatory architectures in metabolic

networks .

The illustrative example can be written in the S-system notation. E.g .

v2 = VF6P -> FBP = a2 G-,12,1
'
Gla~l = a2 F6PF6P P

	

ATPÄTPP

	

(3.12)
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for the reaction from F6P to FBP with two educts . In order to incorporate the

inhibition of this reaction by PEP the equation can be extended to

vF6P -3 FBP = a2 F6PF6P P ' ATPÂTP P ' PEPFÉP P
(3.13)

where the coefficient g2,PEP has a negative value. Equations for the other fluxes of

the illustrative example are formulated in the Same way.

3.4.2 .

	

Simplified enzyme kinetic reactions

Enzyme kinetic reactions are often simplified in order to reduce the number ofkinetic
parameters and to allow inclusion of qualitative effects in an easy way. A few

examples are given in this section . From the discussed examples, the multiplicative
equations, fuzzy logic approximations and pool lumping methods are used in the
models presented in Chapter 6.

Multiplicative equations

Equation 3.3 shows the Michaelis-Menten kinetics as a simple rate equation . The
equation depends only on a single compound concentration. For reactions with
more than one educt, the Michaelis-Menten reaction cannot be used . Instead of
formulating complex equations with a large number of parameters for such cases it
is possible to multiply a number of simple kinetic equations . E.g . in the case of a
reaction with two educts two Michaelis-Menten equations can be multiplied :

Ci C21) = %ax '

	

'
Kml -1- Ci

	

Km2 + C2
(3.14)

C1 and C2 are the concentrations of the two educts of the reaction. An inhibitor
by a third compound concentration C3 can added by another multiplication :

CI C2 KIv = vm,ax ' Km1 -f- Ci

	

Km 2 -!- C2

	

C3 + KI (3.15)

Kj is an inhibitory constant .

The multiplicative approach allows automatic model generation e.g . from a bipar-
tite graph or from databases that contain chemical reactions . In particular, it is
possible to build dynamic models directly from public Internet databases . In the
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illustrative example, it is possible to formulate a rate equation for the reaction from
F6P to FBP as

VF6P --~ FBP = vmax «

	

F6P

	

ATP

	

KI

	

(3.16)
Km,F6P + F6P

	

Km,ATP + ATP

	

PEP -i- KI

Fuzzy logic approximations and piecewise linear approximations

Fuzzy logic based modeling tries to simplify equations for enzymatic reaction rates
and is a canonical approach [59] . Instead of using equations with many parameters,
effectors are included using a fuzzy approach . The approach is often based on
algebraic equations such as the Michaelis-Menten kinetic to describe the chemical
reaction kinetics . Inhibitors and activators are then introduced as effectors that
change e.g . the maximum reaction rate v�Zax . This influence is described with
fuzzy-logic rules .

If the reaction from F6P to FBP in the illustrative example is modeled with a
single substrate Michaelis-Menten kinetic then

F6P
VF6P -+ FBP = vmax ' Km,F6P + F6P

(3.17)

The inhibition of PEP of this reaction can be included using a fuzzy rule for the
parameter vmax :

vmax,l
vmax =

vmax,2

for PEP < KI,PEP
for PEP > KI,PEP

(3.18)

where KI,PEP is a minimal concentration of PEP for inhibition of the reaction .
Notice that vmax,l > vmax,2 for an inhibitory effect .

The reduction in number of parameters simplifies parameter fitting . On the other
hand, some biochemical insight is lost due to the qualitative description of effectors .

It is also possible to use piecewise linear approximations for non-linear rate equa
tions. The Michaelis-Menten equation can be approximated by two functions : one
for small compound concentrations with a proportionality assumption and one for
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large compound concentrations where the reaction rate remains constant, e.g .

ai Ci

	

for C1 < L,
a2

	

for C1 > L,
(3 .19)

where L, is a limit concentration . This approach is similar to the definition of fuzzy
rules for rate equations . The fuzzy approach is more general because it also allows
non-linear functions in the piecewise rate definition .

Power-law formulations

Enzyme kinetic equations can be simplified also by using power-law forms. A power-
law formulation o£ enzyme kinetic rate equations was already introduced in the last
section, see Equation 3.11 . It is also possible to define functions for reaction rates
where compound concentrations appear in the exponent, i.e .

v = aki Cl b-k2 C2 (3.20)

with positive constants a, b, kl and k2 . The second term represents an inhibition of
the reaction by C2 due to the negative sign for k2 . A problem of such a representation
is the fact that reaction rates do not converge to a finite maximal rate for increasing
compound concentrations (here C1) but increase without limit .

In the illustrative example the reaction F6P to FBP can be formulated as

vF6P -~ FBP - a(kl
'F6P) . a(k2 -FBP)

	

a(-ka -PEP)
F6P

	

FBP - PEP (3.21)

where k3 has a negative sign to describe the inhibitioy effect of PEP on the reaction
rate VF6P-+FBP .

Cybernetic modeling

Cybernetic models include genetic and metabolic control and are therefore designed
to build complete models of microorganisms, e.g . [33] or [31] . The approach does not
require the explicit formulation of any chemical reaction rate and therefore simplifies
model setup significantly.

In Cybernetic modeling, cells are thought to aim ~at metabolite homeostasis . while
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growing and adapting to their environment in an optimal way. The term homeostasis
here means almost constant intracellular compound concentrations, a phenomenon
often observed in microorganisms . The reaction network is considered as a network
with multiple feedforward and feedback control mechanisms . Assuming optimality
constraints allows simplifications of the reaction system because details of reaction
kinetics are eliminated. In the illustrative example the dynamics can be described
by

C%=Ni,'t-v-I-N..-v

	

(3.22)

where Nint is the stoichiometric matrix describing network reactions or condensed
pathways . Nex describes exchange fluxes with the medium, i.e . uptake or excretion
processes . v = (vl , . . .vn,) is the vector of fluxes .

The cybernetic modeling approach in general does not formulate equations for the
fluxes vi - even though some fluxes might be explicitly formulated . Instead, in
order to solve the model system, constraints are introduced . Such constraints are
e.g . a maximal growth rate, maximal production of structural compounds, maintain
homeostasis, or maximum GLC uptake rate . Mathematically this is expressed by
minimizing or maximizing one or more ofthe fluxes vi in Equation 3 .22 . The problem
is thus reduced to a solution of the system of linear Equations 3 .22 . If the system
contains some explicitely formulated fluxes, these appear as nonlinear constraints of

the linear system .

In the illustrative example a constraint would be e.g . a maximal flux through

the pathway. Without further constraints, however, there is no limit to the so-

lution and the maximal flux is infinity. This illustrates the need for additional

constraints in cybernetic models . An additional constraint could be a fixed value

for the flUX VPEP-3PEPout . Then the maximal flux through the pathway is limited .

Cybernetic models provide a tool for describing and analyzing cellular networks

without explicitly formulating rate equations . However, finding realistic constraints

that are necessary for system analysis is the crucial point of cybernetic modeling .

3 .4 .3 .

	

Pool lumping to reduce model dimension

A system describing metabolic regulation can be simplified by eliminating compound

concentrations from the model . The approach sets up complex models (possibly also

with complicated reaction rate equations) and then reduces the dimension of the

model by elimination of compounds.
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In [103] a model of the main metabolic pathway of Saccharomyces cerevisiae is

shown. The model consist of 97 differential equations and 83 rate equations . This

system is reduced by excluding pseudo steady state metabolites . These are metabo-
lites that have high turnover rates . The accumulation of such metabolites can be
neglected and these metabolites eliminated . Additionally, a time scale separation is
performed in order to reduce the system further . Very fast and very slow processes
can be eliminated or pools lumped together to a single pool. As a result, a system
of 27 differential equations with 15 reaction rates results .

Elimination of compounds to reduce model dimension requires a significant amount
of knowledge about intracellular behavior . If this knowledge is available, however, a
significant reduction of the problem dimension results .

In the illustrative example, the reactions from FBP to PEP could be eliminated,
see Figure 3.5 . This results in four ODEs instead of five and reduces the number
of rate equations also from five to four . The overall flux through the pathway may
then be described with the same accuracy as with the complex model. However,
other reaction rates in the system that are influenced by FBP cannot be modeled
correctly after elimination of FBP from the system .

F6P

Figure 3.5. : The simplified illustrative example . The FBP concentration is elimi-
nated from the model .

3.4 .4 .

	

Top Down Approach: conceptual modeling

In order to make modeling of a system easier and more obvious, cells can be dividedinto functional units . As described in [55], units consist of cellular processes thatare functionally linked . These units can be described in detail or by using a black

50



3. 5. MODEL DETAILS AND PROBLEMS

box model. The approach aims at representing the complete cell including genetic
and metabolic regulation. It allows to model parts of the cell in a simple way if
no detailed information is available . The idea is combined with a software concept
that allows to model functional units with different levels of detail by exchanging
mathematical modeling objects .

In the illustrative example a first model could summarize the reactions from F6P
to PEP in a functional unit called glycolysis, see Figure 3 .6 . Notice that the func-
tional representation lacks the ATP consumption and the inhibition by PEP. Instead
the flux through glycolysis could be regulated by the available substrate mix: if GLC
is available only, the flux through glycolysis is high. If the cells grow mainly on e.g .
Acetate (ACE), the flux through glycolysis is lower or even reversed because Gluco-
neogenesis may be active .

Substrate
Mix Glycolysis

3 .5 .

	

Model details and problems

Figure 3.6. : The simplified illustrative example . The reactions from F6P to PEP
are summarized as a functional unit "glycolysis" .

In a top down approach more details can be added to the model, finally replac-

ing the functional unit "glycolysis" by the individual reaction steps . This approach

allows to estimate the importance of detailed representation of functional units by

analyzing the différences between models with différent levels of detail . If no differ-

ences between a simple and a detailed model can be seen, the detailed description

can be eliminated from further models.

As described in thé last section, thé models of this work are mechanistic models

based on thé réaction network . Three important points of such an approach are

discussed in more detail in this section due to their importance :



CHAPTER 3. THE MODELING CONCEPT

1. Pathway selection: It is not possible to include the complete pathways in a

model. How to choose and how reliable is the available information from

textbooks and in public databases on the Internet?

2 . Kinetic reaction rates: Metabolic regulation is based on enzyme kinetic reac-

tion rate equations . How to find a reasonable enzyme kinetic equation for this

complicated mechanisms?

3. Additional efPectors of kinetic reaction rates: Many effectors influence the

enzyme reaction rate . How can these effectors be included in a kinetic rate

equation?

3.5 .1 .

	

Pathway selection

The knowledge of the underlying pathways of microorganisms is necessary for simu-

lation and biological interpretation. Even though Escherichia coli is a well studied

organism, identification of pathways is not finished . One reason for this is the fact

that many enzymes catalyze more than one reaction and therefore can accelerate

bypass pathways . Also, some pathways are active under so far unknown conditions
like the methylglyoxylate pathway from DHAP to PYR ([651, chapter 14) .

Of greater importance, however, is the fact that many metabolites are relatively
unstable, making it difficult to measure them. Therefore the actual Escherichia coli
metabolic pathway could be much more complicated then the recent understanding
suggests . It is important to take these effects into account throughout model devel-
opment . A model should not just be able to model certain situations with a fixed
pathway model but be flexible in the whole setup of its components and structure .
Wherever models are presented in this work, a discussion of the currently avail-
able knowledge on Escherichia coli pathways, enzymatic regulation phenomena and
the developed mathematical model will clarify some problems that occurred during
analysis .

With the help of MMT Software that allows to build alternative models of a path-
way in a matter of minutes a large amount of models could be created for this work,
Analysis of these models can give hints which pathway model is the correct one.
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3 .5.2.

	

Kinetic reaction rates

Enzymatic reactions are simple in principle (Figure 3.7) . Substrates bind on enzymes
and are released in a différent chemical composition as products of the enzymatic
reactions . A Gloser look at the enzymatic structure, process of binding and dissoci-
ation, however, reveals a complicated picture (see e.g . [21], [94]) that is summarized
in this section . The phenomena mentioned here are used to derive enzyme kinetic
rate equations. In the models discussed later, such equations are introduced and

used.

association
(binding of compound

to an enzyme)

dissociation Enzyme-Compound
removal of a compound

	

Complex
from an enzym

Figure 3.7. : In enzymatically catalyzed reactions, one or more compounds (sub-

strates) bind to an enzyme (association) and one or more compounds

(products) are released (dissociation) .

Conformation : The three dimensional structure of enzymes cannot be analyzed or

calculated easily. Without knowledge of the exact structure it is only possible

to estimate what kind of substrates can bind on what site of the enzyme .

Number of binding sites : It is often not known how many binding sites an en-

zyme possesses, see Figure 3.8 for an example . Occurrence of several binding

sites can significantly accelerate the reaction rate per enzyme molecule.
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Effector
(Metabolite,
Nucleotide,
Cofactor)

Inhibited/
Activated
Enzyme

Figure 3 .8 . : An enzyme that has already a compound attached has a différent
probability of binding to another compound. Many other effectors not
shown as e.g . other compounds present in thé tell and temperature or
pH influence probability of association and dissociation .

Probability of association (binding) and dissociation :

	

Parameters for thé
probability of association and dissociation are very dif5 .cult to measure . Also,
for more complex enzymes, thé catalyzing effect on a réaction rate is usually
not multiplied by thé number of binding sites of thé enzyme. Instead, thé
association and dissociation constants change .

Reversibility : Most processes are reversible, i.e . products tan bind to thé enzyme
and substrate tan be released from thé enzyme-substrate complex .

Order o£ binding: In multi substrate or multi-product cases it is sometimes im-
portant what substance binds first to thé enzyme.

Cofactors : Compounds that bind to an enzyme tan change association of dissoci-
ation constants for thé binding of compounds to thé saure enzyme.

associ-
ation

dissoci- BE', dissoei-
ation m: ation Enzyme-

Compound-
Effector-

Compound Complex
(Metabolite,
Nucleotide,
Cofactor)

associ-
ation
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Macromolecular crowding: It has been shown in vitro that a high concentration
of certain large compounds can increase reaction rate of enzymatic reactions .
This phenomenon is called macromolecular crowding [105], [83] .

Metabolite channeling: Under yet unspecified circumstances a large fraction of a
specific enzyme can be bound in the enzyme-substrate-product complex leav-
ing only a small fraction of free enzyme, substrate-enzyme, or product-enzyme

complexes . It is thought that a large fraction of enzyme bound in the enzyme-

substrate-product complex increases the reaction speed. This effect is termed

metabolite channeling. Metabolite channeling is connected to conditions of

macromolecular crowding [83] .

Most of the mentioned parameters and effects have to be estimated from in vitro
experiments . It is then not clear if the same ef ects occur also in an in vivo environ-

ment.

3 .5 .3 .

	

Additional effectors of kinetic reaction rates

Other effects influence the association and dissociation constants of enzyme binding .

e.g .

" the concentrations of substrates, producs and cofactors

enzyme concentrations

" compartmentation of tells and distribution of enzymes, substrates and other

compounds inside the tell

temperature, pressure, pH

The number of effects that cannot be quantified makes approximations necessary.

Usually only one or two effectors of enzymatic reaction rates are included in the

rate equations used in this work. The equations are given in Chapter 6 where some

models of the Escherichia coli pathway are presented .
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4. The MMT Software

In order to build complex models describing the pathways of microorganisms, soft-

ware tools are required . Such tools allow to construct and analyze models faster

as if models are build manually from scratch . They also help to keep an overview

of complex pathway models that have been setup and results from simulations and

parameter fitting . This chapter describes the advantages of a modeling tool that is

based on a database system and introduces the structure and functioning of such a

tool: the Metabolic Modeling Tool (MMT) developed for tfiis work .

4.1 .

	

The need of efficient modeling software

There is a number of obvious reasons for using software tools for modeling metabolic

networks in the way used in tfiis work. Some are :

" Tfie models consist of a large number of chemical reactions that can be defined

once and used for several model setups .

" A graphical interface is important to keep an overview of the network of chem-

ical reactions . This interface should be combined with structuring techniques

for the network in order to allow users to explore the networks .

" Parameter values can be defined easier if they are organized nicely, e.g . in a

table, by the software .

" Results can be viewed directly in a structured way after simulations or param-

eter fittings have been performed .

" Setup of the mathematical equations in a format useful for numerical solution

algorithms can be performed by software tools .

Various approaches have been made to build tools that address the above points .

The following sections give an overview of available tools . The List is not complete

but captures the main approaches .

57



CHAPTER 4. THE MMT SOFTWARE

4.1 .1 .

	

General modeling software for dynamical systems

A large number of software is available for simulating dynamical systems . Three
examples are mentioned here that show the variety that exists . Each tool has its
specific capabilities and lacks others, which makes application for the purpose of
this work impossible or inconvenient .

MODELICA [68] is a tool for models based on Differential Algebraic Equations .
It includes algebraic algorithms for an automated model simplification . The
tool lacks a graphical user interface and algorithms for parameter fitting . It is
free of charge.

" SIMULINK is a MATLAB [99] extension that allows to build dynamic models . The
tool allows model setup with a graphical user interface . Simulations and pa-
rameter fitting algorithms are available . SIMULINK and MATLAB are commercial
software tools .

" MAPLE [107] is a software focused on algebraic methods . It eau simplify model
equations and perform model analysis with algebraic models . Solution algo-
rithms for ODES are available but parameter fitting is not possible . MAPLE is
commercial software .

4.1.2 .

	

Modeling software developed for chemical reaction systems

A number of software tools have been developed that focus on chemical reaction
networks . These tools allow definition of chemical reactions and rate equations for
the reactions . Some tools include predefined rate equations . A short overview is
given here .
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" MetaModel [23] is an early development of a metabolic simulator . It contains
algorithms for solution of ODEs. It is limited to 20 reactions and 30 metabo-
lites . The project is discontinued.

" METATOOL [72] is focused on structural properties of metabolic networks and is
limited to steady-state analysis . It calculates the null space o£ the steady-state
equation N - v = 0 and performs system reduction .
MIST [29] is a metabolic simulator with a graphical user interface . It is limited
to calculation of control coefficients . The project is discontinued .

" ESSYNS [49] is a tool that focuses on analysis of models in S-system represen-tation . The latest version dates from 1991 .
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METASIM [97] is another early development . It bases modeling of cell regulation

on an object-oriented approach . The latest version dates from 1992.

KINSIM and FITSIM [6], [25] are tools for setup of chemical reaction systems

and for parameter fitting . The model is setup in a notation similar to a pro-

gramming language . The tool lacks a graphical user interface and is limited

to 100 species and 100 reaction steps . The software is free of charge but no

further development is taking place (latest version from 1997).

" E-cell [100] is an integrated modeling approach, i.e. it includes genetic regu-

lation, metabolism and signaling phenomena. It provides a graphical overview

of the pathway and numerical algorithms for solving the system of ODEs. It

lacks algorithms for parameter fitting . The software is free of charge and an

ongoing project .

SCAMP [85] is a tool developed for chemical reaction networks . The tool is com-

mand line driven and is not limited in the number of reactions and metabolites .

It provides e.g . solution algorithms for the system of ODEs and algorithms

for steady state analysis, model reduction, calculation of control coefficients,

and calculation of the Jacobian. It lacks parameter fitting algorithms. The

software is free of charge and its development has halted . The tool is succeeded

by JARNAC.

JARNAC [86] is the continuation of the SCAMP project . It provides the saine

functionality as SCAMP and additionally includes a graphical user interface .

JARNAC is an ongoing project .

DynaFit [57] is a tool for performing least-square regression of chemical kinetic,

enzyme kinetic, or ligant-receptor binding data . The tool provides a graphical

user interface and contains algorithms for solution of ODES and parameter

fitting . Simulation results (time-courses) are presented in graphical form . The

tool does not provide an interface for easy setup of large chemical reaction

systems and is therefore more suitable for small systems . No information is

available if the project is ongoing .

" DBsolve [37] is a powerful tool for complex chemical reaction networks . It

allows solution of non-linear algebraic equations and systems of ODEs. Addi-

tionally, bifurcation analysis of ODES and parameter fitting to experimental

data can be performed. The tool has an graphical user interface . It is severely

limited in the dimension of the reaction system : a maximum of 50 ODEs and

50 parameters can be used only. It is an ongoing project .

" GEPASI [63] is another powerful too1 for complex chemical reaction networks. It

includes algorithms for solution of ODEs and parameter fitting . Additionally,

metabolic control and stability analysis can be performed . The tool lacks an
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overview of the chemical reaction network which makes it difficult to define

large models . The latest version dates from 1999 .

4.1.3.

	

Criteria for an efficient modeling software

In addition to provide graphical interfaces for model setup and result interpretation,

modeling software useful for large chemical reaction systems should contain more

features . Efficiency of modeling software increases with the following automated

features :

" Simplifications of the mathematical model using algebraic methods .

" Translation of the model equations into a procedure of a programming lan-

guage in order to get better performance during simulation and parameter

fitting .

" Extensions of the model by e.g . explicitly calculating derivatives of model

equations with respect to parameters.

Another important aspect of modeling software is the combination with a database

system . In the field of genomics, database systems are essential for all analysis due

to the amount of available data. In contrast, use of databases for chemical reaction
kinetics including explicit model formulations and experimental data is still rare.
All above mentioned modeling tools are not based on database systems . However,
database systems have some big advantages also for applications in this field:

" an increase in storage efficiency because relations avoid redundant entries

" a systematic approach to data organization and storage

" a data consistency that cannot be maintained without use of database systems

" an easier reuse of information from other models already developed

" an easier storage of results

The need to build software tools on database systems that include chemical reaction
kinetics has been generally realized and software systems are already in the devel-
opment phase [32], [112] . There are also approaches to define a formal modeling
language similar to the SQL language for relational databases and combine such a
modeling language with a relational database system [74] .
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4 .1.4.

	

What is thé différence between available software tools and
MMT?

There have to be good reasons to start programming a software that somebody
else already made available in a similar way. The MMT software was developed
due to limitations in thé available software tools mentioned above that have been
considered too severe .

General modeling software as MODELICA or SIMULINK has been rejected because
this software is not specifically for chemical reaction networks . This makes it diffi-
cult to keep an overview of complex chemical reaction networks that are included
in thé model . Only four of thé above mentioned tools specific for chemical reaction
networks allow parameter fitting to experimental data: KINSIM/FITSIM, DynaFit,
DBsolve and GEPASI . KINSIM/FITSIM does not have a graphical user interface and
is limited to 100 compounds and reactions . DynaFit and GEPASI are not convenient
for large chemical réaction networks because they do not provide pathway overviews .
DBsolve provides such an overview, but it is severely limited in thé model dimen-
sion: 50 ODEs and 50 parameters are not enough for complex pathway models of
microorganisms .

The most important limitation common to all of these points that lead to thé
decision to develop thé MMT software is thé database connection . DBsolve allows
to import information from thé WIT [113] database . This information is limited to
thé réaction network, i.e . thé stoichiometric matrix . None of thé above mentioned
tools are connected to a database system that stores complete model information,
parameters, and results .

4.2 .

	

Global structure of thé MMT software

Simulation software nowadays subdivides modeling into pré-processing, numerical

algorithms, and post-processing [17] . Pre-processing is concerned with thé model

setup, post-processing means analysis and interpretation of results . The MMT soft-

ware is based on thé same approach. A general overview of thé software architecture

can be seen in Figure 4.1 . Figure 4.2 shows thé user interface for representation of

thé chemical reaction network and Figure 4.3 thé window for setup of a kinetic rate

equation for a chemical reaction .
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Figure 4.1 .: Global structure o£ the MMT Software . Red/italic text indicates a
software library, external software, or external algorithm used during
a specific task . Vertical arrows between the processes indicate the
direction when building new models . The random number generator is
called during parameter fitting and to addrandom noise to simulation
output .
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Reaction Details

-- Effectors :
6-phosphofructokinase : ID 2.7 .1 .11

-- Kinetics on this reaction :
Hill kinetics

	

edit

Figure 4.2. : The main window of the MMT user interface provides an overview of

the reaction network. The left part of the window shows the individual

reactions ordered by reaction groups (Glycolysis, PTS) and compart

ments (cytoplasm, extracellular) . The right part of the window shows

the details of the reaction currently highlighted in the reaction tree .

"Effectors" are all compounds that influence the reaction (enzymes,

inhibitors etc.) but are not included stoichiometrically in the reac-

tion. A click on the "edit" button in the "Kinetics" section opens the

window for the kinetic rate equation defined. This window is shown

in the following Figure 4.3 .

- c oplasma F6P -> F1,6P; 2.7 .1 .11 (5)
- PTS

É-Glucose Uptake (1)
- GlyCo lysis

-G6P -~ F6P, 5.3.1 .9 (2)

`-F1,6P -> F6P; 3.1 .3 .11 (4)
ro-F1,6P -> GAP + DHAP (6)
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Reaction
Reaction

byATP
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`ô
ö
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r F6P -> F1,6P: 2.7.1 .11
HIII kinetics
PV"(PS1^Ph)f((PKos^Ph)+(PS1 ^ Ph))

dep : dependentvariable ofODE system, give initial condition
die : dependentvariable, optimize Initial condition, give Inital and range
data : values used from a data column, specify initial for simulation
opt : optlmized parameter, give Inital and range
flx : fixed parameter, gave value

r- Values for Opti5 and other Optimizations

Figure 4.3 . : The window where details of equations for chemical reaction kinetics
are defined . Shown is a Hill kinetic for the reaction from F6P to F1,6P.
The upper part of the window displays the equation and explanations
for the possible definitions of variables and parameters . These defini-
tions can be seen in the lower left part of the window. The lower right
shows initial conditions and ranges for variables and parameters and
corresponding values from other parameter fittings (" Optimizations" )
with the saine model .

The figure shows that the user interface on the left hand side is used to control
all processes from model construction to simulations and parameter fittings . Some
of the subprocesses are performed with external software and numerical algorithms.
A main point of the whole system is that it is based on a relational database that
contains all information about the model and all results . Thus, there is a unique
point where all relevant information is saved in an efficient way.

All external software and algorithms used by the MMT software except for MAPLE
are free of charge for research purposes . Additionally, all parts of the software are
available at least for the two important operating systems LINUX and MICROSOFT
WINDOWS .
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4 .3 .

	

The MMT database

The concept of the MMT software is based on a relational database where complete
model information, simulation and parameter fitting values, and results are stored .
All other parts of the software are build around this database . They have the purpose
to make use of the stored information for model definition, solution of ODES or the
parameter fitting problem, as well as representation of results .

The database system was setup for the specific problem of modeling chemical
reaction systems . It stores such networks in a set of tables . The part of the database
that contains the mathematical models is focused on the specific model formulation
introduced in Chapter 3.

4.3.1 .

	

Structure of the database

For the given problem, the relational database turned out to provide a good struc-
tural background for efficient storage of the required information . The complete

database was designed with an entity relationship diagram [62] . The diagram is too
compeex to be shown here. Setup of the database was then performed according

to the Third Normal Form for relational databases [62] . Therefore, an efficient and

consistent storage is guaranteed already from the architecture of the database. The

structure of the MMT database is divided into three levels that will be explained in

detail in this Chapter . The three levels are (substantially described in more detail in

the next sections) :

1 . Groups of Models: The database contains various groups of model structures,

usually referred to as the modeling groups level of the database.

2. Models: Each group of models can contain différent models . This is the model

level of the database .

3 . parameter fittings : For each model, various simulations and parameter fit-

tings (parameter fittings) can be performed . This is usually referred to as the

parameter fitting level of the database .

Figures 4.4 and 4.5 show the three levels for the illustrative example . Division of

the database into these three hierarchical structures increases efiiciency and allows

easy creation of new models .
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Figure 4.4 . : The group and model levels of the database for the illustrative exam-ple .
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Figure 4.5 . : The parameter fitting level of the database for the illustrative example .

The modeling groups level is the highest level of the database system. It is called

modeling groups level because it stores information that is used by a collection of

models in the next lower level, the model level (explained in the next paragraph) .

The group level contains all information about a chemical reaction network . In

detail, the tables belonging to the level store information about

" the chemical reaction network, i.e . the chemical reactions with educts, prod-

ucts, enzymes, and effectors in a structured way. This means that the reactions

are grouped by location which could be e.g . cytoplasm, extracellular or mito-

chondria .

" reaction groups of the network which are used to divide a large network in

smaller parts . E.g . a network can be divided into glycolysis, PP-Pathway, etc .

spatial compartments of the model, these are e.g . the cytoplasm, mitochondria,

but also the extracellular environment (fermenter) .
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. data files related to compounds, i.e . a relation between a compound of the
network and a column in a file containing experimental data.

. a collection of mathematical formulations for chemical reaction kinetics that
can be related to a chemical reaction if a specific model is defined .

Figure 4.6 gives an overview of the modeling groups level. Notice that the mathe-
matical formulations for chemical reactions are not yet related to a chemical reaction
but are only possible formulations . Figure 4.7 shows how two reactions of the ex-
ample model (influx to F6P and reaction from F6P to FBP) are stored in the group
tables .
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Figure 4.7. : Entries of tables of the modeling groups level for the illustrative exam-ple . Not all tables, columns, and entries are shown . Arrows indicaterelations between the tables .
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The model level

See Figure 4.8 for an overview of the model level .

	

The model level inherits all
information stored in the tables of the modeling groups level. The model level adds
to the information stored on the modeling groups level merely relations between
mathematical formulations of the table QUANT and chemical reactions stored in
the table NET. Mathematical formulations for the rate equations can be added to

each chemical reaction step . However, not every chemical reaction needs to be part

of a specific model.

If a reaction network is defined in the modeling group tables, a new model of this

complete network or parts of this network is defined by two tables that contain

only relations between existing tables . For the purpose of building a large number

of différent models for comparison, this setup therefore allows efficient storage and

quick definition of new models. It also guarantees consistency between thé models

because all information is stored in common tables on thé modeling group level and

therefore is consistent between all models of a group. Figure 4.9 shows parts of thé

entries for thé illustrative example model. It can be seen that thé table KINETIC

relates each variable of a kinetic equation to a compound or parameter .
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Tables on the model level

Tables on the modeling group level

Figure 4.8 . : The, tables of the model level and the relations to the tables of the
modeling groups level .
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Table KINETIC

''Table
QUANT '

Figure 4.9. : Entries of tables of the model level for the illustrative example . Not

all tables, columns, and entries are shown . Arrows indicate relations

between the tables . In table KINETIC compound IDs with value -1

represent parameters whereas values > 0 represent compounds out of

the table POOL.
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The parameter fitting level

The third level of the MMT database deals with the parameter fitting of the defined

models. The purpose of the database development was the usage for simulation

and parameter fitting. These two tasks require some additional information that is

stored in the tables o£ the parameter fitting level (Figure 4.10) :

. information about variables . Options are dynamic, fixed or use values from

experimental data .

. information about which initial conditions and parameters are fixed and which

are subject to parameter fitting .

. definition of ranges for optimized initial conditions and fitted parameters.

. definition of weights for dynamical variables subject to parameter fitting.

Weights will be explained in Section 5.3 .

. definition of the data file used for a specific parameter fitting as a relation to

the table DFILE on the modeling groups level .

" definition of parameters for the simulation and parameter fitting tasks (which

algorithms, tolerance parameters, etc.) .

Storage of all these variables and parameters including all starting parameters and
algorithm specific parameters allows the reproduction of results . In addition to
these information, results are stored in the four results tables RESVAR, RESERR,
RESSIM and RESFLUX. Notice that this information is redundant because, as
just argued, reproduction of results is always possible . However, reproducing results
requires computation time that is usually more limiting than the disk space required
for storage of results .

It is advantageous to allow definition of several parameter fittings for one model.
For a defined parameter fitting, an unlimited amount of simulations and parameter
fittings can be performed . This means that an parameter fitting algorithm can be
started several times with varying initial conditions, parameter values and ranges .
Additionally, different parameter fitting algorithms can be chosen and parameter
values specific for such an algorithm can be changed . All initialization parameters
of the algorithms are stored in RESINIT, results are stored in the four result tables
RESVAR, RESERR, RESSIM and RESFLUX . The user defines how many results
of a single parameter fitting are stored in the result tables . Most of the Limes it
is sufficient to store the final result of a parameter fitting and not all intermediate
results .
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RESSIM

Parameterization
results :
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Figure 4.10. : The tables on the parameter fitting level and their relations to tables
from the other two levels of the MMT database. Tables FUNC and
DIFFNOT are automatically created for construction and export of
the model as a C-code.
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However, a new parameter fitting has to be defined if the structure of the parameter

fitting problem is changed. This can result from changing either the number of

dependent variables or the number of optimized initial conditions or parameters .

Assume that the properties of a parameter v,,,,,, are changed from "optimized" to
"fixed" . This changes the total number of fitted and fixed parameters of the model.
It removes an entry in the vector offitted parameters and adds an entry in the vector
of fixed parameters . Because initial values of parameters or parameter fitting results
are stored in such vectors, changing the dimension of these vectors would result in
inconsistencies in the result tables .

Storage of results was setup in such a way because it is more efficient . Also a clearer
structure of the database results if new tables on the parameter fitting level are used
for a new mathematical structure of the model . Especially the tables RESFLUX
and RESSIM would otherwise be very large and it would be time intensive to extract
a specific result from these tables .

Figure 4.11 shows the tables DIN and PIN on the parameter fitting level for the
illustrative example . The tables with the results are not shown. They consist of
simple lists either of parameter values or of compound concentrations for each time
step .

The relational database System MySQL

The relational database system MySQL was used for implementation of the database
structure . MySQL is a commercial software system [101] . It is, however, free for use
in science and education . The database system MySQL
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is a relational database that supports the standard query language SQL

is easy to install

has no limitations in number and sizes of tables

" is a full multi-user database that can be installed on a server and used by many
users simultaneously

" can be accessed from other programming languages as e.g. C and Tc1/Tk

The database is currently installed on a system with an Intel Pentium III processor
with 800 Mhz. The current size of the database content is 0.78 Gigabyte. About0.75 Gigabyte or 96% of the total are results. The defined 50 Models that are storedin the database take up the remaining 4% (30 MB) of the storage .
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Table POOL

Table DIN
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Figure 4.11 . : Entries of tables of the parameter fitting level for the illustrative

example . Not all tables, columns, and entries are shown . Arrows

indicate relations between the tables .
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Because an increase in the overall size of the database results in bad performance
of the MMT, a change of the result storage is anticipated in the next version of
MMT. It has been axgued above that storage of all results in the database saves
computation time . However, it has become clear now that this computation time is
lost again during database access . Therefore it is more efficient to use the available
parameters in RESINIT and RESVAR and perform. a simulation to obtain the time
courses stored in RESSIM and RESFLUX .

4.4.

	

The MMT user interface

A view at the structure of the MMT database presented in the last section shows
that a user interface is essential for modeling . This interface has to provide an
overview of all the defined models, mathematical details, simulation and parameter
fitting settings, and results . It also has to help the user to quickly define new models
and parameter fittings and show results in a way that allows an interpretation . The
user interface is also needed to guarantee consistency of the database entries .

The MMT user interface is coded in the script language Tcl/Tk . With use of the
publicly available library MySQLTcl [78] SQL commands can be send directly from
Tcl/Tk to the database and results can be stored in Tcl/Tk variables . Widgets from
the publicly available extension library Tixwish were used . This library provides
easy to use widgets that could be included in the Tcl/Tk code of the MMT user
interface .

Some advantages and disadvantages of the script language of Tcl/Tk are :

- Future development of Tc1/Tk has become questionable

- No object oriented language

- Cryptic commands require detailed documentation

- Low performance

+ Widely distributed and free of charge

+ Extension for access of SQL databases available

-!- Supports LINUX and MICROSOFT WINDOWS

+ Easy to program, especially for graphical user interfaces
+ Many extensions as Tixwish available that save programming time
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The MMT user interface consists of a main procedure and 40 subroutines for the
individual tasks (overall number of lines approximately 24000) .

The following sections describe the options and tasks that are handled by the
MMT code. The tool only stores information after performance of a specific task .
All information is checked for consistency and immediately written to the MMT
database .

4 .4.1 .

	

Setup of a modeling group

To setup a modeling group, compounds and reactions have to be written into the
database . This is mainly done by a feature "Add Reactions" . This feature shows in
a table all available compounds of the model . The user can click on a compound to
include it in the chemical reaction . New compounds can be defined also . If a new
compound is defined, the software asks for other details of this compounds in order
to get all required information for consistency of the MMT database . Double entries
are rejected, but still a compound can exist several timms - e.g . a system will have
to include OAA in the cytoplasm and the mitochondria .

After all educts, products, and effectors of a reaction have been typed, the global
properties of a reaction have to be defined . Required fields are a naine and a reaction
group (e.g . glycolysis) . Figure 4.2 above shows a screen shot from the overview of
the chemical reaction network as provided by the MMT user interface .

Mathematical equations for the table QUANT can be defined separately. Equations
are stored in MAPLE notation which allows to copy equations from or to a MAPLE file .
As mentioned in the last section, the database stores relations between compounds
and columns in external data files . These data files contain the experimental data .
The MMT software analyzes the data file and asks the user to assign a compound
to each column of the data file . The data file is then sorted with respect to time
and a new external data file is produced. This data file can be used from all models
of the modeling group, the information which column belongs to which compound
is stored in the table DATPOOL.

After completion of the definitions on the group level, various models can be created
and analyzed . Notice that certain restrictions apply to changes in the modeling
group tables after models have been defined:

" no reactions that are part of a model can be deleted .

" stoichiometric coefficients can be changed only if no results are stored that
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rely on these coefficients . Change of stoichiometric coefficients is sometimes

useful, e.g. for energy costs of groups of reactions . For most reactions the

stoichiometric coefficients are fixed .

. compounds and effectors can be added to reactions but they might make results

unusable .

Two strategies should be used in order to avoid problems that could result from

these restrictions :

1 . Include all reactions needed from the beginning in the modeling group.

2 . If changes or additions have to be made (e.g . inclusion of a new pathway),
copy the complete modeling group information to a new modeling group (this

is automatically done) and make the changes in the new group.

4.4 .2 .

	

Define models

In the MMT database structure models are represented by relations between re-
actions and mathematical formulations . Therefore, defining models for a modeling
group can be done by choosing a mathematical equation and a chemical reaction
from tables produced by the user interface . After an equation and a reaction have
been related, the software asks the user to specify which parameter in the equa-
tion corresponds to a compound in the reaction and which parameter is really a
parameter . E.g ., if the Michaelis-Menten equation

C
v = vmax ' K., .+ C

is used, the user could define the C to be G6P and will define %ax and Km, as
parameters .

Often new models differ from existing models by only a few equations . The MMT
software allows to copy models, and then make these simple changes without a
complete redefinition ofthe system. A complete modeling group including all defined
models can be copied or a single model can be copied . If a single model is copied,
the copy remains in the same modeling group as the original one . This is a necessary
limitation because the tables on the model level are related to the models on the
group level and a différent modeling group possesses différent group tables .
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4.4 .3 .

	

Defining and performing simulations and parameter fittings

Analyzing and using models is a time consuming process . The MMT software helps
to efiiciently perform simulations and parameter fittings . For most tasks mentioned,
default procedures and parameter values exist that can be redefined by the user . In

particular, parameter values from any previous simulation or parameter fitting can

be reused . Either all parameter values from a previous result are copied or single

values can be copied by the user. Figure 4.3 above shows the window for the input

of a kinetic reaction rate for the reaction from F6P to F1,6P. In the lower part of the

window, the user defines the properties of the variables and parameters . Simulations

and parameter fittings require the following definitions and selections from the user:

1 . Define which compounds are not constant and should be included in the system

of ODES. In the illustrative example, F6P, FBP, ADP, ATP, and PEP have

been included as dynamic variables . Dynamic variables can also be defined

with optimized initial conditions which is important if no initial condition is

available from experiments or the available data is inaccurate .

2 . Define variables for which values from an experimental data file are used during

solution of the Initial Value Problem (IVP) . This is useful for compounds where

good measurements are available but that cannot or should not be included in

the model as dependent variables . This could be e.g . the GLC concentration

in a fermenter or the ATP concentration in a cell (the ATP concentration is

important but is difficult to model due to the large number of reactions pro-

ducing and consuming ATP) . Such input data can also be from interpolation

of experimental data. This results in smooth data representations .

3 . Define parameters to be fixed or to be subject to parameter fitting.

4 . Define initial values for all variables and parameters. Define ranges for opti-

mized initial conditions and fitted parameters .

5 . Choose a data file from the table DFILE that has been linked to the modeling

group. Experimental data is not stored in the database because the data

is constant for all models, simulations, and parameter fittings . Because the

numerical algorithms for simulation and parameter fitting require a text file as

input it is more efficient to keep this text file always available . Otherwise this

text file would have to be produced for each simulation and parameter fitting.

6 . Choose an error functional and weights for each compound where data values

are available .

7 . Define tolerances for solution of the IVP.
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8 . Choose an algorithm for parameter fitting and choose all settings specific for

this parameter fitting algorithm .

9 . Choose a method for sensitivity calculation that also allows to reduce the

parameter space to the most sensitive parameters.

10 . Define the number of parameter fitting runs that should be performed . One
run ends if the maximal number of function evaluations or the tolerante of the
fit is reached . The next run starts with randomly changed parameters. The
user needs to define the amount of random change of these parameter changes .

11 . Choose frequency of output : after every parameter fitting, only if an improve-
ment was made, or only at the end of all parameter fitting. These options
are available separately for the fitted variables and the error functional, the
dependent variables versus time, and the fluxes versus time.

4.5 .

	

Automated model generation by the MMT
software

Before the first simulation or parameter fitting with a model, the system of ODEs
for the IVP has to be setup. This setup is performed by the MMT software. Several
steps are necessary that are described here.

4 .5 .1 .

	

Extract information from the database to setup the IVP

The structure of the database system is described at the beginning of this chapter .
In order to setup an IVP the MMT software performs several tasks:

1 . It analyzes each compound that is part of the chemical reaction network . If a
compound is defined as a dynamic variable, an ODE has to be defined . This
ODE consists of several terms . In the illustrative example, the following terms
have to be included in the ODE for the F6P concentration :

F6P = Vinflux, F6P - VF6PHFBP

2. After the ODES for all compounds are defined, the fluxes vi have to be re-
placed by kinetic rate equations, e.g . Michaelis-Menten equations. For the
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F6P concentration in the illustrative example, this will result in the following
equation :

F6P = vinflux F8P - vmax,v2
F6P

	

FBP
Km,, v2 -{- F6P + vmax,v3 '

Km�, +FBP

3 . After the system of ODEs is setup, all variable and parameter names are
renumbered. Compound concentrations are replaced by the components of
a vector y, parameters by the components of a vector P . Then the above
equation for F6P could read (vector indices chosen arbitrarily) :

3'l = Pz -

	

P2 ' Yi + P4 « Y2
P3 + Yl

	

P5 + Y2

This step is necessary to avoid duplicate names in the systems of ODEs that
belong to différent kinetic rate equations . Because the equations stored in the
table QUANT are already in MAPLE 6 format, the replacement is the only
necessary step .

4 . Create output files in text format: the next step is to write the IVP to a text
file . This text file is kept in MAPLE notation . Therefore, MAPLE can be used
to convert the IVP into a C-code . Additionally, several text files are produced
with initial values and parameters for the simulation and parameter fitting
algorithms .

5 . The created C-code is linked into a simulator and optimizer that is described in
the next section . A program function results that can be used for simulations
and parameter fittings .

For the calculation of sensitivities or to use the gradient method for parameter
fitting, each differential equation has to be derived with respect to the initial condi-
tions and parameters it depends on (see Section 5.2.3) . This process is much more
complicated than setting up the IVP. To create the derivatives, a temporary table
is setup that includes all dependencies of the ODEs on parameters and initial con-
ditions . Then the derivatives required can be written in MAPLE format to a text
file . MAPLE is then used to explicitly calculate these derivatives . Figure 4.12 and
Figure 4.13 show examples of the files created by the MMT and passed to MAPLE.

The derivatives of ODEs are converted by MAPLE into a C-code . For the illustrative
example the ODE for yl has to be derived wich respect to P1 to P5 and with respect
to the initial conditions of yl and y2. MAPLE will have to calculate 7 derivatives just
for the first ODE. This shows that the C-code becomes very large if derivatives
are calculated explicitly. More details about the setup of sensitivity equations are

discussed in Section 5 .2.3 .
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with(codegen,C) :
f :- '[

	

vl
- ( 1 .00000000000000 * ( ( p[1841 * y[1]/ p[1861 - p[187] * y[3]/

P[1891 ) / ( 1 + y[1]/ p[1861 + y[3]/ p[1891 ) ) ) -
1 .00000000000000 * ( P[5561 * ( y[11 * y[21] - y[251 * y[22]/ p[561]
) / ( Y[1l * Y[211

	

* ( 1 + y[251/ p[5621 )
+ p[5631 * ( y[11 + P[5641 ) + p[565] * y[211 + p[556] / ( p[5661
p[561] ) * ( P[567] * y[251 * ( 1 + y[1]/ p[564] ) + y[221 * ( p[568]
* ( 1 + p[565] * y[21]/ ( p[564] * p[5631 ) ) + y[251 * ( 1 + y[21]/
p[5691 ) ) ) ) ) ) + ( 1 .00000000000000 * ( p[8461 / ( 1 + y[71/
p[848] )^ p[8491 * ( p[8501 * z[271 * Y[21 - p[8531 * Y[7l * y[11 ) /
( 1 + p[855] * z[27] + p[8561 * y[2] + p[857] * y[2] * z[27] + p[8581
* Y[11 + p[8591 * y[71 + p[8601 * y[1] * z[27] ) ) )

+ ( 1 .00000000000000 * ( ( P[49] * y[11]/ P[511 - p[52) * y[2]/
p[541 ) / ( 1 + y[11]/ P[511 + y[21/ p[54] ) ) ) - ( 1 .00000000000000
* ( P(55) * ( y[2] * Y[9] - y[7] * Y[8]/ p[60] ) / ( Y[21 * Y[9]* ( 1 + y[71/ p[611 ) + P[62] * ( y[21 + P[631 ) + P[641 * y[9] +P[55) / ( p[65] * p[601 ) * ( P[661 * Y[71 * ( 1 + y[21/ p[63] ) +
Y[81 * ( p[67] * ( 1 + p[64] * y[9]/ ( P[631 * p[62) ) ) + Y[71 * ( 1+ Y[91/ p[68] ) ) ) ) ) ) + ( 1 .00000000000000 * ( p[4441 * ( y[81
Y[71 - Y[2] * Y[20]/ P[4491 ) / ( Y[8l * Y[7]* ( 1 + y[21/ p[4501 ) + p[4511 * ( Y[8] + P[452] ) + P[453] * y[71 +P[444] / ( P[4541 * P[449) ) * ( P[4551 ,* y[2] * ( 1 + y[8]/ p[452] )+ y[201 * ( p[456] * ( 1 + p[453] * y[7]/ ( p[452] * p[451] ) ) +Y[21 * ( 1 + y[7]/ p[457] ) ) ) ) ) ) + ( 0 .00000000000000 * ( p[693]* Y[8]/ ( Y[81 + P[6951 * ( 1 + y[9]/ P[697] ) ) * y[31/ ( y[31 +p[6991 * ( 1 + y[8l/ p[7001 + y[9]/ p[701] + y[201/ P[703] ) / ( 1 +Y[9l/ P[7041 + y[20]/ p[7051 ) ) * 1 / ( 1 + ( 1 + y[21/ p[7071 )^4p[7081 / ( 1 + y[3]/ ( P[699] * ( 1 + y[81/ p[7001 + y[9]/ p[701] +Y[20]/ p[7031 ) / ( 1 + y[9]/ P[7041 + y[201/ P[7051 ) ) )^8 ) ) ) -( 1 .00000000000000 * ( ( p[7521 + P[753] * y[121 + p[755] * y[4] +P[7571

	

y[121

	

y[41 ) / ( 1 + P[7581 * y[121 + p[7591 * y[4l )p[760] * y[2]/ ( p[7621 + y[21 ) ) ) + ( 1 .00000000000000 * ( ( 1 -Y(8) - P[7641 ) * P[7651 * P[7661 * y[8l ) * ( P[767] * y[8] * y[13]- P[769*

	

y[9*

	

y[2] ) / ( 1 + p[7721 * Y[81 + p[7731 * y[13] +p[7741

	

Y[8l

	

Y[13l + p[775) * y[2] + p[7761 * Y[91 + p[777] * Y[21Y[91

	

*

	

)

	

)

	

-

	

(

	

1 * 00000000000000

	

*

	

(

	

p[8461

	

/

	

(

	

1

	

+

	

y[7]/

	

p[8481

	

)^P[8491

	

( p[8501
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p [11 5+ *
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*
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+ p[858]p[8591
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Figure 4.12. : Example of a MAPLE file with the fluxes created by the MMT. Thefunctions v1 and v2 are shown. The total model dimension of theexample was 26 .



4.5 . AUTOMATED MODEL GENERATION BY THE MMT SOFTWARE

C([dy[1 ] =ynew[1 ]],
filename="/home/hurlebau/ModellingOptimize/modgroup20/mode119/opti2/odeP1
ainExt .c") ;
C([dy[2 ] =ynew[2]],
filename="/home/hurlebau/Modellingoptimize/modgroup20/mode 119/opti2/odeP1
ainExt .c") ;

C([dy[283 ] =ynew[283 ]],
filename="/home/hurlebau/ModellingOptimize/modgroup20/mode119/opti2/odeP1
ainExt .c") ;

Figure 4.13. : Example of a MAPLE file with command lines for calculation of sensi-

tivities (top) . All rates vi (called fi in the graph) need to be derived

with respect to initial conditions and parameters (yo to Y25 are dy

namic variables, Y26 to Y314 are parameters) . ydot27 to ydot2s3 are

the additional ODES for the sensitivity functions . The bottom of the

figure shows examples of MAPLE commands that are used to write the

system of ODEs to files .

At the end of simulations and parameter fittings, the MMT software outputs statis-

tics about parameter changes during parameter fitting . Additionally, the solutions

of the IVP can be viewed . The MMT software produces graphs showing compound

concentrations versus time or kinetic reaction rates versus time . In the illustrative

example five graphs are produced, one for each compound F6P, FBP, ADP, ATP,

and PEP. The graphs will show the concentrations of these compounds versus time.

Additionally, five graphs are produced that show the values of vi to v5 versus time.

Thus, it is possible to examine how the fluxes between compounds change over time .

The produced graphs are in GIF or POSTSCRIPT format and can be used in other

programs e.g . to write reports or summarize results .
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ydot[27] := + diff(f[1],y[1 ]) * y[28] ;
ydot[29] := + diff(f[1],y[1 ]) * y[29] + diff(f[1],p[189]) ;

ydot[283] :_ + diff(f[7],y[1 ]) * y[53] + diff(f[7],p[860 ])
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Performane issues

The process of setting up the system of ODEs from the information stored in the

database and the compiling, linking, and creation of the initial files can take several

minutes for a system of 30 ODEs and 300 parameters . The most time consuming

process is the setup of the ODE system, i.e . the extraction of the information from

the tables of the relational database . However, if the number of SQL statements

is reduced by using additional temporary tables, this process can be accelerated
significantly.

For simulations or parameter fittings with the saine model, the time consuming
setup and creation of the C-code is not required anymore. Thus, the initializing time
is reduced to a few seconds. Only after choosing a different model for simulation
or parameter fitting, the C-code has to be recreated . Future versions of MMT will
keep the C-codes belonging to a specific parameter identification in place in order to
directly compile and link it to the simulation and parameter identification code . This
requires to take care that the C-code on disk is really in correct form, i.e . if changes
in the model have been made, the C-code has to be updated . If the model did
not change, however, usage o£ an already available C-code can save several minutes
computation time .

If the C-codes are kept in place, it will also be easier to reproduce simulation
results with parameters saved in the database . As argued before, this would reduce
the amount of data stored in the database by approximately 90%.

4.6 .

	

The MMT simulator and optimizer

The complete simulation and parameter fitting procedure of the MMT software is
performed by a C procedure . Figure 4.14 shows the structure o£ the MMT simulator
and optimizer . The main procedure is started from the MMT user interface . Once
started, the software runs independently of the user interface . Input is provided from
text files and the output is written to such . The MMT user interface, however, keeps
track of the progress of the simulation or parameter fitting . After recognizing the
termination of a simulation or parameter fitting, it reads the data files that contain
the results and fits them into the MMT database. The simulator and optimizer
maintains the following features :
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largeGrads.c
error functional

	

sort parameters
or :

	

according to
Efun.c

	

sensitivities
solution of IVP including
gradients evaluation of

error functional

Figure 4.14. : The structure of the MMT simulator and optimizer . Arrows indicate

data flow from the external files or function calls and returns . If

explicit gradients are needed, Ef ex . c replaces Bf ex . c and Efun . c

replaces fun . c .
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1 . Initialization of parameters: All initial parameters and the data file are read

and stored in vectors and arrays . Additionally, a significant number of param-

eters for the simulation and parameter fitting algorithm have to be read from

external files . Global variables are setup that can be used in the subroutines .

2 . Do an initial simulation: An initial simulation is done by calling the subrou-

tine fun. fun calls the Fortran algorithm LSODA for solution of the IVP. Also,

fun calculates the error functional if a data file was specified for the simulation .

If only a simulation should be performed, the results are written to the result

file and the program terminates.

3 . Initialize the parameter fitting: Depending on the user choice, sensitivities

of all parameters are calculated and parameters with low sensitivities are ex-
cluded from parameter fitting . The sensitivity calculation is performed by the
C procedure sensitivity. sensitivity calls fun for solution of the IVP,
calculation of fluxes and the error functional . Sorting parameters according to
their sensitivities is done by the procedure largeGrads .

4. Start parameter fitting loop: After all preliminary tasks are finished, the
Fortran parameter fitting routine SUBPLEX or LBFGS is called . These rou-
tines call fun for calculation of the error functional which they are trying to
minimize . Figure 4.15 shows a flowchart for the function fun.

5. Result and termination check: After the parameter fitting algorithm ends
it returns the identified parameter values and the minimal value of the error
functional to the main procedure . The main procedure compares results with
previously identified results and writes them to a temporary result file . The
result file is analyzed be the MMT software and the information stored in the
database. The main procedure also checks if the maximum number of restarts
is reached. If yes, final results are written and the program exits .

6 . Restart of the parameter fitting algorithm : The parameter fitting algorithm
will be called repeatedly until the maximal number of parameter fittings is
reached . According to strategy parameters set by the user, parameters will be
changed randomly with a normal distribution over a given range before the
parameter fitting restarts . This procedure is important to identify the global
minimum of the error function . The algorithm now continues at point 4 with
the start of the parameter fitting algorithm .
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initialization of parameters
t=to

LSODA mode parameter to
"initial solution"

search for next time t data
with datapoint,
calculation of

next time tr for regular output

tout - tr

call LSODA for solution of
IVP from t to tollt

tout

	

tdata ?
no

evaluation of
Error Functional

t --- tout
set LSODA mode parameter

to "continuation"

Termination -O--yes tend reached ?

Figure 4.15 . : Flowchart for the C-routine fun for solving the IVP. Regular output

times are set by the user . Solution is always performed to the next

output time (which is either regular output time or a time with a

data point) . After the first call, LSODA is called with the continuation

mode in order to increase performance and accuracy.
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Notice that the routines Efex and Efun that are shown also in Figure 4.14 are
replacements for fex and fun . These replacements are needed if explicit gradi-
ent information is needed for an parameter fitting (parameter fitting) algorithm as
LBFGS, or if explicit sensitivities should be calculated by solving the extended system
of ODEs (variation of parameters) . As mentioned above, Efex and Efun are also
created from the MMT user interface and coexist with fex and fun.

4.7.

	

Summary of the MMT software

In this chapter, the MMT software has been described . The tool can be used for
model setup and model analysis with simulations and parameter fittings . The setup
was created in the framework of a relational database system that combines effi-
ciency and consistency. Thus, models can be build and analyzed quickly, results are
reproducible, and prior results can be reused in new models .



5. Numerical Algorithms

This Chapter describes the numerical treatment of the system of ODEs that results
from the modeling approach and model generation presented in Chapters 3 and 4 .
Then, methods for applying the developed models to

" simulate chemical reaction networks

" reproduce experimental data sets from experiments described in Section 2.1 .

are shown. An easy example consisting of a single ODE shows the probem that

occur during fitting of a model to experimental data .

Simulations require the solution of an IVP by numerical algorithms. Reproduction

of experimental data sets is based on finding parameters for the model such that the

model fits to the data. This is a parameter fitting or optimization task that has to

be solved . Figure 5.1 summarizes the general approach of the numerical treatment .

Recall of the Model Properties

The structure ofthe models developed in this work have been described in Chapter 3 .

Recall Equation 3.5, which formulates the general non-stationary problem:

C=N-v(C,S,P)

As mentioned, the reaction rates vi are usually of non-linear form . The System of

ODES is high dimensional and, because the chemical reactions operate on different

time scales, the System is often stiff.
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parameterization :
new parameters
to fit model

to data

setup IVP:
initial conditions,
parameter values

simulation:
solve model
system (IVP)

check results:
evaluate the

error functional

sufficient
model fit?

yes

model that
describes

experiments

Figure 5.1 .: General procédure for setup of thé mathematical model, simulation,
and fitting to data using numerical algorithms . Evaluation of thé error
functional means calculating a différence between model output and
experimental data set as described in thé text .



Additional Notations for this Chapter

In Equation 3.5 the vector v of reaction rates is written in general form and depends
on extracellular and extracellular compound concentrations, enzyme concentrations,
and parameters .

Flom now on S and E will be omitted because the enzyme concentrations are not
used in the presented models and the only extracellular concentration that is part
of the models is the GLC concentration . This concentration is fixed and changes
its value only at the time when the substrate pulse is given . Therefore the GLC
concentration will be considered as an entry in the parameter vector P. At time
t = 0, the initial conditions are written as Co := C(0) and C = C(CO,P,t) is the
vector of compound concentrations at time t that depends on the initial conditions
and the parameters.

The total differential operator of a function f with respect to x will be denoted in

the following by

DXf
:	ôxf

to distinguish it from the partial derivative ax .

To illustrate the differente between the total dif%rential operator and the partial

derivative, the following functions are defined :

If the partial derivative of f(x, y, t) with respect to x is calculated, all other variables

are considered as constants . The same principle is used for the partial derivatives

wich respect to y and t. Therefore

_Of
= 2y

8y
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f(x, y, t) = x2 + y2 + t

x(t) = sint

(5.2)
y(t) = cos t
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af = 1

	

(5.3)
ät

In contrast to thé partial derivatives calculation of thé total differential operator D.
does not assume y and t to be constants . Instead, y and t are treated as functions
of t. Calculation of Dx and Dy yields thé same result as thé partial derivatives .
Calculation of Dt yields a different result because x and y are functions of t:

D~

Dy

Dt

Yi = M - C(ti)

af _ax _af _ay _af at
ax ax + ay ax + at ax

_Of _ax _Of _ay _af _at
ax~ay + ay 8y + et ~ay

2x+0+0=2x

0+2y+0=2y

af

	

ax + af . ây + af , at = 2cost - 2sint + 1

	

(5.4)ax Ft ay at et ai

The experimental data at time ti will be modeled by a measure equation

Yi = M - C(ti) + Ei

(5.5)

where C(ti) are thé compound concentrations at time ti calculated by solving thé
IVP and M is a measure matrix . This matrix is necessary because usually experi-
mental data is not available for all compound concentrations Ci . M is setup such
that yi contains only non-zéro values for compound concentrations wich available
experimental data . Also, if thé vector of measurements at time ti is denoted by yi
let

(5.6)

where si is thé vector of différences between thé calculated values C(ti) and thé
measurements -~i . The matrix of all measurements is denoted byY.



then

3 .26 /

Yi = M - C(ti) + ei =

ppyJ
1.11 F6P(ti)

0.24 FBP(tl)

3.26 PEP(tl)

5 .1 .

	

Solutions of the system of ODES

5.1 . SOLUTIONS OF THE SYSTEM OF ODES

In the illustrative example, suppose measurements exist only for F6P, FBP and
PEP. Then

1

	

F6P(ti) F6P(ti)
1 FBP(ti) FBP(ti)

Yi = M . C(ti)

	

.

	

.

	

.

	

.	ADP(ti)

	

_
ATP(ti)

1 PEP(ti) PEP(ti)

Suppose that for a time ti the following measurement vector yl exists :

or for the difference between calculated values and experimental data

The non-linear structure of the system of ODEs of the model makes an analyti-

cal solution impossible . Therefore, numerical solution techniques have to be used .
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AU important points for successful and accurate numerical solution of the model

equations are discussed in this section .

5 .1 .1 .

	

Properties of rate equations

All rate equations that have been used are smooth functions . If parameters are

set in a "reasonable" range, no singularities are possible . "Reasonable" means that

the parameter is in a range that can be justified from biological interpretation . It

is still possible that during a simulation a floating point overflow occurs because

a substrate concentration increases unlimited . This indicates, however, that the

parameter values are setup incorrectly or the model is incorrect . Such unrealistic

results can therefore be excluded . In addition to these properties, all functions used

to describe reaction rates are at least twice differentiable .

For the simulation of the metabolic network, initial compound concentrations are

given . Therefore an IVP has to be solved . Existence and uniqueness of a solu-

tion with the above mentioned properties of the rate equations follows from the

Fundamental Existence- Uniqueness Theorem, see e.g . page 74 in [70] .

5.1 .2 .

	

Accuracy of solutions

An important point when solving IVPs with a numerical algorithm on a computer
is the accuracy of the solution. Chemical .reaction systems often consist of several
reactions that operate on a different time scale . If the time scales differ by many
orders of magnitude, the system is called a system of stiff ODEs. There are, however,
no exact mathematical methods on how to classify a system of ODEs as stiff. The
historically first opinion is from Curtiss & Hirschfelder in [24] : "Stiff equations
are equations where certain implicit methods perform better, usually tremendously
better, than explicit ones." Quantities that give information about the stiffness of
a system are the Jacobian 8f/8y, the dimension of the system, smoothness of the
solution, or the integration interval . A detailed discussion about these measures and
solutions of stiff systems can be found in [39] . In [39] a comparison of numerical
solution algorithms for stiff problems on a set of test problems can also be found .

For solution of the system of ODEs, a publicly available solver LSODA [71] was cho-
sen. This algorithm is a variant o£ the MODE solver (Livermore Solver in [45]) that
includes also a solution algorithm for non-stiff problems and switches automatically
between the two methods . This results in a better performance of the algorithm.
The LSODE algorithm has been examined and tested during the above mentioned
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comparison of solution algorithms in [39] . Results of these tests indicate that MODE
is a very fast code that works accurately for tolerantes of < 10-6 but might have
some accuracy problems for tolerantes > 10-6 .

During usage of LSODA, some accuracy problems occurred . If the error tolerante
was chosen larger than 10-6 , sometimes negative values resulted for some of the
compound concentrations . However, reaction rates vi are always chosen to decrease
to zero if the compound concentration approaches zero, i .e .

lim vi(C) = 0
C->o

for a compound concentration C. Thus negative values result from numerical in-
accuracy. These numerical problems Gould usually be avoided by decreasing the
tolerante to values between 10-8 and 10-12 .

5.1 .3 .

	

Intermediate time steps

A simulation with an ODE solver from an initial to to a final time t1 is straightfor-

ward if only initial conditions are used as input variables . However, sometimes the

solution has to be divided into several time periods . This happens if

. intermediate results are needed, e.g . for comparison wich experimental data.

parameters are not constant with time . This problem occurs for the extracel-

lular GLC concentration that changes in the experiments when the pulse is

given . The saine problem occurs if a compound is not included in the IVP as

a dynamic variable but experimental data is used . Then for all times where an

experimental data value is available, this value has to be passed to the ODE

solver .

With the algorithm LSODA intermediate time steps do not cause any Problems . The

algorithm tan be stopped at intermediate times. The algorithm then returns to the

routine it was called from . Intermediate results tan now be stored and Parameters

tan be changed for the next time period . For reasons of performance and accuracy it

is important to call LSODA with the 'continuation' option . In this case, the algorithm

continues from the time it was stopped using results from the Prior time step . This

is possible because the results are stored in global variables and are therefore still

stored in the memory of the machine .
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5 .1 .4 .

	

Smoothing of the experimental data

During simulations experimental data values can be used for some compound con-
centrations . This is a useful approach especially for compounds with inaccurate
measurements or with unknown reaction kinetics . E.g. a correct description of the
ATP concentration would have to include several hundred reactions where ATP is
produced or consumed . Instead, experimental data values can be used as an input
to the simulation algorithm .

In order to remove noise in the experimental data, a polynomial approximation was
performed to obtain a function that represents the data in a smooth way. Because
polynomial approximations are likely to show oscillations it was important to manu-
ally inspect the approximations . See Figure 5 .2 for an example of the approximation
of the G6P and PEP concentrations .

The probem resulting from polynomial approximations ofhigher order, mainly the
oscillating behavior are well known [108] . A much better approach is the approxima-
tion with splines, see also [108] . Notice, however, that even a spline approximation
is difficult for the given data due to the dynamics present in some measurements . A
closer look at Figure 5 .2 shows the problem : in the left figure for G6P concentration
at time 10s a few measurements are far from the other measurements . A smoothing
spline with high accuray would include these data points which would result in an
oscillatory fashioned graph. A louver accuracy of the spline would not pass through
these points and provide an approximation similar to the one shown. Because esti-
mation of the measurement error is difficult, it is not clear which would be the more
realistic approximation .

The obtained approximation was also used during parameter fitting. Because a
larger number of values results for fitting the parameters, it can be hoped that an
optimal solution can be identified faster .

5.1 .5 .

	

Summary of the solution algorithm for the IVP

With the LSODA algorithm, the given IVP is solvable fast and wich good accuracy,
taking into account the stiffness and non-linearity of the problem . Computation
time for a system of 26 ODES wich 302 parameters, a time period of 40 seconds
with an accuracy of 10-1° is approximately 1 second. If sensitivities are calculated,
computation time increases significantly. The number o£ sensitivity functions to
be evaluted depends on the number of parameters and initial conditions and on
the function dependencies (a sensitivity has to be calculated for each parameter
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Figure 5.2. : Example of approximation function for thé experimental data of G6P

and PEP.

and initial condition for each function they are part of) . In thé above mentioned

example, solution of thé system of ODEs inclduing all sensitivity functions requires

approximately 10 seconds time. The computation time increases with thé number

of intermediate times ti for which results are to be stored . The user defines these

intermediate time steps .

Results obtained with LSODA were compared with solutions from thé commercial

software tool MAPLE in order to check correct setup of thé code. No différences

could be seen . The solver was used for all solutions of thé IVP. Choosing thé

tolerance parameter correctly is important . A large tolérance significantly increases

efficiency of thé algorithm, however, final results should be checked with a small

tolerance to identify possible inaccuracy problems.

In thé last section it has been argued why thé solution of thé given IVP with specific

numerical algorithms is not a problem . This assumes, however, that thé following

requirements are fulfilled :

1 . Requirement : A set of initial conditions (Le . compound concentrations at an
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initial time to) is given

2 . Requirement : A set of initial parameters (i.e . all parameters occurring in the

enzymatic reaction kinetics, e.g . Michaelis-Menten 3.3) is given

Under these conditions, a solution can be calculated for a user defined time frame

from initial to to final time tl (see also Figure 5 .1) . This also allows simulation of

system behavior under different conditions.

Requirement 1 is partly fulfilled with the available experimental data. Other com-
pounds that have not been measured and are still included as dynamical variables
in the model need to be estimated . Requirement 2 usually can only rely on data
from in vitro measurements. These values vary in a broad range . Therefore, three
sources of uncertainty result for a given model structure .

1 . Initial concentrations of compounds that have not been analyzed .

2 . Parameter values for the enzyme kinetic equations.

3 . Initial concentrations of compounds that have been measured but with low
accuracy.

The relevance of these uncertainties depends on the sensitivities of an initial con-
dition or a parameter value on the simulation result . It is therefore important to
calculate sensitivities for each compound and parameter . Then the relevance of
errors in initial compound and parameter guesses and relevance of errors in mea-
surements can be estimated .

5 .2 .1.

	

Interpretation and usage of sensitivities

Sensitivity calculations can be useful and important in many respects. Parameters
with low sensitivities are difficult or impossible to be identified . Sensitivity infor-
mation can be used for

Model Setup: Parameters with low sensitivities can be eliminated from the sys-
tem.

Experimental Design: Experiments can be planned that allow better identifia-
bility of parameters .



Data Fitting, Parameter Fitting : During fitting of a model to experimental
data, sensitivity information can be used to reduce computation time. This
aspect will be discussed in more detail in the following section .

Analysis: Sensitivities are needed for statistical analysis of the fitting results . E.g.
calculation of the covariance matrix or control coefficients depends on the

sensitivities [67] .

Conclusions: Interpretation of results has to be carried out keeping in mind sensi-

tivities of each parameter . Parameters with low sensitivities allow less strong

conclusions.

5 .2.2.

	

Sensitivities of reaction rates at a fixed time t

Sensitivities can be calculated from the reaction rates vi itself. Only changes of these

rates are analyzed in dependence on parameters and initial conditions . Evaluation

is carried out at a fixed time t .

For calculation,

_a

	

v(C, P)
aP

a
v(Co, P)

aco

have to be calculated and evaluated at time to .

5.2. SENSITIVITY ANALYSIS

(5.7)

Sensitivities calculated in this way show the influence of parameters and initial

conditions on the kinetic reaction rates . The calculation is fast, but the sensitivities

can change and be completely different at other times.

5.2 .3 .

	

Variation of parameters

Because the used enzyme kinetic rate equations are differentiable with respect to

the initial conditions and parameters, it is possible to setup a variation equation

system of ODES that explicitly contains sensitivities of the system (Variation of

parameters) . This approach is based on the calculation of the sensitivities

Dp

	

C(Co, P, t)
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Dco

	

C(Co,P, t) (5.8)

Calculation of these sensitivities cannot be done directly because the function vector
C(Co.P, t) is not explicitly known. However, it is possible to calculate the sensitiv-
ities indirectly. If the functions in Equation 5.8 are derived with respect to t and
Schwarz' s Rule for exchanging derivatives is used, then

The total differential operator D is calculated as illustrated in the example above
(Equation 5.2) . Therefore a system of ODEs for the sensitivities results (1 is the
unitary matrix)

_d

	

__

	

_av _ac

	

_av

	

oc
dtDpc

	

N

	

(ac- aP + aP)

	

' âp (0) = 0

	

(5.11)

dt
DcoC

	

- N ' ac aC0
'
aco

(0) = 1 (5.12)

It is now possible to integrate the two additional systems of ODEs that result
together with the original system of ODEs to obtain the sensitivity functions . This
integration is done numerically due to the complexity of the derived system of ODEs.

To illustrate the concept, consider a single ODE

x=a - x

	

x(0) =xo

where a is a constant . The explicit solution of this equations is

x(t) = xo , eat

d
DPC(Co,P,t) = Dp

L
dt C(C,P,t)I

av av= Dp [N - v(C, P)] = N .
(
_ac + (5.9)

aC aP aP

dtDcO
C(Co,P,t) Dco [ddtC (C,P,t)J

av OC
= Dco [N - v(C, P)] = N .

ac "à-CO (5 .l0)
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The sensitivity functions are obtained from

Define ya := âa and yx := âa as the sensitivity functions with respect to a respec-
tively x . Then

ya(x, t)

	

=

	

Pat -I- axt -f- xt - 1

Pat

Notice that y, depends only on a and t. This means that the sensitivity of the

solution is independent of xo . The sensitivity function ya, on the other hand, depends

on a and xo . This sensitivity increases with increasing xo and a .

Calculation of the sensitivity functions is time consuming . If the number of param-

eters is Np, an IVP of dimension N2 +N Np has to be solved . It provides, however,

a clearer picture of parameter sensitivities . Sensitivity functions show the influence

of a certain parameter at a certain time : Thé solution of the IVP gives P(t) and

âeo (t) . This is of great interest in a dynamic situation, because it can be seen if the

sensitivity functions of several parameters or initial conditions are linearly depen-

dent . This allows conclusions about the identifiability of certain parameters with

the available data set . In [48] a discussion of sensitivity functions for parameters of

a Michaelis-Menten based microbial growth model is presented .

5.3 .

	

Error functionals

An error functionalF for thé différence between experimental data and model output

is needed for fitting thé model to thé experimental data (Parameter fitting) . Different

error functionals were used for parameter fitting that are summarized in this section .

Recall that ej = yi - Yi was defined as thé différence between thé vector of cal-

culated values yi and Yi at time ti . Then sij is thé différence between calculated

value and measurement of compound j at time ti and yi,9, Yij is thé calculated

respectively measured value of compound j at time ti . This notation will be used in

this section . Additionally, it is assumed that for each compound a weighting factor
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_d _ex âx ex âx ex_ _
.

_ =
a xdt ôa (â

_
xàa + 8a)

_
8a +

8x =' âa
(0)0 (5 .13)

d ex 8x ex ex___ __ __
-

a '
8x (0) = 1 (5 .14)
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wj is defined . It is usually small if a compound can be measured with low accuracy
only.

Many error functionals depend on weighted, relative squared différences between
simulated values and experimental data values . Several modifications have been used
in order to improve performance of thé parameter fitting algorithm or to influence
thé fit of specific compounds. In thé following, measures dk(ei) are defined that
calculate différences between calculations and measurements for a specific time ti .
In order to get thé overall goodness of fit, thé sum of dk (ei) for all times ti has to
be calculated:

Nt
F(Co, P) = E dk (Ei)

i=1
(5.15)

where Nt is thé number of time steps ti . Notice that thé error functional depends on
thé vectors of initial conditions Co and parameters P. Five différent formulations
of dk(Ei) are implemented in thé MMT software :

1 . Standard powered différences : The standard measure for différences between
calculated values and measurements is a weighted sum. of least squares

N

	

Ieij lpd1(ei) :=E wj _

	

(5.16)
j-1

	

yz,j

where p = 2 and wj is a weighting factor for compound yj at time ti . In some
parameter fitting runs, P was set to 4 in order to increase thé influence of small
values . This sometimes yielded slightly better performance of thé parameter
fitting algorithm . However, statistical theory is limited to p = 2.

2 . Relative Powered différences : Instead of taking thé déviation of thé simu-
lated value relative to thé data value, various différent methods have been
used :

" a deviation relative to thé absolute value of thé model output,
N

leijlpd2(ei) := ~wj
j=1 Yi,j

(5.17)

This increases thé influence of times tj where small calculated values
coincide with large data values .

" a deviation relative to thé minimum of thé absolute value of thé model
output and thé data value, i.e .

N
d3 (ei) :=E wj

	

Ieij 1p
j=1

	

min(yi,j, fi,j)
(5.18)

104



5.3. ERROR FUNCTIONALS

This calculation increases thé influence of times tj where a small model

output or a small experimental value are present .

" a deviation relative to thé sum of thé absolute value of thé model output

and thé data value, i.e .

N
I Ei,J l pd4 (Ei) _ Zwj

j=1

	

Yi,j -I- Yi,j
(5 .19)

Calculation of the error functional is according to equation 5.15 with the ap-

propriate definition of d.

3 . Relative différence in Derivatives: The rate of change of a metabolite con-

centration Ci at a time t is given by Ci (C, P, t) . From thé experimental data,

a rate of change has been estimated using three measurement points at times

ti-1, ti, and ti+l :

"

	

1 [Yij - 3'i-1,j + Yi+l,j - Yi,j l

	

(5.20)
Yi- "i = 2

	

ti - ti-1

	

ti+1 - ti

The value of this approximated derivative was compared with thé value of

~i,j = Cj (Co, P, ti) . If no data point for ti-1 or ti+l was available, thé next

points (ti-2 or ti+2) have been checked until a data point was found . Now

N

	

léi,j IP
d5
(M := j~,wiL + jYi,jj=1

(5 .21)

where èi is now thé différence between thé calculated Yi,j and thé estimated yi j

derivative. L is a number introduced to avoid division by zéro in thé fractional

part of thé équation . Calculation of this differential operator is problematic for

noisy data, even if three points are included in thé calculation . The functional

d4 should therefore only be used for smoothed data.

In all cases deviations have been calculated only for times where a data point was

available . In thé numerical algorithm this was accomplished by setting thé weights

wj = 0 for times and metabolites where no data point was available . Notice that dl

to d4 can attain large values if one or more Yi,j attain small values . This problem

can be overcome by adjusting thé weights w according to thé measurement size .

During parameter fitting usually errors of thé second type (d2) have been used,

where thé powered distance is taken relative to thé minimum of calculated value

and data value . The third kind of error functional including derivatives (d5) was

only used in combination with one of thé other functionals. This was accomplished

by adding thé error functionals (e.g . Adk + (1 -ß)d3 for 1 < k < 4 and 0 :5 A :5 1) .
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5.3 .1 .

	

Sensitivities with an error functional

Sensitivity estimations can be done by checking the influence of a parameter on the
value of the error functional. Instead of calculating only the change of one rate
equation at the initial time to as described, an overall influence can be calculated .
For calculation DpF and DcpF have to be calculated . Such sensitivities depend on
the error functional, the data set used, and the time frame considered. Therefore
they give more insight in the global influence of parameters than sensitivities that
are calculated at time t = 0 only. Particularly the influence on the quality of the
model to fit experimental data is visible .

5 .4.

	

Parameter fitting

The parameter fitting problem of this work is described as follows :

Find a parameter vector P such that the distance between the vector function of the
compound concentrations C(CO, P, t) and the matrix of experimental data points Y
is minimal, i. e. find minco ,p(F(CO,P)) where Co and P are variable in a certain
range.

Because some compound concentrations may be unknown or measured with a large
error only, some initial conditions of the vector Co are subject to parameter fitting
also . These will be referred to as optimized initial conditions in the following .

Minimizing F for the developed pathway models is difficult because

" lots of parameters (and initial conditions) are subject to parameter fitting

" the number of available data with high accuracy is small

" no good initial values for starting a parameter fitting are available

" due to the non-linearity of the model, local minima are likely to occur that
may be identified during parameter fitting

To solve the given problem, two different parameter fitting algorithms have been
used that are described in the next section . Additionally, the following strategies
have been applied in order to find a good minimum for F:
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1 . Define starting points: Starting points for parameters and optimized initial
conditions have been estimated from literature values . Mostly, however, these
values resulted from in vitro measurements . Additionally, values of these pa
rameters vary in a avide range in literature . Still, literature values are impor-
tant at least to define a reasonable value for parameters and initial conditions .

2 . Introduce constraints: For each parameter and optimized initial condition, a
range has been defined . It is required that Pi,min <_ Pi < Pi,max for 1 < i <
Np where Pi,min is the minimal value Pi can obtain and Pi,rnax is the maximum
value, Np is the number of parameters . Also, Ci,min <_ Co,i <_ Ci,max for all

initial conditions subject to parameter fitting . At the beginning, these ranges

keep the parameter value close to the literature values, but during longer

parameter fittings, constraints can be relaxed to increase the probability of

finding the global minimum .

The following sections describe the parameter fitting algorithms used.

5 .4.1 .

	

Parameter fitting algorithms

The given problem is non-linear and allows no explicit solution of the parameter

fitting problem . A variety of numerical methods exists that can be successful for

solving such a problem. However, one cannot be sure a priori which method will

work best, or which method will fail. Also, computational costs cannot be estimated

a priori . Two methods were used for solution of the problem :

1 . Gradient method: As described in this chapter, explicit derivatives with re-

spect to initial conditions and parameters can be calculated from the System .

Therefore, a gradient method can be used for parameter identification [69] .

However, gradient methods do not show good performance in finding global

minimum if applied to dynamical systems . In particular, the gradient method

requires good starting points and is likely to end up in local minima . Results

gained from the gradient method were not very promising.

2 . Nelder-Mead simplex method: A popular method for parameter fitting is

the Nelder-Mead simplex algorithm [66] . It is based on a simplex in the n-

dimensional parameter space. A simplex is then an n + 1-dimensional convex

hull in this space . The Nelder-Mead algorithm moves this simplex around in

the parameter space, changing its size and shape, always calculating and com-

paring the error functional F. When a minimum is approached, the simplex is

reduced in size, and smaller movements of the simplex in the parameter space

are performed .
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3. Tom-Rowans SUBPLEX method: Tom Rowan developed the SUBPLEX parame-

ter fitting method [84] which is based on the Nelder-Mead algorithm . The algo-

rithm uses the Nelder-Mead simplex algorithm on subspaces of the complete

parameter spare. The purpose of dividing the parameter spare into several

subspaces is an increased performance of the Nelder-Mead simplex method for

lower dimensional subspaces . Subspaces are chosen in the direction of improve-

ment, i.e . initially the algorithm evaluates which parameters are sensitive and

are promising for improvement of the fit . After a tolerante is reached during

parameter fitting in a subspace another subspace is chosen .

Other parameter fitting algorithms that might be helpful for the problem but have

not been tested so far are

1 . Stepwise problem solving : Dennis et al [26] developed a method for partial

parameter identification of a dynamical parameter fitting problem of non-linear
ODEs . During the process, the problem is solved for the time period to to tl,

then from tl to t2 and so on. The algorithm is available publicly but has not
been tested yet . Therefore it cannot be estimated if the higher computational
cost is overcome by a much better performance .

2 . Levenberg-Marquardt algorithm : This popular method [64] is a well per-
forming algorithm that is also able to obtain an optimial solution if the initial
parameters are far from this solution. However, the approach is usually not
successfull for identification of a global minimum .

3 . Multiple shooting : A multiple shooting approach ran be used for parameter
identification of a system of ODEs. The code PAREST [41] has been evaluated
and some first experimental parameter identifications have been performed .
The code seems to require good initial parameter estimates for successfull
parameter identification which are usually not available .

4 . Genetic algorithms: For genetic algorithms, parameter vectors are considered
as a vector of genes . Genetic vectors can mutate and an evolutionary approach
is used to improve the genetic vectors . It cannot be proven that the algorithm.
per£orms well on a problem . Collections of genetic algorithms can be found
in [54] and [34] .

5.4.2.

	

Improving the performance of parameter fitting algorithms

In order to get a good performance of parameter fitting algorithms when searching
for global minima, the following methods are frequently used:
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1 . Restart thé parameter fitting: Because thé algorithms depend on thé initial
parameter vector Po, and often end up in a local minimum that depends on
this initial values, it is useful to restart thé parameter fitting after a local
minimum is reached . For thé restant, thé initial parameter vector Po can be
changed randomly, where each parameter should remain in its range in order
to fulfill thé constraints .

2 . Range extensions: After an parameter fitting is finished, analysis of parame-
ter values can give indications which parameters are close to their upper or
lower bounds . A range extension can be performed for parameters that are
uncertain from a biological point of view . For other parameters, however, a
range extension should not be performed because it would detach thé model
from its biological meaning.

3. Parameter space reduction : The dimension of the parameter fitting prob-
lem, i.e . the parameter space can be reduced . As described before, this can
be done by using simpler enzyme kinetic equations, or reducing by fixing cer
tain steady state compound concentrations . At the parameter fitting stage,

however, additional parameter space reduction can be performed . This can

be done by fixing certain parameters where good guesses are available . Also,

parameters with low sensitivities can be neglected . The methods implemented

in the MMT software are described in Chapter 4 .

Another approach proved to be helpful : If panameters were scaled according to

their sensitivities, data fitting efficiency sometimes increased significantly. If Pi,o

is the initial value of parameter Pi and the parameter fitting algorithm sets the

parameter to a value Pi, then the scaling method sets the parameter to

Smax
Pi = (Pi _ Pi,o) .

	

SP,i

(5.22)

where SP,i is thé sensitivity of parameter Pi and S,nax is thé maximum sensitivity of

all values Sp,i . Le ., thé lower a sensitivity is, thé more thé increase in thé différence

of thé parameter to its initial value Pi,o . The idea is that parameters with low

sensitivities have to be changed more in order to influence thé result .

A pnoblem of the scaled parameters is the fact that some parameters change by

a large amount. This can only be justified for parameters where no good litera-

ture values exist . Otherwise, the approach of scaled parameters leads to unrealistic

results .
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5 .4.3 .

	

Reasons for failing parameter fittings

Parameter fitting of a model can fail for several reasons . Numerical methods as the
SUBPLEX algorithm cover the whole parameter range while fitting a model. Therefore
if a model cannot be fitted to experimental data after sufficient computation time
(which depends on the dimensions of the problem), then the numerical algorithm is
most likely not the reason. Instead, one of the three following situations may have
occurred :

Underfitting:

	

If a model cannot be parameterized because too many of the pa-
rameters have been fixed or the ranges of the parameters have been chosen
too small, then the model is underfitted . This problem can be overcome by
allowing more parameters to change or allowing parameters to change over
larger ranges .

Overfitting :

	

A model is overfitted if parameter fitting does not lead a unique re-
sult . Le . if several parameter vectors P exist that fit the model to experimental
data with the saine quality. Overfitting can be a result of too mang parame
ters for the available data or too large ranges of the parameters . The Problem
can be overcome by fixing parameters with low sensitivities that cannot be
identified by the available experimental data. These parameters then have to
be estimated using other sources (Literature, other experimental data) .

Incorrect Model: If a model cannot be parameterized even though all its param-
eters are allowed to change without limits, then the model itself is incorrect .
The model does not describe the system correctly and changes in the model
setup have to be made.

5.4.4 .

	

Performance and problems of parameter identifiability : an
example

An example is given that shows performance and the Problems of parameter iden-
tification. The following steps have been carried out and will be discussed in this
section :

1 . A simple model with one ODE is setup .

2 . A simulation is performed with the model . The simulated compound concen-
trations are modified randomly to add a hypothetical measurement error .
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3 . A Monte-Carlo simulation is performed in order to parameterize the model to
the randomly modified simulation output .

Figure 5.3 shows the fluxes into and out of the intracellular G6P metabolite pool
of Escherichia coli . The following system was setup :

vl describes the uptake of GLC with the PT-System, v2 the flux from G6P into the
PP-Pathway, and v3 the flux from G6P into glycolysis . One ODE was setup for the
concentration of G6P:

G6P = vl - v2 - v3

	

(5.24)

G6P ' -	6PGL

NADP' 'NADPH

Figure 5 .3 . : Fluxes in and out of the intracellular metabolite pool G6P as included

in the example model.

Vl = VGLC G6P

V2 = VG6P --> 6PGL

(5.23)V3 = VG6P --> F6P
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For vi and v3 a Michaelis-Menten kinetic is assumed, whereas the flux v2 is assumed

to depend proportionally on the G6P concentration . The system then is

GLC
G6P = vynax,l ' .K�,,,i -f- GLC - vrrzax,2 ' G6P - v~rcax,3 '

	

G6P
Krn,3 + G6P

	

(5.25)

This model with three kinetic equations contains five parameters and is of non-

linear form . For the specific case of GLC uptake with the PT-System (flux vl), a

molecule of PEP is needed for each GLC molecule taken up. Therefore, a correct

representation of the rate equation rGLC->GSP needs to include the PEP concentra-

tion and then is a function of at least two concentrations . This would add more
complexity and parameters to the model. For illustration of the Parameter fitting

problem only the simple model in Equation 5.25 is discussed here .

A simulation with the example model

Concentration of GLC was used as input to the system . It was set in the saure way
as in the GLC pulse experiments described in Section 2.1 :

GLC(t) _

	

0.1

	

mM

	

, for

	

O.Os < t

	

< 5 .Os
- {

	

15.0

	

mM

	

, for

	

5-Os < t

	

< 40.Os
(5.26)

In order to solve the IVP generated by the model, parameter values of the model
have been set to the values shown in Table 5 .1 . The table also shows the ranges of the
parameters that are used in the parameter fitting described in the next paragraph .
The parameter values have been chosen to obtain an uptake flux v,, = 0.35 mM/s
of GLC before and v1 = 1 .45 mM/s after the pulse of GLC. These fluxes are in a
reasonable range from a biological point of view .

Measurements o£ intracellular compound concentrations are not exact . In order
to account for this efFect, the obtained simulated compound concentration of G6P
from solution of the IVP is provided with an error . It is assumed that a normally
distributed error of 5% per average is present in the measurements . This "noise"
was added to the simulated values for G6P. Figure 5.4 shows the exact model output
and the G6P concentrations with added noise .
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Table 5 .1 . : Parameter values used for the simulation with the experimental model .
Shown are also the ranges of the parameters that are used during pa-
rameter fitting.

Monte-Carlo simulation to test parameter fitting

In order to test if the parameters used for generation of the data Shown in Fig-
ure 5 .4 can be identified during parameter fitting, a Monte-Carlo simulation has
been performed . The following setup was used:

vmax,11 Km,1 1 vmax,3, and Km,3 have been used for parameter fitting . vmax,2 has
been assumed constant . Initial values for the parameters subject to parameter

fitting have been chosen randomly by assuming a uniform distribution of the

initial parameters over their ranges . The ranges used are Shown in Table 5.1 .

" As a fitting criteria, the relative least square dl was used .

. 25000 parameter fitting runs have been performed wich the randomly changed

initial parameter values . Each parameter fitting was carried out with the

SUBPLEX algorithm until a relative tolerance of 0.0001 was reached .

Performance of the parameter fittings

Each parameter fitting of the model requires several hundreds or thousands of solu-

tions of the IVP . The exact number depends on the initial parameter values . During

the Monte-Carlo simulation performed, one parameter fitting could be performed

per average in 1.5 seconds on a Pentium III with 800 Mhz.

Parameter
iI

Unit Initial value I Lower Limit I Upper Limit

vmax,l mM/s 1 .5000 0.1500 15.0000

Km,, mm 0.5000 0.0500 5.0000

vrrcax,2 mM/s 0.1500 fixed fixed

vmax 3 mM/s 1 .3650 0.1365 13 .6500

Km,3 mm 0 .7500 0.0750 7.5000
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10 20 30 40
Time [s]

Simulation output

Simulation with added noise

Figure 5.4. : Exact solution of the IVP and the solution with added noise. The

pulse of GLC was assumed at time t = 0.

The Monte-Carlo simulation shows that the parameters of the example model

cannot be identified uniquely. Figures 5 .5 and 5 .6 summarize the results obtained.

In Figure 5 .5 for each o£ the four parameters that had to be identified the final
parameter is printed against the final value of the relative least square . Each dot in
the graph represents a local minima identified by the parameter fitting algorithm .

If the parameters would be clearly identifiable, the correct parameter value would
have to coincide with small values of the relative least square. This is clearly not the
case for all four parameters . Parameters vmax,l and vmax,3 show an accumulation
of dots close to the correct value with a small least square error . Km,l and Km,3
show accumulation of dots for small least square error over the whole range of
parameter values . This indicates that these parameters are very unlikely to be
identified correctly. The figure also shows dependencies

In Figure 5.6 histograms are shown for each parameter showing the frequency of the
result parameter . It can be seen that the frequency of the result parameters is high
close to the correct value . This indicates that a large number of parameter fittings
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v»>~,1

vmax,3

0,0
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Squared Differente

	

Squared Differente

Km,3

x, .u.... .

20 40 60 ." 0 20 40 60
Squared Differente
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Figure 5 .5 . : Parameter values that resulted from 25000 parameter fittings versus
relative least square error . Vertical dotted lines represent the initial
parameter value that should have been identified .

increases the accuracy of the fitting . At least it would allow a conclusion about a
range of a certain parameter . Notice, however, that it requires high computational
costs to perform thousands of parameter fittings wich a larger model than the one
used for this illustrative example . It should be also noted that a different error
measurement d2 did not lead to significant changes in parameter identifiability.
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Figure 5.6. : Histograms showing parameters values that resulted from 25000 pa-
rameter fittings as optimal solutions . Vertical dotted lines represent
the initial parameter value that should have been identified .

Summary of the example model

It has been shown that the MMT software can be used efficiently for parameter
fitting of models that represent chemical reaction networks . 25000 parameter fittings
could be performed in 10 hours which means that one parameter fitting requires per
average 1.5 seconds of computation time .

The example system also shows that problems occur when identifying parameters
of chemical reaction systems . In the presented example the parameters are not
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clearly identifiable . A parameter fitting leads to a local minimum quickly but the
result is not useful without further investigations . It can be also concluded that
a large number of parameter fittings allows the identification of a range where the
parameters are likely to be located .

As a conclusion, it is important to investigate parameter identifiability e.g . with a
Monte-Carlo simulation . Some help also comes from sensitivity analysis . However,
sensitivity analysis depends on the initial parameter values of an parameter fitting .
Therefore excluding parameters with low sensitivities from parameter fitting is not
a solution . E.g . calculation o£ the sensitivities according to Equation 5.7 using
the correct parameter values at time t = 0 leads to the conclusion that &',j and
Km,,3 are more sensitive parameters than vmax,l and v,nax,3 . The resuts from the
Monte-Carlo simulation suggests the contrary.

5.5 .

	

Summary of solution techniques

This chapter describes the numerical methods used for solution of the IVP and
the parameter fitting problem. Whereas solution of the IVP does not provide any
problems, the parameter fitting problem is difficult to solve . A variety of methods
is combined in order to obtain a good performance . Many times a combination of
différent error functionals, weight factors, or parameter space reductions provided
the best solution for the parameter fitting problem . The MMT Software allows to

change such strategies during an parameter fitting and therefore provides a good

base for the strategy mix. More details about the software setup can be found in

Chapter 4 . The example model, however, clearly showed that the model setup is

crucial for parameter identifiability.
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6 . Modeling the Escherichia coli Network

The MMT software presented in Chapter 4 was used to develop models for chemical
reaction pathways . In order to test the tool models ranging from 3 dynamic vari-
ables with 13 parameters to 26 dynamic variables with 302 parameters have been
developed . More than 50 models have been created and analyzed .

This chapter first presents an isolated system for enzyme production with 7 differ-

ential equations . Following a top down approach (see Section 3 .4 .4), first two small

models for the PT-System and the reaction from F6P to FBP catalyzed by phos-

phofructokinase (PFK) of Escherichia coli are shown . Than two complex models of
glycolysis, PP-Pathway, and TCA-cycle of Escherichia coli with 16 respectively 26

dynamic variables are presented .

6 .1 .

	

A model for enzyme production

The MMT software can be used to model chemical reaction systems . Its intention is

the modeling of complex in vivo situations . The software, however, can also be used

for in vitro systems . Two models describing in vitro reactions for production of an

enzyme have been set up in order to check the functioning of the MMT software .

The system shown in Figure 6.1 produces 7 differential equations and 24 parameters .

Differential equations are defined for B1, C1, C2, C3, S, I and P . B2 does not have

any known influence on the system . Therefore, it is not integrated in the model. The

model was fitted to experimental data gained at the Institute of Biotechnology [12] .

During the experiment, 2 mM of substance S are converted to P via an intermediate I .

The cosubstrate C2 of the conversion reactions is recycled through C1. It uses an

additional substrate B1. The amount of B1 in the system at time 0 is 200 mM.

Because only 2 mM can be used (substrate S is only present wich 2 MM and C1

+ C2 + C3 = 0.4 mM), B1 does not effect the system significantly. An additional

flux v5 exists in the model that describes the degradation of C3. Therefore C1 +

C2 -I- C3 is decreasing with time .

Figure 6 .2 summarizes experimental data and parameter fitting results obtained
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Figure 6 .1 . : An in vitro chemical reaction system that is used to produce com-
pound P from substrate S through an intermediate I . The cosub-
stance C2 of the two reactions is recycled by two reactions catalyzed
by enzymes added to the in vitro system . Substance B1 is converted
to B2 during this process .

with the model. It tan be seen that the initial parameters could not represent any
efiects from the experiment . The reason is that for about two thirds of the 24
parameters of the model no literature values or experimental data was available .
Therefore, the initial guesses of these parameters have been set to 1.0 . In order to
get the fitted parameters, a strategy was applied that randomly changes parameter
values by a large percentage before each parameter fitting run . Using this strategy,
the presented result was calculated in less than 15 minutes . The identified Parame-
ters can now be optimized further with a strategy more suitable for parameter values
close to an optimal solution .

6 .2.

	

Glycolytic oscillations

Oscillations in the glycolysis of yeast have been reported and modeled before [95],
[35], [77], [94], [43] . The oscillatory behavior usually is assumed to result from
inhibitions of the enzyme PFK. This enzyme catalyzes the glycolytic reaction fromF6P to FBP. PFK is one of the most analyzed enzymes and is considered to playa significant role in metabolic regulation . However, the amount of available datadoes not allow a unique interpretation . The entry for PFK in the public databaseBRENDA lists 67 possible inhibitors . Most of these inhibitors cannot be measuredin the tell . It is therefore not possible to find a correct mathematical representation
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Figure 6.2 . : Experimental data and simulation results before and after parameter
fitting of parameters . Shown are values for the substrate S, interme-
diate I and final product P.

for the PFK. Also, a complete model including all these efects would include too
many parameters to be useful in a pathway model.

The models for PFK in the literature mentioned focus on the main inhibitors
and activators of the enzyme. Goldbeter and Lefever [36] developed a model that
describes the allosteric behavior of the enzyme (Figure 6.3) . The term allosteric
is used to describe a reaction in which the binding of a compound to the enzyme
influences the binding of the second. For PFK, a positive feedback exists, i.e . the
reaction rate from F6P to FBP increases if more FBP is available. Additionally,
cooperative behavior between the enzyme subunits is a necessary assumption for
instability in the PFK kinetics [94] .

It has been demonstrated by the authors with a simple mathematical model that
these two effectors on the enzyme PFK are sufficient for an oscillatory behavior .
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FBP

Figure 6 .3. : The model for PFK that results in oscillations . vi is a constant input
flux into the G6P pool .

A system with three dynamic concentrations was derived from this description in
order to examine the findings of the models . In addition to the concentrations of
F6P and FBP, the G6P concentration is included as a dynamical variable in the,
model. Together, a constant flux into G6P, the reactions G6P --} F6P, F6P -+ FBP,
and a flux exiting FBP are included in the model.

In the following, all concentrations are non-dimensionalized according to Goldbeter
and Lefever . The following equations result :

For v2 a reaction rate proportional to the concentration of G6P was assumed

V2 ° vmax,2 - G6P

For the flux from F6P to FBP, Goldbeter and Lefever developed the following equa-
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For the flux out of the F6P pool again a kinetic rate proportional to the F6P con-

centration is assumed

V4 = -vmam,4 - FBP

Details of the kinetic parameters eau be found in Goldbeter and Lefever [36] . The

authors use parameters from kinetic studies with PFK from yeast and Escherichia

coli. Figure 6.4 shows simulation results with the parameters from the authors .

The obtained results were compared with the experimental data from GLC pulse

experiments that are discussed in Section 2.2 . This data suggests an oscillatory

behavior of the intracellular G6P concentration after the pulse of GLC was given .

However, the F6P concentration in this experiment was below the detection limit .
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Figure 6 .4 . : Simulation results of the PFK mode, from Goldbeter and Lefever .
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The original system of ODEs from Goldbeter and Levefer consists of two differential
equations for F6P and FBP only. This allows analytic stability analysis of the
system . This analysis is based on a linearization at an equilibrium point . The
equilibrium points of the system can be calculated explicitly. With the help of this
stability analysis the occurrence and frequency of oscillations Gould be estimated.
The MMT software provides the derivatives of the model equations that are needed
for stability analysis .

In order to apply the PFK model to the experimental data in this work, parameter
values were sought that maintain oscillations in measured ranges of the concentra-
tions . Due to the large number of parameters, this problem Gould not be solved
explicitly. Instead, a large amount of simulations was performed . Parameters var-
ied in large ranges for these simulations . No parameter values Gould be found that
resulted in oscillations by maintaining a concentration relation of FBP < 0.002 as
found in the experiments . It was concluded that the PFK feedback inhibition by
FBP and the cooperative interactions between the subunits of PFK Gould not be
the reason for the oscillations suggested by the experimental data .

6 .3 .

	

Models for GLC uptake of Escherichia coli

The PT-System for GLC uptake of Escherichia coli consists of three phosphotransfer
reactions, Figure 6.5 . Next to the PFK, the PT-System is often thought to have the
strongest influence on metabolic regulation.

Two models for the system can be found in [83] and [82] . Both models describe
the system in a detailed way by including all three phosphotransfer reactions. Each
reaction step is divided into two steps :

1 . Building of the enzyme-substrate complex : The first step of each reaction
is the binding of the substances to the enzyme .

2. Dissociation o£ compounds from the complex : The second step of the re-
action is the dissociation of the products from the enzyme-substrate complex.

This straightforward approach is not based on the pseudo-steady-stane assump-
tion which would assume the substrate-enzyme complex concentration to be con-
stant [21] . This assumption is part of most kinetic rate equations . Without this
assumption, each reaction step can be modeled with a simple kinetic rate equation :
assume that the reaction rate is proportional to the substrate concentration of the

124



and the fluxes are modeled as

6.3 . MODELS FOR GLC UPTAKE OF ESCHERICHIA COLI

vl = ki

	

PEP - El

v2 = k2 Cl

v3 = %3 - C2

E1P

HPrP

HPr

Enzyme-Substrate'' Complexes

PT-System Enzymes

Figure 6.5. : The three phosphotransfer reactions of the PT-System in Escherichia
coli. El, HPr: unphosphorylated enzymes of the PT-System. ElP,
HPrP : phosphorylated enzymes . Arrows indicate the direction o£ the
irreversible reactions assumed in the model .

educt, or to the product of the concentrations if two educts are present . This is
based on a common assumption that the substance concentration is proportional to
the reaction rate in chemical reactions . The stoichiometric matrix of the PT-System
reads as
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G6P F6P El ElP HPr HPrP Cl C2 C3
vi 0 0 -1 0 0 0 1 0 0
v2 0 0 0 1 0 0 -1 0 0

v3 0 0 1 0 0 1 0 -1 0

v4 0 0 0 -1 -1 0 0 1 0

v5 0 0 0 0 0 -1 0 0 1
VG 1 0 0 0 1 0 0 0 -1

v7 -l 1 0 0 1 0 0 0 -l
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A detailed model that includes enzyme substrate complexes as dynamic variables

can lead to différent results if compared to models based on thé pseudo-steady-state

assumptions . This, however, strongly depends on thé total enzyme concentration .

If thé concentration is high, a large amount of enzyme could be e.g . stored in

thé enzyme-substrate complex. A sudden change in thé growth conditions could

change this situation and free a significant amount of enzyme and substrate from

thé complex . This phenomenon can only be explained with detailed models of these

processes .

For thé experimental conditions of thé available data (pulse experiments) it is likely

that thé above mentioned effects occur. Before thé pulse was given in thé experi-

ments, tells were grown on GLC limited conditions . Relatively high PEP concentra-
tions coincided with low G6P concentrations . This suggests that thé concentration
of thé PEP-enzyme I complex is high whereas concentration of thé GLC-HprP com-
plex remains low . Due to thé sudden and large increase in thé extracellular GLC
concentration this situation is likely to change . In order to evaluate thé effects of
such changes, both models presented in [83] and [82] have been analyzed with thé
experimental data sets . The models use GLC, PEP and PYR as input variables and
only model thé PT-System plus thé flux from G6P to F6P. For thé flux from G6P
to F6P, a reversible Michaelis-Menten equation was assumed.

In Figure 6.6 two simulations produced with thé model in [83] are shown and
data from a GLC pulse experiment as described in Section 2.1 . Initial parameters
were also taken from [83] . The upper graph in thé Figure shows a result from an
parameter fitting for thé first 17 seconds of thé experiment. The result shows good
agreement between thé model and thé data for this time frame. The second part of
thé experiment cannot be reproduced by thé model. I£ thé complete data file until
36 seconds was used during parameter fitting, no satisfactory fit was achieved . The
lower part of thé figure shows thé Best result obtained wich parameter fitting casing
thé SUBPLEX algorithm . It tan be clearly seen that thé first part of thé model does
not agree very well . The parameter fitting algorithm Gould not identify parameters
that fit thé first and thé second part of thé data.

v4 = k4 E1P - HPr

v5 = k5 HPrP - GLC

V6 = k6 C3

vf
G6P F6P_ V
KmS T KmP (6 .4)

V7
-

1 + GGP ,+ F6P
Kms KmP
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Parameterization for the time
G6P [mM]
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l,2 ~
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0,6-
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0,0

This part of the
data not used in
Parameterization

-5 to 17 s

0 10 20 30 40
Time [s]

Parameterization over total time

G6P [mM]

1,6

l,4 1

1,2

l,0

0,8

0,6

0,4-

0,2-

-a- Experimental Data

	

Simulation output

Figure 6.6. : Simulation results of the PT-System model. The upper graph results
from a parameter fitting for the time frame -5 to 17 seconds. The
lower graph is £rom an parameter fitting for the complete time frame
from -5 to 36 seconds .
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It has been discussed previously in Section 2.2 that there is no biological Inter-

pretation for the time window 17 to 32 seconds where a high G6P concentration

coincides with a low PEP concentration . Both models for the PT-System also were

not able to explain this phenomenon . The hypothesis that effects of the three phos-

photransfer reactions and the enzyme-substrate-complexes alone are responsible for

such behavior has to be rejected .

6.4 .

	

A model of glycolysis, PP-Pathway, and
TCA-cycle

A complex model for Escherichia coli was constructed . The goal was to find the
effects of metabolic regulation that are able to explain the behavior seen in the data .
Figure 6.7 shows the reactions of the central metabolic pathway of Escherichia coli
that were included in the model.

Simplifications have been made at the following points (compare with the more
detailed pathway in Figure B.1) :

" The PP-Pathway has been reduced to one compound called PPP (black-box
formulation) . A flux is assumed from G6P to PPP and a flux from PPP
directly to PEP. This is a simplification because there are additional pathways
from the PP-Pathway back to glycolysis (F6P, GAP) . Additionally, a flux from
PPP to biomass production is assumed in the model.

" The two reactions PYR --> MAL and MAL -+ OAA are reduced to one reaction .

" The TCA-cycle is reduced to OAA, ACoA, CoA, CIT, and a compound called
TCA (black-box assumption) . From CIT, a flux enters TCA. TCA is then
directly converted to OAA. Fluxes to biomass from the TCA-cycle are sum-
marized in one equation, a flux from CIT into biomass.

" Fluxes into biomass are summarized in three kinetics : from PPP, CIT, and
PYR. It keeps the number of parameters small if compared to an approach
where many compounds have a flux into biomass .

Additionally, a reaction that includes maintenance and production o£ ATP has
been assumed. Together, the model contains 16 compound concentrations . The
kinetic rate equations used contain a total of 159 parameters . The rate equations
are summarized in Table 6.1 . Formulas for these equations are summarized in Ap-
pendix C.
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Table 6 .1 . : All reactions of the model and the rate equations used in the mathe-
mfitical model.

Reaction Rates vi

	

- - II Rate equation

GLC + PEP t-> G6P + PYR
G6P <-> F6P
G6P -> PPP

PPP + ATP -> BM + ADP

PPP t-> PEP
F6P + ATP <--> FBP + ADP

FBP t-> DHAP + GAP
DHAP H GAP
GAP + NAD t-+ DPG + NADH

DPG + ADP t-> 3PG + ATP
3PG t-> 2PG
2PG 14 PEP

PEP + ADP +-~ PYR + ATP
PYR + ATP <-> OAA + ADP
PYR + ATP <--> BM + ADP

PEP -4 OAA
OAA + ATP ~ PEP + ADP
OAA + ACoA ++ Cit + CoA

CIT -3 TCA

CIT + ATP <-> BM + ADP
CIT <--> OAA

ATP fa ADP

see text
reversible Michaelis-Menten
Michaelis-Menten

Michaelis-Menten, two substrates,
irreversible
reversible Michaelis-Menten
ordered inhibited Bi-Uni-Kinetic (inhibition
by PEP)

ordered Uni-Bi-Kinetic
reversible Michaelis-Menten
ordered Bi-Bi-Kinetic

ordered Bi-Bi-Kinetic
reversible Michaelis-Menten
reversible Michaelis-Menten

ordered Bi-Bi-Kinetic
ordered Bi-Bi-Kinetic
ordered Bi-Bi-Kinetic

See text
see text
ordered Bi-Bi-Kinetic

Michaelis-Menten, two substrates,
irreversible
ordered Bi-Uni-Kinetic
reversible Michaelis-Menten

reversible Michaelis-Menten



As listed in the table, the reaction from F6P to FBP is described with an ordered
inhibited Bi-Uni-Kinetics which has the following form [63] :

VF6P-+FBP

analyzed in detail .

6.4 . A MODEL OF GLYCOLYSIS, PP-PATHWAY, AND TCA-CYCLE

(F6P - ATP - B9 )
vmax

	

F6P * ATP + Km F6P - ATP -}- Km,ATP - F6P . . .

. . . + VTKeq - (Km,FBP +FBP - (1 + I F6P) + (PEP)nl

	

(6.5)

This reaction assumes an inhibition by PEP. The importance of this inhibition was

For GLC uptake, PEP --> OAA, and OAA --> PEP, equations from [59] have been
used . In [59], the derivation of these rate equations is based on a combination of
algebraic equations and fuzzy logic in order to incorporate qualitative knowledge
into the enzyme kinetic equations . For the reaction from OAA --> PEP an algebraic
equation was combined with a set of fuzzy rules . These fuzzy rules described a strong
inhibition of the reaction by high concentrations of ATP, a phenomenon derived from
experimental data presented also in [59] . The authors suggested to include the fuzzy
rules in the algebraic equation by multiplying v,,,ax with a factor apck of the form

1 .0

	

, for

	

ATP

	

< 0.9

apck =

	

(1 - AT0P20.9 ) , b .ATP

	

, for 0.9<

	

ATP

	

<1.1

1 .0 - apck

	

, for 1.1 <

	

ATP

where b was estimated by the authors as 0.24 mM-1 . Notice that even though the

the authors derived the equation from fuzzy rules, they have chosen to formulate the

equation as a simple piecewise function . The equation for apck is problematic because

it attains negative values for large ATP concentration . To avoid this problem, the

following equation has been used for apck to include a strong inhibition by ATP:

1
apck = 1 + eb,ATP

The overall equation for the reaction OAA -+ PEP is then

VOAA-+PEP = vmax
Kl - ATP - OAA - K2 - PEP - ADP

(1 + eb-ATP) . p

(6.6)

(6.7)



Notice that this equation assumes a strong inhibition of the PT-System for high
PEP-concentrations which is not reported in the literature . This assumption will be
discussed later .

The second equation used from [59] describes the reaction from PEP to OAA. In
this case, however, the algebraic form of the equation was used because it shows the
same behavior as the equation based on fuzzy logic . The equation reads as

VPEP->oAA =

	

PEP

	

K1 +K2 - ACoA + K3 - FBP + K4 - ACoA - FBP
(6.11)Km + PEP

	

1 + K5 - ACoA + K6 - FBP

The equation is based on a Michaelis-Menten type kinetics . The second fraction is a
formula for the maximum reaction rate v,»,ax of the reaction . The equation assumes
inhibitions of the reaction by ACoA and FBP which is modeled by variation of v�,,ax .

6.4.1 .

	

Fitting the model to data from the steady state

After defining initial parameters, a parameter fitting was defined for fitting the modelto the steady state of the underlying experimental data, i.e . to the time frame -5 to0 seconds . As expected, the results showed an overfitting problem. The steady state
measurements are not sufficient for estimation of all kinetic parameters . Thereforethe obtained fitted parameter values are only useful as initial values for a parameterfitting of the dynamic data (after the GLC pulse) .
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with

P = 1 + K2 - ATP + K3 . OAA + K4 - ATP - OAA

+ K5 - PEP + K6 - ADP + K7 - PEP - ADP (6.8)

The same equation was used for the uptake of GLC with the PT-System :

Ki-PEP-GLC-K2-G6P-PYR
VGLC-3G6P = vmax ' (1 + eb'PEP) . p (6 .9)

with

P = 1 + K2 - PEP + K3 - GLC + K4 - PEP - GLC

+ K5 - G6P + K6 - PYR + E7 - G6P - PYR (6.l0)
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Figure 6.8 shows an example of a steady state simulation for PEP. All other com-
pounds also reached steady state about 0.5 seconds after thé simulation was started .
Because some compound concentrations do not start at time t = -5s at their steady
state values, thé system has first to evolve to thé steady state . For thé compounds
where measurements are available, thé initial conditions have been taken from these
experiments . As shown for PEP in Figure 6.8, thé model calculates a slightly dif-
ferent steady state . It is reached after about 0.5 seconds .

Figure 6 .9 summarizes thé différences between thé simulated steady states and thé
steady states measured in thé experiment for each individual compound concentra-
tion . The steady state of thé réaction system is represented by thé model with good
accuracy . Problematic is thé fact that this good representation of thé steady-state
is usually lost if thé model is fitted to thé data of thé pulse experiment .

The results obtained for thé steady state show a GLC uptake rate of 0.58'M .
The estimated range for this value from thé experimental data was calculated as
0 - 0.9 'M . If flux estimation are available, it is also possible to include this flux as a
constraint in thé parameter fitting for thé steady state . Because in thé experimental
data, no measurements from flux-analysis are available, such a constraint was not
used .

PEP [mm]
1 1 51

0,5

. . ..... . . . ...... . .

-4 -3 -2 -1
Time [s]

0

Experimental data

	

--- Simulation output

Figure 6.8. : Simulation results for thé steady state. Shown is thé intracellular PEP

concentration .
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G6P F6P
FBP DHAP

GAP 3PG
2PG PEP PYR ADP ATP CIT

Compound

Simulation output

	

" Experimental data

Figure 6 .9. : Steady state concentrations of the intracellular compounds and values
from experiments.

Additionally, a futile cycling by the reactions PEP -3 PYR, PYR -> OAA, OAA
--> PEP is calculated by the model [18] . The size of this flux is 0.25 "M . Due
to the setup of the system, one ATP is lost in this futile cycling . Notice that this
result can depend on the simplifications made for the model for the reactions from
PYR to the TCA-cycle . Also, energy limitation does not occur in the steady state
situation because the additional reaction ADP <-+ ATP can provide enough energy.
This reaction is assumed to include maintenance terms (i.e . usage of ATP) and the
respiratory chain (i.e . production of ATP) . The respiratory chain was included in
the model with this single reaction . As a result, the steady state situation does not
give new insights in the system .
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6.4.2.

	

Fitting the model to data from a pulse experiment

The described model was fitted to the experimental data of the GLC pulse experi-ment shown earlier. The Locus was again on the PT-System and the PFK inhibition.Both reaction rates include a strong inhibition by PEP. For PFK, a strong inhibitionby PEP is reported in [8] . The inhibition of the PT-System by PEP is hypotheti-cal . It could represent a situation where the tell needs to decrease the uptake rate
because intracellular compound concentrations are too high .

Oscillations of the compounds

During simulations, the inhibition constants KI in the Equations 6.9 and 6.7 have
been set to 1.0 mM indicating a weak inhibitioy effect . In contrast, parameter n
in Equation 6.5 has been set to 8.0 with an value for KI of 1.0, indicating a strong
inhibition of PFK by PEP as reported in the literature . As in the Goldbeter models
described earlier, the inhibition of PFK leads to oscillations in glycolyses . Fig-
ure 6.10 shows simulated values for G6P, PEP, and ATP. All other compounds show
oscillations of the same frequency. The period of oscillations is in the range of 20
seconds and therefore in a range that is of interest for the data from the GLC pulse
experiments .

Dynamit behavior for various inhibition constants

Inhibition of PFK by PEP leads to oscillations, however, this effect was not sufficient
to fit the model to the experimental data . parameter fittings were performed that
also included inhibition of the PT-System by PEP as mentioned before. In general
results from the parameter fitting using the dynamic data from the pulse experiment
do not represent the steady state situation, even if the initial parameter vector
showed good representation of the steady state . Figure 6.11 compares three different
results from parameter fittings :

1 . Inhibition of the PT-System and PFK: If both reactions are assumed to
be inhibtted, a dynamic behavior was achieved that cornes somewhat close to
the experimental data. Notice that a steady state for the PEP concentration
actually seems to evolve right before the pulse (see Figure 6.11, top right, PEP
concentration right before the pulse at time t = 0) . Apparently the system has
to reach a steady state first as could be seen in the steady state simulations
above .
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Figure 6 .10 . : Simulated values over a time period of 250 seconds . Oscillations
result from the strong inhibition of PFK by PEP.

2 . Inhibition of PFK only : If the inhibition of the PT-System is not assumed
but only an inhibition of PFK by PEP, the qualitative behavior of the system
does not match the experimental data anymore . The result shown is obtained
through an additional parameter fitting without the PT-System inhibition.
Especially the dynamics of the G6P concentration are effected .

3 . Inhibition of the PT-System only : The third results shown in the figure is
obtained by assuming inhibition of the PT-System by PEP and neglecting
inhibition of the PFK by PEP. The result from parameter fitting shows thatthe dynamics seen in the experiment cannot be explained with this model . Asin the second case, the dynamics of the G6P concentration change significantly.
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Simulation result

Figure 6.11 . : Experimental data and simulated values for G6P and PEP over a

time period of 35 seconds . From -5 to 0 seconds a steady state

situation was present in the experiment, the pulse of GLC was given

at time t = Os. A qualitative behavior close to the experimental

data occurs only if inhibitions for the PT-System and the PFK are

assumed .
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6.4.3 .

	

Summary of the complex model of glycolysis, PP-Pathway,
and the TCA-cycle

From the complex model with 16 dynamic variables and 159 parameters of the central
metabolic pathway of Escherichia coli the following conclusions were obtained :

1 . The model provides a good description for the steady state before the GLC
pulse.

2 . The model fails to explain dynamic behavior of the system after the GLC
pulse.

3 . The system shows strong dynamics and oscillations . These efects can result
from one single inhibition (PT-System or PFK) and increase if more efects
are included in the model .

4 . The fit obtained for the steady state situation resulted from parameter fittings
with little computing time . This suggests that the parameter fitting algorithm
might not be the problem. for the unsatisfactory fit in the dynamic situation . It
is concluded that the model must be modified in order to explain the measured
values .

6 .5 .

	

A detailed model of the central metabolic
pathways

The complex model described in the last section suggests that inhibitory efects
have a strong influence on the dynamics of the system . Following this conclusion,
the model was extended to include more regulating phenomena of the pathway. This
extensions mainly focused on more detailed descriptions of the PP-Pathway and the
TCA-cycle and additional fluxes to biomass. Also, NAD, NADH, NADP, NADPH,
and AMP were included in the model to follow energy charges . Figure 6.12 and 6.13
show all reactions of the system . The following assumptions have been made for the
model :

1 . The flux in the PP-Pathway was modeled with a reaction from G6P to 6PGL,
replacing the black-box assumption of the above model . The flux back to
glycolysis was modeled by the reactions from 6PG -+ GAP and 6PG --> PYR.
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140

NADH

	

NAD

	

NADPHX NADP

ADP

	

~1, ATP

	

AMP

	

ADP

NADPHX NADP

ADP ATP

AMP

ATP

NADH X NAD

AMP ADP

ADP

Figure 6.l3 . : Reactions for the nucleotides included in the extended model . NADH
and NADPH each are assumed to convert 3 ADP or 3 AMP into
3 ATP respectively 3 ADP. AMP and ATP can be converted into
2 ADP. These reactions simulate the respiratory chain and energy
conversion reactions in Escherichia coli .

2 . The TCA-cycle was assumed to be active in the branched form, i.e . the cycle
is not closed . See Figure 6.12, where no connection between OXO and FUM
exists . For growth of Escherichia coli this is supposed to be the form of action
of the cycle [65] .

3 . In Figure 6.13 all reactions for interconversion of the nucleotides are shown.
Reactions for the production of ADP and ATP from NADH, NADPH, AMP,
and ADP have been assumed for energy balancing .

4. Biomass production represented by the fluxes from 6PG, 3PG, PYR and OXO
to biomass has been assumed to consume ATP. For the reaction from 3PG and
PYR to biomass, the ATP consumption was a parameter that was adjusted
during the process of parameter itting. NAD, NADP and NADH, NADPH
were not assumed to be necessary for biomass production but are only included
in the reactions in the central metabolic pathway.
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The equation includes a product inhibition by PYR of the PT-System . Because
the phosphotransfer reactions of the uptake system are reversible, this assumption
has a biological background . However, kinetic constants of the system from [82]

mentioned earlier suggest that the inhibition is not strong .

Equation 6.5 for the reaction from F6P to FBP was replaced by an equation

from [47] . The equation is based on a four state allosteric enzyme model and in-

cludes AMP and ADP as activators and ATP as an inhibitor . The equation takes

the following form:

ATP
VF6P->FBP

	

=

	

vmax ' ATP + Km,ATP ' (1

	

ADP
+ KA,ADP )

F6P
(F6P + P) ' (1 + ~1+

	

)
(6.14)

where

ATP

	

ADP +, AMP1 + j{1,ATP + K1,ADP

	

K1,AMP
P = Km,F6P '

	

ADP

	

AMP1 + K2,ADP + K2,AMP

(6 .15)

For the reaction rate from 6PG to biomass and from PYR to biomass, the following

equation for synthesis of monomeric building blocks was used [5] :

6PG ATP

	

(6 .16)
V6PG-+BM = vmax '

	

'
Km,6PG Km,ATP

Equation 6.9 for the PT-System has been changed to the following form :

Kl - PEP - GLC - K2 - G6P - PYR
= ' (6.12)VGLC-+G6P vmax (1 +xR) . P

with

P = 1 + K2 - PEP + K3 - GLC + K4 - PEP - GLC

+ K5 - G6P + K6 - PYR + K7 . G6P - PYR (6.l3)
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A hypothetical equation was used for the Raction from PYR to ACoA. The equation
is derived from the ordered Bi-Bi-Kinetics but includes CoA as a third substrate:

vf - (PYR - NAD - CoA - ACoKNADH )

VPYR->ACoA = P
(6.17)

ACoA
P

	

=

	

PYR - NAD . (1 +
K

	

) + Km,NAD ' (PYR+ Ki,PYR) + . . .
i,ACoA

+

	

Km,PYR ' NAD + Km,CoA - CoA + . . .

+

	

v.. vKe

	

' [Km,ACoA ' ACoA - (1 +

	

P
p R

) +
9

	

K e

+

	

NADH -

	

Km,ACoA ' (1 + Km,PYR ' NAD ) + ACoA - (1 .+
NAD )

KiA ' KmB

	

Ki,NAD

	

]

Reaction kinetics of the model are summarized in Table 6 .2 .

Table 6.2. : All reactions of the model and the rate equations used in the mathe-
matical model .

Reaction rates vi

	

Rate equation

GLC + PEP <-> G6P + PYR
G6P t-> F6P
G6P ---> 6PGL

6PGL ++ 6PG
6PG 4--> PYR
6PG <-> GAP

6PG + ATP fa BM + ADP
F6P + ATP t-> FBP + ADP
FBP --> F6P

FBP <-> DHAP + GAP
DHAP <--> GAP

see text
reversible Michaelis-Menten
ordered Bi-Bi-Kinetic

reversible Michaelis-Menten
reversible Michaelis-Menten
reversible Michaelis-Menten

See text
see text
Hill-Kinetic

ordered Uni-Bi-Kinetic
reversible Michaelis-Menten

continued on next page
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Table 6.2 . : (continued)

Reaction rates v2 Rate equation

GAP + NAD 4--> DPG + NADH ordered Bi-Bi-Kinetic

2PG + ADP <4 WG + ATP
3PG <--> 2PG
3PG + ATP <-> BM + ADP

ordered Bi-Bi-Kinetic
reversible Michaelis-Menten
ordered Bi-Uni-Kinetic

2PG f-> PEP
PEP + ADP e~ PYR + ATP
PYR + ATP <-> PEP + AMP

reversible Michaelis-Menten
ordered Bi-Bi-Kinetic
ordered Bi-Bi-Kinetic

PYR + NADH i-> LAC + NAD
PYR --> ACE
PYR + ATP <-> BM + ADP

ordered Bi-Uni-Kinetic
Hill-Kinetic
see text

PYR + NAD + CoA +> ACoA +
NADH
MAL + NAD ++ PYR + NADH
MAL + NADP ~-> PYR + NADPH

see text

ordered Bi-Bi-Kinetic
ordered Bi-Bi-Kinetic

MAL + NAD +> OAA + NADH
PEP --> OAA
OAA + ACoA +> CIT + CoA

ordered Bi-Bi-Kinetic
see text in the model of the last section

ordered Bi-Bi-Kinetic

CIT +4 ISC
ISC + NADP +> OXO + NADPH

MAL --> BM

reversible Michaelis-Menten
ordered Bi-Bi-Kinetic
Hill-Kinetic

MAL --> FUM
ATP + AMP +> ADP

ADP + NADH + ATP + NAD

Hill-Kinetic
ordered Bi-Uni-Kinetic
ordered Bi-Bi-Kinetic

ADP + NADPH +> ATP + NADP

AMP + NADH +> ADP + NAD

AMP + NADPH +> ADP + NADP

ordered Bi-Bi-Kinetic
ordered Bi-Bi-Kinetic
ordered Bi-Bi-Kinetic
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6 .5.1 .

	

Fitting the model to data from a pulse experiment

The presented model includes 26 dynamic variables and 302 parameters . The only
data input to the simulation is the extracellular GLC concentration and initial con-
ditions for the compounds. All other compound concentrations are calculated from
the model equations . Solutions of the parameter fitting problem were intended to
give insight in the dynamic behavior of the system after the GLC pulse. Due to the
low number of data points compared to the number of parameters, an parameter
fitting is difficult . During the parameter fitting, the weights for the error functional
have been set to high values for G6P and PEP. For these metabolites, weights factors
100 to 500 times larger than for all other metabolites have been used . During the
parameter fitting using the SUBPLEX algorithm of the MMT software, the following
strategies have been applied:

Scale parameter changes by their sensitivities as described in Section 5.4.

" Restart the parameter fitting and increase ranges of fitted initial conditions
and parameters .

Perform the parameter fitting stepwise, i .e . fit for the first 5 seconds of the
steady state . Then use this result to fit for the first 8 seconds that includes the
steady state and the initial reaction of the system to the GLC pulse. Continue
to increase the time window after the parameter fitting is finished .

Using this strategy, a reasonable fit of the simulated G6P and PEP concentrations
to the measured values could be achieved for the time frame -5 to 25 seconds .
Figure 6.14 shows the best result obtained with parameter fitting . Figure 6 .15 shows
the other compounds of the model where experimental data was available . Due to the
low priority during the parameter fitting, the simulated metabolite concentrations
only remain in a reasonable range if compared to the experimental data . Figure 6 .16
shows the GLC uptake rate and the fluxes around G6P.

If the parameter fitting was extended to a time frame of more than -5 to 25 seconds,
the model failed to represent the experimental data. Therefore, the model is not
capable to explain the complete dynamics of the experiment .

6 .5.2 .

	

Conclusions and summary of the model

The model shown was able to describe well the intracellular G6P and PEP concen-
trations for the time -5 to 25 seconds . The following conclusion are made:
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..a- Experimental data

	

-- a Initial simulation

,,

	

Parameterization result

Figure 6 .14 . : Initial simulation results and results obtained from parameter fit-

tings . parameter fittings for a time window larger then -5 to 25

seconds was not successful.
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6.5 . A DETAILED MODEL OF THE CENTRAL METABOLIC PATHWAYS

1 . The model describes the dynamic behavior of G6P and PEP after the GLC
pulse. This includes the increase of the PEP concentration at time 10 seconds
and the high G6P concentration for low PEP concentrations in the time frame
18 to 25 seconds .

2 . Analysis of the fluxes that occur in the simulation shows that the flux through
the PP-Pathway is very high (Figure 6.16) . During the steady state, about
50% of the GLC that is taken up enters the PP-Pathway whereas the other
50% are processed by glycolysis . After the GLC pulse, however, the flux
through glycolysis remains almost the same whereas the flux through the PP-
Pathway increases almost 10-fold . Therefore about 90% of the GLC taken up
is processed by the PP-Pathway.

3 . The increase in PEP after the initial drop is a result of a high flux from PYR
to PEP. There was no net-flux from OAA to PEP in the model. Instead, a net-
flux in the other direction resulted from the parameter fitting . Additionally, a
strong flux was present from OAA to MAL and from MAL to PYR. Therefore
OAA is one of the sources for the PEP production . This conversion from OAA
to MAL and further to PYR and PEP is already present during the steady
state . However, the rate of these anaplerotic reactions is estimated to high
because the OAA concentration decreases already during the steady state . In
general OAA may be important to provide PEP under certain environmen-
tal conditions. Intracellular measurements of OAA could help to answer the
question of intracellular OAA concentrations but no data was available for this
parameter fitting .

4 . The high G6P concentration that coincides with the low PEP concentration
between time 18 and 25 seconds is an effect of a lover flux from G6P to 6PGL.
As mentioned above, about 90% of the GLC taken up enters the PP-Pathway
through this reaction in the simulations . This reaction is NADP dependent .
NADP decreases after the pulse and may be a reason that leads to a decrease
in the reaction rate £rom G6P to 6PGL. The high G6P concentration at time
18-25 second follows .

5 . Simulation of the nucleotide concentrations AMP, ADP, ATP, NAD, NADH,
NADP, and NADPH were problematic . Finding a balance between the reac-
tions is difiicult . Either a situation occurred where high ADP, AMP, NADH,
and NADPH concentrations and low ATP, NAD, and NADP concentrations
were present or the contrary. Even though the results suggest that the de-
creasing NADP concentration is a reason for the decreased flux from G6P to
6PGL, a correct model for nucleotide concentrations has not been found yet .
Also, the nucleotide concentrations that could be measured show large errors
and it is not clear if dynamic changes in these concentrations are present at
all .
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7 . Summary and Future Development

The large increase in available quantitative and qualitative knowledge has been men-
tioned at the beginning of this work . The data results from genomics, proteomics,
transcriptomics, metabolomics and fluxeomics . These techniques are more and more
automated and create large amount of data that has to be analyzed .

During this work, a modeling framework was developed . The approach is based
on a Metabolic Modeling Tool (MMT) software . The MMT software was especially
developed for a specific kind of models for chemical reaction kinetics : non-stationary
pathway models of microorganisms based on chemical reaction kinetics . The models
lead to a system of ODES. Simulations require then the solution of an IVP and
parameter fittings require numerical algorithm due to the non-linearity of the ODES.
These algorithms are also part of the MMT software .

All information of the models, including the chemical reaction networks, mathe-
matical equations, parameters, and results of simulations and parameter fittings,
are stored in a relational database . The software allows to build new models or
to modify existing models in a very short time . Because the software sets up the
IVP automatically and creates a C-code using MAPLE 6, the user only needs to
define the model and specify the parameters . The approach signifiicantly reduces the
number of errors usually made when typing equations by hand into a C-code.

Summary of the presented models

It is demonstrated in this work how the software is set up and what the advantages
are . An example is given that illustrates the database setup and the mathematical
problem that results from the modeling approach. An example is also given to

illustrate the probem of fitting a complex pathway model to experimental data .

Models of different dimension are shown. The results show that the MMT software

is useful
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. for modeling chemical réaction kinetics of différent dimensions .

for simulations with defined models .

to fit models with several hundred parameters to experimental data .

" to define a large amount of models in a short time and compare their results
obtained through parameter fitting.

The analyzed and developed models show that

" the exact knowledge of metabolic control effects as inhibitions and activations
has very strong effects on the dynamics of a metabolic pathway.

" small models are able to describe phenomena of isolated systems well .

" large models are able to describe some effects at least in a qualitative way but
produce questionable results due to thé lack of available data, e.g . for thé flux
through thé PP-Pathway.

Project continuation

The Institute of Biotechnology is further developing thé MMT software . A paral-
lel analysis of competing models will be included in thé near future . Further, thé
performance of thé software will be increased by parallelization methods and recod-
ing of parts of thé routines . Two major points for improvement have already been
mentioned. One is thé storage of results . Instead of storing all results, thé C-code
produced by thé MMT and parameter values should be stored . This saves thé large
amount of time needed for thé setup of thé system of ODES from thé database con-
tent as well as thé storage space needed for thé simulation results . Results can then
be reproduced in a few seconds . The second point is thé inclusion of smoothing
splines for experimental data .

In coopération with thé University of Siegen a project "Modellierung Metabolischer
Netzwerke auf der Grundlage von in vivo, invitro and in silico Daten", financed
by thé Deutsche Forschungsgemeinschaft (DFG), will begin this year . During thé
project, an extended software tool for models of cellular systems will be developed .

Additionally, experiments with new and existing strains are performed and ana-
lyzed. Methods from genomics, proteomics, transcriptomics, fluxeomics, and metabolomics
are used at thé Institute of Biotechnology. An additional focus is thé optimization of
fermentation conditions for thé new strains . The data generated from metabolomics
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will be analyzed with the MMT software . For model set up and parameter identifi-cation data from the other methods will be helpful .

The future of metabolic engineering

In the future, tools similar to the MMT software will be used to analyze the largeamount of available data. These tools will be able to describe cellular systems withincreasing accuracy and detail . A prerequisite is, however, continued effort frombiologists, chemists, engineers, mathematicians, physicists, and computer scientistsin this field . A close cooperation of scientists from these research areas is absolutelynecessary for further progress .

A second prerequisite is a continued inflow of money that is necessary for support
of such research activities . Methodological projects in the field of biotechnology
usually require high investments over a long period of time . Because many products
in the field are used in medical treatments, significant revenues in this field are likely
to produce a continued inflow of money for research projects .

A final comment

During the time this thesis is written there are already new methods for analyzing
microorganisms coming into focus. A method termed interactomics is used with in-
creasing success to examine all possible interactions between proteins in a tell . The
method is based on analysis with coimmunoprezipitations or the Yeast-Two Hybrid-
Analysis [30] . For Escherichia coli, 5.6 Billion so called "Bait-Prey" combinations
have been tested with the Yeast-Two Hybrid-System and 1200 protein-protein in-
teractions have been identified .

Another promising field is the single-molecule analysis [92] . The method is based
on fluorescent analysis of molecules stimulated by a laser beam. Thus intracellular
concentrations of molecules can be measured with high accuracy. Additionally, the
method allows to view the location of molecules inside a cell . The method has a
great potential for analyzing compartmentation in cells .

Interactomics increases the complexity of models significantly Because protein-
protein interactions have to be included in models for an exact description of regu-
latory effects . Data about distribution of molecules inside tells will lead to models
Chat include spatial and conceptual compartmentation of tells . These two examples
show how fast the field of metabolic engineering is progressing and again empha-
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sizes the need of software tool for developing complex models of chemical reaction
kinetics .
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A. Table of Symbols and Abbreviations

A.1 .

	

Metabolites and nucleotides

Table A.1 summarizes the symbols used in this work and shows the page number
where the symbol is defined or introduced . Table A.2 shows abbreviations of the
metabolites, Table A .3 abbreviations of nucleotides and other compounds . The last
column "Substance ID Internet" contains the official compound ID as used in public
databases on the Internet .

Table A.1 . : Symbol table . The page number in column 3 refers to the context of
the definition . Therefore the definition of the specific symbol might
sometimes be found on the following page .

Symbol

	

11
Meaning

	

I Definition

C, ci

C, Ci

C
Co

dk

Dx

E

concentration of a compound (for a list of
compounds see Table A.2)
derivative of a compound concentration
with respect to time (for a list of
compounds see Table A.2)
vector of intracellular compounds
vector of intracellular compounds at initial
time to
measure for the distance between calculated
and measured values for calculation of the
error functional F
total differential operator
distance between calculated and measured

value
vector of enzyme concentrations

see page 36

see page 41

see page 40
see page 102

see page 103

see page 93
see page 94

see page 40

continued on next page
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Table A.1. : (continued)

Symbol

	

11
Meaning

	

I Definition

Ei

F
IVP
K, Km,*
KI, KI,*
M
MMT
N
Np
Nt

N�

ODE
P
PFK
PT-system
Po
PP-Pathway
S
SQL
TCA-Cycle
v
Vi
VA-~B) VA->B

v(C, S, E, P)

Vmaxi Vmax,*

WJ

elements of the vector of enzyme
concentrations E
error functional
initial value problem
kinetic parameter, saturation constant
kinetic parameter, inhibition constant
measure matrix
metabolic modeling tool
stoichiometric matrix
number of parameters of the vector P
number of intermediate time steps ti where
measurements exist
number of reaction rates vi summarized in
the flux vector v
ordinary differential equation
vector of kinetic parameters
phosphofructokinase
phosphotransferase system
vector of initial kinetic parameters
pentose-phosphate pathway
vector of extracellular compounds
standard querylanguage
citric-acid-cycle
vector of fluxes (reaction rates)
element i of the flux vector v
flux vector (reaction rate) corresponding to
the reaction from A to B . -+ (4-) indicates
an irreversible (reversible) reaction
vector of reaction rates vi as functions of C,
S, E, P.
kinetic parameter, maximal velocity of a
kinetic rate v
weighting factor in the error functional F

see page 40

see page 103
see page 81
see page 40
see page 46
see page 94
see page 8
see page 38
see page 107
see page 94

see page 36

see page 41
see page 40
see page 119
see page 18
set page 109
see page 18
see page 40
see page 60
see page 18
see page 36
see page 36
see page 36

see page 40

see page 40

see page 103
continued on next page



Table A .1 . : (continued)

Symbol Meaning

Yi

Y

calculated value at time ti

	

see page 94
experimental data at time ti

	

see page 94
matrix of all experimental data points

	

see page 94

Table A.2. : Table of abbreviations : Metabolites .

A.1 . METABOLITES AND NUCLEOTIDES

Definition

Abbreviationll Full Name

	

I chemical

	

I Substance
Composition

	

ID Internet

2-Phospho-D-glycerate,
D-Glycerate 2-phosphate
3-Phospho-D-glycerate,
D-Glycerate 3-phosphate
6-Phospho-D-gluconate

D-Glucono-1,5-lactone
6-phosphate
Acetate, Acetic acid,
Ethanoic acid
Acetyl-CoA, Acetyl
coenzyme A
Acetyl phosphate
Acetaldehyde, Ethanal
Citrate, Citric acid,
2-Hydroxy-1,2,3-
propanetricarboxylic acid,
2-Hydroxytricarballylic acid
Coenzyme A, CoA-SH
Glycerone phosphate,
Dihydroxyacetone phosphate

C3H707P

C3H707P

C6H13010P

C6H1109P

C2H402

C23H38N7017P3S

C2H505P

C2H40
C6H807

C21H36N7016P3S

C3H7O6P

2PG

3PG

6PG

6PGL

ACE

ACoA

ACP
ADH
CIT

CoA
DRAP

C 00631

C 00197

C 00345

C01236

C 00033

C 00024

C 00227
C 00084
C 00158

C 00010
C 00111

continued on next page



APPENDIX A. TABLE OF SYMBOLSAND ABBREVIATIONS

Table A.2. : (continued)

Abbreviationll Full Name

	

I chemical

	

I Substance
Composition

	

ID Internet

3-Phospho-D-glyceroyl
phosphate,
1,3-Bisphospho-D-glycerate,
(R)-2-Hydroxy-3-
(phosphonooxy)-1-
monoanhydride
beta-D-Fructose 6-phosphate
beta-D-Fructose
1,6-bisphosphate
Formate, Methanoic acid,
Formic acid
Fumarate, Fumaric acid,
Trans-butenedioic acid
sn-Glycerol 3-phosphate,
Glycerophosphoric acid,
Phosphatidyl glycerol,
sn-Gro-1-P,
Glycerol-3-phosphate
(2R)-2-Hydroxy-3-
(phosphonooxy)-propanal,
D-Glyceraldehyde
3-phosphate
alpha-D-Glucose 6-phosphate

D-Glucose extracellular,
Grape sugar, Dextrose
Glycerone,
Dihydroxyacetone,
1,3-Dihydroxyacetone,
1,3-Dihydroxy-2-propanone,
1,3-Dihydroxypropan-2-one
Glycerol, Glycerin,
1,2,3-trihydroxypropane,
1,2,3-propanetriol

C3H8010P2

C6H1309P
C6H14012P2

CH202

C4H404

C3H906P

C3H7O6P

C6H1309P

C6H1206

C3H603

C3H803

DPG

F6P
FBP

FOR

FUM

G3P

GAP

G6P

GLC

GLO

GLY

C 00236

C 05345
C 05378

C 00058

C 00122

C 00093

C00118

C 00668

C 00031

C 00184

C00116

continued on next page



Table A.2. : (continued)

Abbreviation

ISC

LAC

MAL

MGL

OAA

OXA

OXO

PEP

PYR

R5P

Full Name

A.1 . METABOLITESANDNUCLEOTIDES

chemical Substance
Composition

	

I
ID Internet

Isocitrate, Isocitric acid,
1-Hydroxytricarballylic acid,
1-Hydroxypropane-1,2,3-
tricarboxylic
acid
(R)-Lactate, D-Lactate,
D-Lactic acid,
D-2-Hydroxypropanoic acid,
D-2-Hydroxypropionic acid
(S)-Malate, L-Malate,
L-Apple acid, L-Malic acid,
L-2-Hydroxybutanedioic acid
Methylglyoxal,
Pyruvaldehyde, Pyruvic
aldehyde,
2-Ketopropionaldehyde,
2-Oxopropanal
Oxaloacetate, Oxalacetic
acid, Oxaloacetic acid,
2-oxobutanedioic acid,
Oxosuccinic acid
Oxalosuccinate,
Oxalosuccinic acid
2-Oxoglutarate, Oxoglutaric
acid, 2-Ketoglutaric acid,
alpha-Ketoglutaric acid
Phosphoenolpyruvate,
Phosphoenolpyruvic acid
Pyruvate, Pyruvic acid,
2-Oxopropanoate,
2-Oxopropanoic acid,
Pyroracemic acid
D-Ribose 5-phosphate,
Ribose 5-phosphate

C6H807

C3H603

C4H605

C3H402

C4H405

C6H607

C5H605

C3H506.P

C3H403

C5H1108p

C 00311

C 00256

C 00149

C 00546

C 00036

C 05379

C 00026

C 00074

C 00022

C 00117

continued on next page
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Table A .2 . : (continued)

Abbreviationll Full Name

	

I chemical

	

I Substance
Composition

	

ID Internet

D-Ribulose 5-phosphate

Succinyl-CoA, Succinyl
coenzyme A
S-Succinyldihydrolipoamide
D-Sedoheptulose
7-phosphate,
D-altro-Heptulose
7-phosphate
Succinate, Succinic acid,
Butanedionic acid,
Ethylenesuccinic acid
D-Xylulose 5-phosphate

C5H1108P

C25H4oN7019P3s

C12H21NO4S2
C7H15010P

C4H604

C5H1108P

RL5P

SCoA

SDHP
SDP

SUC

X5P

C 00199

C00091

C 01169
C 05382

C 00042

C 00231



A.1 . METABOLITESAND NUCLEOTIDES

Table A.3. : Table of abbreviations : Nucleotides and others .

Abbreviationll Full Name

	

I chemical

	

I Substance
Composition

	

ID Internet

Adenosine 5'-diphosphate
Adenosine
5'-monophosphate, Adenylic
acid, Adenylate, 5'-AMP,
5'-Adenylic acid,
5'-Adenosine
monophosphate, Adenosine
5'-phosphate
Adenosine 5'-triphosphate

3',5'-Cyclic AMP, Cyclic
adenylic acid, Cyclic AMP,
Adenosine 3',5'-phosphate
Nicotinamide adenine
dinucleotide

Nicotinamide adenine
dinucleotide phosphate,
beta-Nicotinamide adenine
dinucleotide phosphate

Orthophosphate, Phosphate

C10H15N5010P2

C10H14N5O7P

C10H16N5013P3

C10H12N5O6P

C21H28N7014P2

C21H29N7014P2

C21H28N7017P3

C21H30N7017P3

H304P

ADP
AMP

ATP

cAMP

NAD

NADH
NADP

NADPH
P

C00008
C00020

C00002

C00575

C00003

C00004
C00006

C00005
C00009



APPENDIX A. TABLE OF SYMBOLS AND ABBREVIATIONS

A.2 . Enzymes

Table A.4 summarizes enzyme abbreviations, names, and official E .C . numbers .
Shown is also the main reaction that is Caayzed by each enzyme.

Table A.4. : Table of abbreviations : Enzymes .

EC
Number

Glycerol:NAD+
2-oxidoreductase
Sn-Glycerol-3-
phosphate:NAD+
2-oxidoreductase

(R)-Lactate:NAD+
oxidoreductase
(S)-Malate:NAD+
oxidoreductase
(S)-Malate:NAD+
oxidoreductase
(oxaloacetate-
decarboxylating)
(S)-Malate:NADP+
oxidoreductase
(oxaloacetate-
decarboxylating)

Isocitrate:NADP+
oxidoreductase
(decarboxylating)
6-Phospho-D-
gluconate:NADP+
2-oxidoreductase
(decarboxylating)

Glycerol
dehydrogenase
Glycerol-3-
phosphate
dehydrogenase
(NAD+)
D-Lactate
dehydrogenase
Malate
dehydrogenase
Malate
dehydrogenase
(oxaloacetate-
decarboxylating)
Malate
dehydrogenase
(oxaloacetate-
decarboxylating,
NADP+)
Isocitrate
dehydrogenase
(NADP+)
Phosphogluconate
dehydrogenase
(decarboxylating)

1 .1 .1.28

1.1 .1 .37

1.1 .1.38

1 .1.1 .40

1 .1 .1.42

1 .1.1.44

Systematic Name Recommended
Name

continued on next page

Reaction
Catalysed

GLY + NAD t~
GLO + NADH
G3P + NAD ~->
DHAP + NADH

PYR + NADH
~-+ LAC + NAD
OAA +NADH
<--> MAL + NAD
MAL + NAD --~
PYR + NADH +
C02

MAL + NADP
-> PYR +
NADPH + CO2

ISC + NADP -->
OXO + NADPH
+ C02
6PG + NADP
RL5P + NADPH
+ C02



EC

	

11 Systematic Name

	

Recommended
Number

	

I
Name

D-Glucose-6-
phosphate:NADP+
1-oxidoreductase

Aldehyde:NAD+
oxidoreductase

Acetaldehyde:NAD+
oxidoreductase
(CoA-acetylating)
D-Glyceraldehyde-3-
phosphate : NAD+
oxidoreductase
(phosphorylating)

Lactaldehyde:NAD+
oxidoreductase

Pyruvate :
ferricytochrome-bl
oxidoreductase
2-Oxoglutarate :
lipoamide
2-oxidoreductase
(decarboxylating
and acceptor-
succinylating)

Succinate :
(acceptor)
oxidoreductase

Sedoheptulose-7-
phosphate:D-
glyceraldehyde-3-
phosphate
glycolaldehydetrans-
ferase

Glucose-6-phosphate
1-dehydrogenase

Aldehyde
dehydrogenase
(NAD +)
Acetaldehyde
dehydrogenase
(acetylating)

Glyceraldehyde-3-
phosphate
dehydrogenase

Lactaldehyde
dehydrogenase

Pyruvate
dehydrogenase
(cytochrome)
Oxoglutarate
dehydrogenase
(lipoamide)

Succinate
dehydrogenase

Transketolase

1 .1.1 .49

1.2.1.10

1.2.1.12

1 .2.1 .22

1 .2.2.2

1 .2.4.2

1 .3.99.1

continued on next page

A.2 . ENZYMES

Reaction
Catalysed

G6P + NADP t~
6PGL + NADPH

ADH + NAD +
H20HACE+
NADH

ACoA + NADH
<-> CoA + ADH
+ NAD
GAP + NAD +
P <-> DPG +
NADH

MGL + NAD +
H20 ~-+ PYR+
NADH

PYR + H20 -~
ACE + C02

OXO -+ SDHP +
C02

FUM +NADH
H SUC + NAD,
FUM +NADPH
t-> SUC + NADP

X5P + R5P t->
GAP + SDP,
X5P + GAP <->
F6P
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EC

	

II
Systematic Name

	

Recommended

	

Reaction
Number

	

I Name

	

I Catalysed

Acetyl-CoA:formate
C-acetyltransferase

Succinyl-

	

1 Dihydrolipoamide
CoA :dihydrolipoamid S-succinyltransferase
S-succinyltransferase
ATP :D-fructose-6-
phosphate
1-
phosphotransferase
ATP :glycerol 3-
phosphotransferase
ATP :pyruvate 2-0-
phosphotransferase
ATP :3-phospho-D-
glycerate
1-
phosphotransferase
ATP :pyruvate,water
phosphotransferase

6-Phospho-D-
glucono-1,5-lactone
lactonohydrolase
Orthophosphoric-
monoester
phosphohydrolase
(alkaline optimum)
D-Fructose-1,6-
bisphosphate
1-phosphohydrolase

Formate
C-acetyltransferase

6-
Phosphofructokinase

Glycerol kinase

Pyruvate kinase

Phosphoglycerate
kinase

Pyruvate,water
dikinase

6-Phosphoglucono-
lactonase

Alkaline
phosphatase

Fructose-
bisphosphatase

2.3 .1 .54

2.3.1.61

2.7.1 .11

2.7 .1 .30

2.7 .1 .40

2.7.2.3

2.7.9.2

3.1 .1 .31

3.1 .3.11

continued on next page

CoA + PYR +
NAD -~ ACoA +
FOR + C02
SCoA ~ SDHP
+ CoA

F6P + ATP -~
FBP + ADP

GLY + ATP
G3P + ADP
PEP + ADP 44
PYR + ATP
DPG + ADP t~
3PG + ATP

PYR + ATP +
H20-> PEP +
AMP + P
6PGL + H20
6PG

GLO+PH
DHAP + H20

FBP -> F6P + P



EC

	

II Systematic Name
Number

Orthophosphate :
oxaloacetate
carboxy-lyase
(phosphorylating)
ATP:oxaloacetate
carboxy-lyase
(transphosphorylat-
ing)
D-Fructose-1,6-
bisphosphate
D-glyceraldehyde-3-
phosphate-lyase
Citrate
oxaloacetate-lyase
((pro-3S)-
CH2C00-L
acetyl-CoA)
(S)-Malate
hydro-lyase
Citrate
(isocitrate) hydro-
lyase
2-Phospho-D-
glycerate
hydro-lyase
Glycerone-
phosphate
phospho-lyase
D-Ribulose-5-
phosphate

~ 3-epimerase

Phosphoenolpyruvate
carboxylase

Phosphoenolpyruvate
carboxykinase
(ATP)

Fructose-
bisphosphate
aldolase

Citrate (si)-synthase

Fumarate hydratase

Aconitate hydratase

Enolase

Methylglyoxal
synthase

Ribulose-phosphate
3-epimerase

4.1.1.31

4.1 .1.49

4.1 .2.13

4.1 .3 .7

4.2.1.2

4.2 .1 .3

4.2 .1 .11

4.2.99 .11

Recommended Reaction
Name

	

1 Catalysed

continued on next page

A.2. ENZYMES

PEP + H2O +
C02 --> OAA +
P

OAA + ATP -a
ADP + C02

FBP ~-> DHAP +
GAP

OAA + ACoA +
H20t->CIT+
CoA

MAL <--> FUM +
H20
CIT H ISC +
H20

2PG ~-> PEP
+H20

DHAP H MGL
+P

RL5P t-+ X5P
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D-Glyceraldehyde-3-
phosphate
ketol-isomerase
D-Ribose-5-
phosphate
ketol-isomerase
D-Glucose-6-
phosphate
ketol-isomerase
D-Phosphoglycerate
2,3-phosphomutase
Succinate:CoA
ligase
(ADP-forming)

Triose-phosphate
isomerase

Ribose-5-phosphate
isomerase

Glucose-6-phosphate
isomerase

Phosphoglycerate
mutase
Succinate-CoA
ligase
(ADP-forming)

EC

	

II Systematic Name

	

Recommended

	

Reaction
Number

	

I
Name

	

I
Catalysed

5.3 .1.6

5.3 .1.9

5.4.2.1

6.2.1.5

DRAP t-> GAP

RL5P <-> R5P

G6P t-+ F6P

3PG 14 2PG

SUC -I- ATP t->

SCoA + ADP +
P



B . Pathway Maps

Figure B.1 shows the main metabolic pathwys of Escherichia coli . Figure B .2 shows
the reactions of the nucleotides and the legend for all pathway figures in this work.
Figure B.3 shows the saine pathway as Figure B.1 but includes the official enzyme
number for each reaction step.
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C. Kinetic Rate Equations

Notation

This appendix shows the kinetic rate equations used in the metabolic pathway mod-
els in this work . The following notation is used in this appendix:

. A and B are substrates of the reactions .

. P and Q are products of the reactions .

9 I is an inhibitor .

. v is the reaction rate . vma , is a maximum reaction rate in irreversible reac-
tions . vf and v,. are the forward respectively reverse reaction rates in reversible
reactions .

. Ke9 is the equilibrium constant in reversible reactions .

. Km, are half saturation constants .

. Ki . are inhibition constants .

Irreversible Michaelis-Menten Equation:

A
v = vmax ' KmA + A

Irreversible Michaelis-Menten Equation with two Substrates

v-A-B
v = KmA .KmB+Km,B .A+Km,A .B+A .B

(C.2)
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Reversible Michaelis-Menten Equation

A p
71f " KmA

	

- vT

	

KMPv _-

	

A

	

P1 + KMA + Kmp

Hill-Kinetics

v = vmax ' Kh + Ah

h: Hill coefficient

Ah

Ordered Uni-Bi Kinetics

vf-(A- KQ)v =
KmA+A- (l+Kp )+ v,,K -(K.Q-P+K.p-Q+P-Q)

Ordered Bi-Uni Kinetics

Ordered Inhibited Bi-Uni Kinetics

Ordered Bi-Bi Kinetics:

(C .3)

(C .4)

(C .5)

vf .(A-B-
~)

	

(C.6)v =
A-B+KmA-B+KmB- A+ vTKq-(Kmp+Q'(1+KzA))

v =

	

vf .(A .B_ Q)
A-B+K�,,A-B+KmB- A+ vrK (KmP+Q ' (1+KiA )) - I- (Ki ) n (C.7)

n: strength of inhibition, usually dependent on number of bindings sites of the
enzyme.

vf . (A .B_PQ)v, _
X

	

. (c.s)



where the denominator is

X

	

=

	

A-B-(1+

	

P
) + Km,B-(A+Ki,A)+Km,A-B + . . .

Ki,p

v,f . (Km,P'P(1+ FfA))+Q . (Km,P . (1+ KiAKB) +P'(1 +IfBs»
vr . Ke9

(C.9)
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